UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D. C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): June 30, 2015

Applied Optoelectronics, Inc. (Exact name of Registrant as specified in its charter)

Delaware (State or Incorporation 001-36083 (Commission File Number) 76-0533927 (I.R.S. Employer Identification No.)

13115 Jess Pirtle Blvd. Sugar Land, TX 77478

(address of principal executive offices and zip code)

(281) 295-1800 (Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

o Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01 Entry into a Material Definitive Agreement.

On June 30, 2015, Applied Optoelectronics, Inc. (the "Company"), entered into with East West Bank and Comerica Bank (collectively, the "Lenders") a credit agreement, a second lien deed of trust, multiple security agreements and promissory notes (collectively, the "Loan Agreements") evidencing (a) two revolving credit facilities, (herein referred to as the "Revolving Facility A" and "Revolving Facility B") and (b) a term loan (collectively, the "Credit Facilities"). The Credit Facilities provide that the Company may (a) from time to time prior to June 30, 2018, borrow, pay back and re-borrow up to an aggregate amount of \$25 million and (b) borrow \$10 million under the term loan facility.

The Company is required to repay all outstanding principal plus all accrued unpaid interest on or prior to June 30, 2018 for the revolving facilities and June 30, 2020 for the term loan facility. In addition, the Company will pay regular monthly payments of all accrued unpaid interest due as of each payment date, beginning July 31, 2015, with all subsequent interest payments to be due on the same day of each month thereafter. The Company may pay without penalty all or a portion of the amount owed earlier than due.

Under the Loan Agreements, the Credit Facilities bear interest at an annual rate based on the London Interbank Offered Rate (LIBOR) for deposits in U.S. Dollars plus: (i) Revolving Facility A + 2.75%, (ii) Revolving Facility B + 3.00%, and (iii) Term Loan + 2.75%.

The Company's obligations under the Loan Agreements will be secured by substantially all of the Company's tangible and intangible property, including but not limited to the Company's inventory, accounts, instruments, equipment and a second lien on its new Houston facility under construction, but excluding such items that are located at and used by the Company's Taiwan branch.

The Loan Agreements shall be subject to customary financial covenants, including the maintenance of a minimum current ratio, measured on a quarterly basis, of 1.25 to 1.00, and minimum consolidated adjusted annual EBITDA, measured quarterly on a rolling-4-quarter basis, of greater than or equal to: (a) \$11 million for each reporting period ending January 1, 2015 through December 31, 2015, (b) \$12.5 million for reporting period ending January 1, 2016 through December 31, 2017, (b) \$12.5 million for reporting period ending after December 31, 2017 and (c) \$12.5 million or as otherwise determined by the Lenders thereafter. The covenant levels for reporting period ending after December 31, 2017 and thereafter are to be reset annually based on Company projections. The Loan Agreements also requires the Company to maintain a consolidated leverage ratio not to exceed: (i) 5.00 to 1.00, at the end of any fiscal quarter ending on or after January 1, 2018 and on or before December 31, 2019, and (iii) 4.00 to 1:00, at the end of any fiscal quarter thereafter.

The Loan Agreements contain customary events of default that include, among others, non-payment of principal, interest or fees, violation of certain covenants, defective collateralization, inaccuracy of representations and warranties, and insolvency events. The Credit Facilities are also cross-defaulted with the Company's construction loan from East West Bank on its new Houston facility. The occurrence of an event of default will increase the applicable rate of interest by 5.00% and could result in the acceleration of the Company's obligations under the Loan Agreements.

The foregoing description of the Loan Agreements does not purport to be a complete statement of the parties' rights and obligations under the Loan Agreements and is qualified in its entirety by reference to the full text of the Loan Agreements, copies of which are attached as Exhibit 10.1, 10.2, 10.3, 10.4, 10.5, 10.6, and 10.7 to this Current Report on Form 8-K and incorporated by reference herein.

Item 1.02 Termination of a Material Definitive Agreement.

In connection with entering into the Loan Agreements described in Item 1.01 of this report, the Company, on June 30, 2015, terminated its existing \$15 million Revolving Credit Agreement with East West Bank which was originally executed on July 18, 2014.

Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

The information contained in Item 1.01 of this Current Report on Form 8-K with respect to the Loan Agreements is incorporated by reference herein and made a part hereof.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

10.1 Credit Agreement, dated June 30, 2015, between Applied Optoelectronics, Inc. and Lenders.

10.2 Security Agreement, dated June 30, 2015, between Applied Optoelectronics, Inc. and Lenders.

10.3 Patent Security Agreement, dated June 30, 2015, between Applied Optoelectronics, Inc. and Lenders.

10.4 Trademark Security Agreement, dated June 30, 2015, between Applied Optoelectronics, Inc. and Lenders.

10.5 East West Bank Promissory Note, dated June 30, 2015, between Applied Optoelectronics, Inc. and East West Bank.

10.6 Comerica Bank Promissory Note, dated June 30, 2015, between Applied Optoelectronics, Inc. and Comerica Bank.

10.7 2nd Lien Construction Deed of Trust, dated June 30, 2015, between Applied Optoelectronics, Inc. and Lenders.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Company has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: July 7, 2015

APPLIED OPTOELECTRONICS, INC.

 By:
 /s/ David C. Kuo

 Name
 David C. Kuo,

 Title:
 General Counsel and Vice President

Exhibit 10.1

CREDIT AGREEMENT

Dated as of June 30, 2015

by and between

APPLIED OPTOELECTRONICS, INC., as the Borrower,

the LENDERS listed on the signature pages hereto, as the Lenders,

and

EAST WEST BANK, as the Agent

TABLE OF CONTENTS

		Page
ARTICLE	IDEFINITIONS	1
1.1	Defined Terms	1
1.2	Other Definitional Provisions	15
ARTICLE	II THE LOANS	15
2.1	The Revolving Loans.	15
2.2	The Term Loans.	15
2.3	Repayment.	16
2.4	Interest Rates	17
2.5	Continuation and Conversion Options	17
ARTICLE	III GENERAL PROVISIONS CONCERNING THE LOANS	18
3.1	Use of Proceeds	18
3.2	Making the Loans	18
3.3	Transactional Amounts	18
3.4	Post-Maturity Interest and Late Fees.	18
3.5	Computation of Interest and Fees; Determinations by Lender.	19
3.6	Payments	19
3.7	Payment on Non-Business Days	19
3.8	Inability to Determine Interest Rate; Ineffective Interest Rate	19
3.9	Breakage Fees	20
3.10	Reserved.	20
3.11	Calculations	20
3.12	Deposit Account	20
3.13	Special Funding Provisions	20
ADTICLE		22
4.1	IV REPRESENTATIONS AND WARRANTIES Organization	22
4.1	Authorization	22
4.2	No Conflict	22
4.5	Governmental Approval	23
4.5	Validity	23
4.6	Financial Matters.	23
4.7	Corporate Structure and Ownership	23
4.8	Partnerships	24
4.9	Insurance	24
4.10	Litigation	24
4.11	Employee Benefit Plans	24
4.12	Environmental Matters	24
4.13	Title to Properties; Liens	25
4.14	Payment of Taxes	25
4.15	Governmental Regulation	25
4.16	Governmental Approval, Intellectual Property, Etc.	25

i

4.17	Labor Disputes and Casualties	25
4.18	Compliance	25
4.19	Margin Stock	26
4.20	Personal Property Collateral Matters.	26
4.21	USA PATRIOT Act, OFAC and Other.	27
4.22	Solvency	27
4.23	Disclosure	27
ARTICLE	V CONDITIONS OF LENDING	28
5.1	Conditions Precedent to Initial Loans	28
5.2	Conditions Precedent to Each Borrowing	29
	VI COVENANTS	30
6.1	Financial Information	30
6.2	Notices and Information	31
6.3	Corporate Existence, Etc.	32
6.4	Payment of Obligations	32
6.5	Maintenance of Properties	32
6.6	Insurance	33
6.7	Inspection	33
6.8	Compliance with Laws, Etc.	33
6.9	Books and Records	33
6.10	Additional Subsidiaries	33
6.11	Post-Closing Matters	34
		24
	VII NEGATIVE COVENANTS	34 34
7.1	Financial Covenants	
7.2 7.3	Liens, Etc. Debt	35 35
7.3 7.4		33
7.4	Lease Obligations	36
7.5	Equity Payments, Etc. Fundamental Changes	30
7.0	5	36
7.8	Loans, Investments, Contingent Liabilities Asset Sales	37
7.9	Transactions with Affiliates	37
7.10	Conduct of Business	37
7.10	Fiscal Year	37
7.12	Security Matters.	37
7.12	Limitation on Other Restrictions on Liens	37
7.13	Limitation on Other Restrictions on Amendment of the Loan Documents	38
7.15	Limitations on Modifications of Certain Agreements and Instruments	38
7.16	Agreements Related to Negative Covenants	38
7.10	Compliance with Anti-Terrorism Regulations	38
/.1/	Companie with find ferrorisin regulations	50

ii

ARTICLE VIII EVENTS OF DEFAULT		39
8.1	Events of Default	39
8.2	Application of Funds	41
ARTICLE	IX AGENCY	41
9.1	Appointment and Authority	41
9.2	Rights as a Lender	42
9.3	Exculpatory Provisions	42
9.4	Reliance by Agent	43
9.5	Delegation of Duties	43
9.6	Resignation of Agent	43
9.7	Non-Reliance on Agent and Other Lenders	44
ARTICLE	X MISCELLANEOUS	44
10.1	Amendments, Etc.	44
10.2	No Implied Waiver; Remedies Cumulative	45
10.3	Notices	45
10.4	Expenses	45
10.5	Indemnity.	46
10.6	Assignments and Participations.	46
10.7	Entire Agreement	49
10.8	Survival	49
10.9	Counterparts	49
10.10	Severability	49
10.11	Headings	49
10.12	Setoff	49
10.13	Sharing of Payments By Lenders	50
10.14	Limitation on Payments	50
10.15	Disclosure of Information to Affiliates, Confidentiality.	51
10.16	Governing Law	52
10.17	Waiver of Jury Trial	52
10.18	Consent to Jurisdiction; Venue	52
10.19	USA Patriot Act Notice	52
10.20	Keepwell	52
10.21	Limitation of Liability	53

iii

ANNEXES	
1	Commitments
2	Agent's Office
EXHIBITS	
А	Compliance Certificate
В	Assignment and Assumption
С	Notice of Loan Request
D	Borrowing Base Certificate
E	Revolving Notes
F	Term Notes
G	Global Intercompany Note
SCHEDULES	
4.6	Liabilities
4.7	Capitalization and Ownership
4.10	Litigation
4.12	Environmental Claims
4.16	Licenses and Intellectual Property Matters
4.20	Security Matters
6.11	Post-Closing Matters
7.2	Liens
7.3	Debt
7.7	Investments
7.9	Transactions with Affiliates

iv

CREDIT AGREEMENT

THIS CREDIT AGREEMENT, dated as of June 30, 2015, between APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "<u>Borrower</u>"), each lender from time to time party hereto (collectively, the "<u>Lenders</u>" and individually, a "<u>Lender</u>") and EAST WEST BANK, a California state chartered bank, as Agent (in such capacity, the "<u>Agent</u>"). The parties hereto, intending to be legally bound hereby, agree as follows:

ARTICLE I DEFINITIONS

1.1 Defined Terms. In addition to terms defined elsewhere in this Agreement, as used in this Agreement, the following terms have the following meanings:

"Administrative Questionnaire": an Administrative Questionnaire in a form supplied by the Agent.

"<u>Advance Period</u>": As applied to any Borrowing of Facility B Revolving Loans, the date beginning on the first day and ending on the last day of the calendar month in which such Borrowing is made.

"<u>Affiliate</u>": As applied to any Person (the "<u>Specified Person</u>"), any other Person directly or indirectly controlling, controlled by, or under common control with, the Specified Person. For the purposes of this definition, "control" (including, with correlative meanings, the terms "controlling", "controlled by" and "under common control with"), as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of the Specified Person, whether through the ownership of voting securities or by contract or otherwise.

"Agent": East West Bank in its capacity as agent under any of the Loan Documents, or any successor agent.

"<u>Agent's Office</u>": The Agent's address and, as appropriate, account as set forth on <u>Annex 2</u>, or such other address or account as the Agent may from time to time notify to the Borrower and the Lenders.

"Agreement": This Credit Agreement, as amended, supplemented or modified from time to time.

"<u>Anti-Terrorism Law</u>": Any Law related to money laundering or financing terrorism including the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act ("<u>USA PATRIOT Act</u>") of 2001 (Title III of Pub. L. 107-56), The Currency and Foreign Transactions Reporting Act (also known as the "Bank Secrecy Act", 31 U.S.C. §§ 5311-5330 and 12 U.S.C. §§ 1818(s), 1820(b) and 1951-1959), the Trading With the Enemy Act (50 U.S.C. § 1 et seq., as amended) and Executive Order 13224 (effective September 24, 2001).

"<u>Applicable Margin</u>": The Applicable Margins applicable to LIBOR Rate Loans and Base Rate Loans shall be determined from the following table:

Loan Type	LIBOR Rate Loans	Base Rate Loans
Facility A Revolving Loans	2.75%	0.00%
Facility B Revolving Loans	3.00%	0.25%
Term Loans	2.75%	0.00%

"<u>Applicable Percentage</u>": With respect to a Lender at any time, the percentage of the aggregate of all Revolving Commitments and Term Commitments represented by such Lender's Revolving Commitment and Term Commitment or if one or more of such Commitments has been terminated, such percentage of the aggregate outstanding Revolving Loans and Term Loans represented by such lender's Revolving Loans and Term Loans, as the case may be.

"AR Availability": Eighty-five percent (85%) of the Net Value of Eligible Accounts.

"<u>Assignment and Assumption</u>": An assignment and assumption entered into by a Lender and an Eligible Assignee (with the consent of any party whose consent is required by <u>Section 10.6</u>) and accepted by the Agent, in substantially the form of <u>Exhibit B</u> or any other form approved by the Agent.

"<u>Base Rate</u>": At any time, the Prime Rate.

"Base Rate Loans": Loans bearing interest at a rate based upon the Base Rate.

"Borrowing": Each borrowing of a Loan under Section 2.1 or 2.2.

"Borrowing Base": At any time, the sum, at the date of the most recent Certificate required to be furnished pursuant to Section 6.1(e) or (f), of AR Availability, plus Inventory Availability.

"<u>Borrowing Base Certificate</u>": A certificate substantially in the form of <u>Exhibit D</u> hereto, appropriately completed, signed by an officer of the Borrower and setting forth the Borrowing Base as of the end of the previous month and the other information required therein.

"Business Day": A day other than a Saturday, Sunday or a day on which commercial banks in Los Angeles, California are authorized or required by Law to close.

"<u>Capital Expenditures</u>": For any period, as applies to any Person, the aggregate of all expenditures for such Person for the purchase or other acquisition of any machinery and equipment, computer equipment and software and excluding land improvements, building improvements, furniture and fixtures and transportation equipment.

"<u>Capital Lease</u>": As applied to any Person, any lease of any property (whether real, personal or mixed) by that Person as lessee that, in conformity with GAAP, is or is required to be accounted for as a capital lease on the balance sheet of that Person.

"<u>Cash Management Agreements</u>": Any agreements regarding treasury management arrangements or depositary and other cash management services (including overdrafts and related liabilities arising therefrom) or in connection with any automated clearing house transfers of funds, in each case, provided by the Agent or an Affiliate of the Agent.

"<u>Change in Control</u>": Any transaction in which any "person" or "group" (within the meaning of Section 13(d) and 14(d)(2) of the Securities Exchange Act of 1934) becomes the "beneficial owner" (as defined in Rule 13d-3 under the Securities Exchange Act of 1934), directly or indirectly, of a sufficient number of shares of all classes of stock then outstanding of the Borrower ordinarily entitled to vote in the election of directors, empowering such "person" or "group" to elect a majority of the Board of Directors of the Borrower, who did not have such power before such transaction.

"Closing Date": June 30, 2015.

statute.

"Code": The Internal Revenue Code of 1986, as amended, and any successor statute or provision thereof.

"<u>Collateral</u>": As defined in the Security Agreement, but which expressly excludes Excluded Collateral.

"<u>Commitment</u>" or "<u>Commitments</u>": The commitment of each Lender to make Loans to the Borrower pursuant to <u>Article II</u> in the amount or amounts referred to therein.

"<u>Commodity Exchange Act</u>": The Commodity Exchange Act (7 U.S.C. Section 1 et seq.), as amended from time to time and any successor

"<u>Consolidated EBITDA</u>": For any period, an amount equal to the sum of net income (exclusive of extraordinary gains and losses) <u>plus</u>, without duplication, (a) Consolidated Interest Charges, (b) Consolidated Tax Expense, (c) depreciation and amortization expense in each case, (d) stock option expenses, (e) other non-cash expenses acceptable to the Agent in its sole discretion, and (f) non-recurring expenses satisfactory to the Agent in an aggregate amount not to exceed \$1,000,000, in each case, for such period, and determined and consolidated for the Borrower and its Subsidiaries in accordance with GAAP.

"<u>Consolidated Funded Debt</u>": At any time, Debt of the Borrower and its Subsidiaries at such time (including the current portion thereof) which would as of such date be classified in whole or in part as a long-term liability (including, without limitation, Capital Leases and bankers' acceptances), determined and consolidated in accordance with GAAP.

"<u>Consolidated Interest Charges</u>": For any period, the aggregate regularly scheduled interest due and payable by the Borrower and its Subsidiaries on Debt during such period, determined and consolidated in accordance with GAAP.

"Consolidated Leverage Ratio": The ratio set forth in Section 7.1(b).

"<u>Consolidated Tax Expense</u>": For any period, income tax expense for such period, determined and consolidated for the Borrower and its Subsidiaries in accordance with GAAP.

"<u>Current Ratio</u>": For any period, ratio of total current assets to total current liabilities, in each case, determined and consolidated for the Borrower and its Subsidiaries in accordance with GAAP.

"Debt": As applied to any Person, (a) all indebtedness for borrowed money, (b) that portion of obligations with respect to Capital Leases which is properly classified as a liability on a balance sheet in conformity with GAAP, (c) notes payable and drafts accepted representing extensions of credit whether or not representing obligations for borrowed money, (d) any obligation owed for all or any part of the deferred purchase price of property or services (other than trade accounts payable arising in the ordinary course of business for which payment is due and is made within 90 days or less), (e) all indebtedness secured by any Lien on any property or asset owned or held by that Person regardless of whether the indebtedness secured has been assumed by that Person or is nonrecourse to the credit of that Person, (f) obligations in respect of letters of credit, (g) obligations under Hedging Contracts (the amount of which shall be determined by reference to the termination cost on the date of determination), and (h) guarantees of, or similar obligations with respect to, any of the foregoing of any other Person.

"<u>Default</u>": A condition or event which, after the giving of notice or the lapse of time or both, would, unless cured or waived, constitute an Event of Default.

"Default Rate": Five percent (5%) above the highest rate which would otherwise be applicable to the Loans pursuant to Section 2.4.

"Defaulting Lender": Any Lender that has (i) failed to fund any portion of its Loans within three (3) Business Days of the date required to be funded by it hereunder, (ii) notified the Borrower and the Agent in writing that it does not intend to comply with any of its funding obligations under this Agreement or has made a public statement to such effect, or (iii) otherwise failed to pay over to the Agent or any other Lender any other amount required to be paid by it hereunder within three (3) Business Days of the date when due, or (iv) (A) become or is insolvent or has a parent company that has become, the subject of a bankruptcy or insolvency proceeding or has had, or had a parent company has had, a receiver, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or custodian, appointed for it, or has taken, or has a had a parent company that has taken, any action in furtherance of, indicating its consent to, approval of or acquiescence in, any such proceeding or appointment.

"Dollars" and "\$": The lawful currency of the United States of America.

"<u>Domestic Subsidiary</u>": If any, any Subsidiary that is organized and existing under the laws of the United States or any state or commonwealth thereof or under the laws of the District of Columbia.

"<u>Eligible Accounts</u>": Any account receivable, net of any prepayments, progress payments, deposits and retentions, owing to the Borrower which met the specifications established from time to time by the Agent, in its sole discretion, at the time it came into existence and continues to meet such specifications until it is collected in full.

Any other account must meet the following specifications at the time it comes into existence and continue to meet such specifications until it is collected in full in order to be an Eligible Account:

(a) The Borrower has good title to such account, free and clear of any Lien, except for the Liens in favor of the Agent, and such account is subject to such a valid and perfected Lien in favor of the Agent.

(b) Such account constitutes an "account" as defined in the Uniform Commercial Code as in effect in the State of New York (and, accordingly, without limitation, is not evidenced by any promissory note or other instrument).

(c) The account is not more than ninety (90) days from its invoice date (an "<u>Overdue Account</u>"), and not more than fifty percent (50%) of the accounts of the same obligor or its Affiliates (or, solely with respect to Biogenomics Corp., twenty-five percent (25%)) are Overdue Accounts.

(d) The accounts of the same obligor represent in excess of twenty-five percent (25%) (or, solely with respect to Synnex Corporation, fifty percent (50%) and solely with respect to World Wide Technology, thirty percent (30%)) of all Eligible Accounts.

(e) The account arose from the performance of services or an outright sale of goods by Borrower in the ordinary course of Borrower's business, including its customary credit practices, and such goods have been shipped, or services provided, to the account debtor, and invoices have been issued therefor;

(f) The account is not an account that the Agent has determined may not be paid by reason of the account debtor's financial condition or inability to pay;

(g) The account is not subject to any prior assignment, claim, set-off, credit, allowance or adjustment by the account debtor, except discounts allowed for prompt payment, and the account debtor has not complained as to its liability on the account and has not returned, or retained the right to return, any of the goods from the sale of which the account arose;

(h) The account is not owed by (i) the government (or any department, agency, public corporation, or instrumentality thereof) of any country other than the U.S. unless such account is backed by a letter of credit acceptable to the Agent which is in possession of the Lender, or (ii) the government of the U.S. or any department, agency, public corporation, or instrumentality thereof, unless in the case of accounts owed by the U.S., the Federal Assignment of Claims Act of 1940, as amended (31 U.S.C. § 3727 et seq. and 41 U.S.C. § 15 et seq.) and any other steps necessary to perfect the Lien of the Lender in such account have been complied with to the Lender's satisfaction;

(i) The account is not booked or collected in any of the Borrower's oversees branch offices;

(j) The obligor on such account (i) is a Person whose principal office is located in the United States (other than Biogenomics Corp. or a foreign publicly-traded company with operating results satisfactory to the Agent in its sole discretion and approved in writing by the Agent from time to time), (ii) is not the Borrower, any Subsidiary of the Borrower or an Affiliate of the Borrower, and (iii) is not insolvent, subject to any bankruptcy, insolvency or similar proceeding or unable to pay its debts as they become due;

(k) The account is not for goods that have been sold under a purchase order or pursuant to the terms of a contract or other agreement or understanding (written or oral) that indicates or purports that any Person other than the Borrower has or has had an ownership interest in such goods, or which indicates any party other than the Borrower as lender or remittance party;

(l) The account is owed in U.S. Dollars only;

(m) The account does not constitute a finance charge, a service charge or lease receivable; and

(n) Neither the obligor on such account nor any of its Affiliates is also a supplier to, or creditor of, the Borrower or any of its Subsidiaries, unless such supplier or creditor has executed a no offset letter in form and substance satisfactory to the Agent.

"<u>Eligible Assignee</u>": means (a) a Lender and (b) any other Person (other than a natural person) approved by (i) the Agent and (ii) unless an Event of Default has occurred and is continuing, the Borrower; <u>provided</u> that notwithstanding the foregoing, "Eligible Assignee" shall not include (i) the Borrower or any of the Borrower's Affiliates or Subsidiaries or (ii) any Person, that to the knowledge of the Agent, primarily invests in distressed securities.

"<u>Eligible Inventory</u>": Finished goods which are owned by the Borrower and available and suitable for use by the Borrower as inventory for sale to customers of the Borrower in the ordinary course of the Borrower's business, which meet each of the following requirements at such time:

(a) The Borrower has good title to such inventory, free and clear of any Lien (including, without limitation, Liens otherwise permitted by Section 7.2(f)), except for the Liens in favor of the Agent securing the Obligations, and such Eligible Inventory and proceeds thereof is subject to such a valid and perfected Lien in favor of the Agent.

(b) Such inventory is determined by the Agent to be in good and merchantable condition, is not obsolete, defective or unfit for sale, is readily saleable by the Borrower in the ordinary course of its business and has not been held by the Borrower for more than 360 days.

(c) Such inventory is not spare parts;

(d) Such inventory is not the subject of a consignment by the Borrower as consignor;

(e) Such inventory is located in the United States, and is (i) in the possession of the Borrower, or (ii) is in a location for which either (x) a warehouseman's or bailee's agreement in form and substance satisfactory to the Agent, in its sole discretion, has been entered into or (y) an adequate rent reserve in an amount determined by the Agent, in its sole discretion, has been established; <u>provided</u> that inventory located in the United States that is designated to be sold by Amazon or Cisco may be included in Eligible Inventory prior to the entry into a warehouseman's or bailee's agreement with respect to the location thereof so long as the Borrower is in compliance with <u>Section 6.11</u>;

(f) Such inventory is not in transit unless it is insured to the reasonable satisfaction of the Agent; and

(g) Such inventory is not reported under any of the Borrower's overseas branches.

"<u>Employee Benefit Plan</u>": Any employee benefit plan which is described in Section 3(3) of ERISA and which is maintained for employees of the Borrower or any ERISA Affiliate of the Borrower.

"Environmental Laws": Any and all current or future Laws, or any other requirements of Governmental Authorities relating to (a) environmental matters, or (b) occupational safety and health, industrial hygiene, land use or the protection of human, plant or animal health or welfare, in any manner applicable to the Borrower or any of its Subsidiaries or any facility owned, leased or operated by the Borrower or any of its Subsidiaries.

"<u>Equity Interests</u>": With respect to any Person, (a) all of the shares of capital stock of, or other ownership or profit interests in, such Person, whether voting or non-voting, and including any partnership, membership or trust interests, (b) all securities or Debt convertible into or exchangeable for any of the foregoing, whether directly or indirectly, and (c) all warrants, options and other rights to purchase or acquire any of the foregoing, whether directly or indirectly.

"ERISA": The Employee Retirement Income Security Act of 1974, as amended from time to time and any successor statute.

"ERISA Affiliate": As applied to any Person, any trade or business (whether or not incorporated) which is a member of a group of which that Person is a member and which is under common control within the meaning of Section 414(b), (c), (m) or (o) of the Code.

"ERISA Event": (a) A "Reportable Event" described in Section 4043 of ERISA and the regulations issued thereunder (other than a "Reportable Event" not subject to the provision for 30 day notice to the Pension Benefit Guaranty Corporation under such regulations), or (b) the withdrawal of the Borrower or any of its ERISA Affiliates from a Pension Plan during a plan year in which it was a "substantial employer" as defined in Section 4001(l) (2) or 4068(f) of ERISA, or (c) the failure to meet the minimum funding standard of Section 412 of the Code with respect to any Pension Plan (whether or not waived in accordance with Section 412(d) of the Code) or the failure to make by its due date a required installment under Section 412(m) of the Code with respect to any Pension Plan or the failure to make any required contribution to a Multiemployer Plan; or (d) the provision by the administrator of any Pension Plan pursuant to Section 4041(a)(2) of ERISA of a notice of intent to terminate such plan in a distress termination described in Section 4041(c) of ERISA, or (e) the institution of proceedings to terminate a Pension Plan by the Pension Benefit Guaranty Corporation, or (f) the withdrawal of the Borrower, any of its Subsidiaries or any of their respective ERISA Affiliates in a complete or partial withdrawal (within the meaning of Sections 4203 and 4205 of ERISA) from any Multiemployer Plan if there is any potential liability therefor, or the receipt by the Borrower, any of its Subsidiaries or any of their respective ERISA, or that it is in reorganization or insolvency pursuant to Section 4241 or 4245 of ERISA, or that it intends to terminate or has terminated under Section 4041A or 4042 of ERISA; or (g) the imposition of a lien pursuant to Section 412(n) of the Code.

"Excluded Collateral": As defined in the Security Agreement.

"Excluded Swap Obligation": With respect to any Guarantor, any Swap Obligation if, and to the extent that, all or a portion of the guaranty of such Guarantor of, or the grant by such Guarantor of a security interest to secure, such Swap Obligation (or any guaranty thereof) is or becomes illegal under the Commodity Exchange Act or any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof) by virtue of such Guarantor's failure for any reason to constitute an "eligible contract participant" as defined in the Commodity Exchange Act and the regulations thereunder at the time the guaranty of such Guarantor or the grant of such security interest becomes effective with respect to such Swap Obligation. If a Swap Obligation arises under a master agreement governing more than one swap, such exclusion shall apply only to the portion of such Swap Obligation that is attributable to swaps for which such guaranty or security interest is or becomes illegal.

"Existing Line of Credit Agreement": The Business Loan Agreement, dated July 15, 2014, between the Borrower and East West Bank, as

lender.

"Existing EWB Loan Documents": Each of (i) that certain Business Loan Agreement, dated July 31, 2014, between the Borrower and East West Bank, as lender; (ii) that certain Promissory Note issued by the Borrower in favor of East West Bank, dated July 31, 2014; (iii) that certain Construction Loan Agreement, dated January 26, 2015, between the Borrower and East West Bank, as lender; (iv) that certain Promissory Note issued by the Borrower in favor of East West Bank, dated January 26, 2015; (v) that certain Commercial Security Agreement, dated January 26, 2015, by the Borrower in favor of East West Bank; and (vi) each of the "Related Documents" as defined in the foregoing.

"Expiration Date": June 30, 2018.

"Facility A Revolving Loan": As defined in Section 2.1(a).

"<u>Facility B Revolving Commitment</u>": The commitment of a Lender to make Facility B Revolving Loans to the Borrower pursuant to <u>Section 2.1(b)</u> in an aggregate principal amount not in excess of the amount set forth opposite such Lender's name on <u>Annex 1</u>, as such amount may be adjusted from time to time in accordance with this Agreement.

"Facility B Revolving Loan": As defined in Section 2.1(b).

"<u>Federal Funds Effective Rate</u>": On any day, a fluctuating interest rate per annum (rounded upward to the nearest 1/100th of 1%) equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or if such rate is not so published for any day which is a Business Day, the average rate (rounded upward to the nearest 1/100th of 1%) charged to the Agent on such day on such transactions as determined by the Agent.

"Foreign Branch": Any foreign branch office of the Borrower or a Domestic Subsidiary, whether registered or unregistered.

"<u>Foreign Subsidiary</u>": Any Subsidiary of the Borrower that is not a Domestic Subsidiary.

"<u>Fund</u>": means any Person (other than natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of business.

"GAAP": United States generally accepted accounting principles applied on a consistent basis.

"<u>Global Intercompany Note</u>": That certain Global Intercompany Note among the Borrower and its Subsidiaries, dated as of the date hereof, as may be amended, restated, modified or supplemented from time to time.

"<u>Governmental Approval</u>": Any approval, order, consent, authorization, certificate, license, permit or validation of, or exemption or other action by, or filing, recording or registration with, or notice to, any Governmental Authority.

"<u>Governmental Authority</u>": Any government or political subdivision or any agency, authority, bureau, central bank, commission, department or instrumentality of either, or any court, tribunal, grand jury or arbitrator, in each case whether foreign or domestic.

"<u>Guarantors</u>": (a) Each Domestic Subsidiary who is required to become a Guarantor pursuant to <u>Section 6.10</u> and (b) with respect to the payment and performance by each Guarantor described in the foregoing clause (a) of such Guarantor's guaranty obligations as they relate to any Swap Obligation, the Borrower.

"<u>Guaranty</u>": A guaranty, in form and substance satisfactory to the Agent, for the benefit of the Lender Parties, executed and delivered by the Guarantors, for the benefit of the Lender Parties, as amended, modified or supplemented from time to time.

"<u>Hedging Contract</u>": Any rate or currency swap, cap or collar agreement or any other agreement designed to hedge risk with respect to interest rate or currency fluctuations, pursuant to a Master Agreement.

"Indemnified Liabilities": As defined in Section 10.6(a).

"Information": As defined in Section 10.15(b).

"<u>Intellectual Property</u>": Any patent, copyright, service mark, trademark or trade name registered, or for which registration has been applied for, in the United States or any jurisdiction thereof.

"<u>Intercreditor Agreement</u>": An intercreditor agreement, in form and substance satisfactory to the Agent, for the benefit of the Lender Parties, executed by the Borrower, the Agent and East West Bank as lender under the Existing EWB Loan Documents, as amended, modified or supplemented from time to time.

"Interest Payment Date": The first day of each calendar month and the Expiration Date and the Term Loan Maturity Date, as applicable.

"Interest Period": With respect to any LIBOR Rate Loan, the period commencing on the Borrowing, conversion or continuation date with respect to such LIBOR Rate Loan and ending one month, two months, or three months thereafter as selected by the Borrower in its Standard Notice of Borrowing as provided in <u>Section 3.2</u> or its notice of conversion or continuation as provided in <u>Section 2.5</u>; provided, however, that: (i) the first day of each Interest Period must be a LIBOR Business Day; (ii) the Borrower may not select an Interest Period with respect to any portion of principal of a LIBOR Rate Loan which extends beyond a date on which the Borrower is required to make a scheduled payment of that portion of principal or in any case beyond the Expiration Date or the Term Loan Maturity Date, as applicable; (iii) any Interest Period which would otherwise expire on a day which is not a LIBOR Business Day, shall be extended to the next succeeding LIBOR Business Day, unless the result of such extension would be to extend such Interest Period into another calendar month, in which event the Interest Period shall end on the immediately preceding LIBOR Business Day; and (iv) any Interest Period that begins on the last LIBOR Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last LIBOR Business Day of a calendar month.

"Inventory Availability": Thirty-five percent (35%) of the Net Value of Eligible Inventory; provided that for the purposes of calculating the Borrowing Base, Inventory Availability shall not exceed the lesser of (a) \$5,000,000 or (b) fifty percent (50%) of AR Availability.

"Investments": As defined in Section 7.7.

"<u>Law</u>": Any law (including common law), constitution, statute, treaty, convention, regulation, rule, ordinance, order, injunction, writ, decree, award, settlement agreement, requirement or determination of an arbitrator or a court of any Governmental Authority.

"Lender Parties": The Agent and the Lenders.

"LIBOR Business Day": A day which is a Business Day and on which dealings in Dollar deposits may be carried out in the London interbank market.

"LIBOR Rate": For each Interest Period, a rate per annum (based on a year of 360 days and actual days elapsed) equal to the rate of interest (which shall be the same for each day in such Interest Period) determined by the Agent in accordance with its usual procedures to be the rate at which Dollar deposits are offered by leading banks in the London interbank deposit market two (2) LIBOR Business Days prior to the first day of such Interest Period in an amount approximately equal to the then outstanding principal amount of the Loans for the designated Interest Period, as quoted by ICE Benchmark Administration ("ICE") or any successor thereto as approved by the Agent if ICE is no longer making a LIBOR rate quotation available (an "<u>Alternate Source</u>"), and as published as the "London Interbank Offered Rate" by Bloomberg (or if, at any time, for any reason, such rate is no longer published by Bloomberg or provided by ICE or any Alternate Source, a comparable replacement rate determined by the Agent at such time). Notwithstanding the foregoing, if at any time the LIBOR Rate as determined above is less than zero (0), it shall be deemed to be zero (0) for the purposes of this Agreement.

"LIBOR Rate Loans": Loans bearing interest at a rate based upon the LIBOR Rate.

"<u>Lien</u>": Any lien, mortgage, deed of trust, pledge, security interest, charge or encumbrance of any kind (including any conditional sale or other title retention agreement, any lease in the nature thereof, and any agreement to give any security interest).

"Loan Documents": This Agreement, the Notes, the Security Documents, the Intercreditor Agreement, any Master Agreement, and each additional document, notice or certificate delivered to the Agent and signed by or on behalf of a Loan Party in connection with this Agreement, the credit extended hereunder or the collateral securing the Obligations hereunder.

"Loan Party": The Borrower and any Guarantors, and "Loan Parties" means all such Persons, collectively.

"Loans": The Revolving Loans, the Term Loans and any combination thereof.

"<u>Master Agreement</u>": An ISDA Master Agreement, as in effect from time to time, including all schedules, confirmations and other documents delivered thereunder, pursuant to which the Borrower and the Agent (or an Affiliate of the Agent) may from time to time hereafter enter into interest rate hedging transactions.

"<u>Material Adverse Effect</u>": (i) A material adverse change in, or material adverse effect on, the business, operations, properties, assets, condition (financial or otherwise) or prospects of the Borrower or the Borrower and its Subsidiaries, taken as a whole, or (ii) the impairment of the ability of any of the Loan Parties to perform, or the Agent to enforce, the Obligations.

"<u>Maximum Facility A Revolving Commitment</u>": The commitment of a Lender to make Facility A Revolving Loans to the Borrower pursuant to <u>Section 2.1(a)</u> in an aggregate principal amount not in excess of the amount set forth opposite such Lender's name on <u>Annex 1</u>, as such amount may be adjusted from time to time in accordance with this Agreement.

"<u>Mortgage</u>": A deed of trust, in form and substance satisfactory to the Agent, executed and delivered by the Borrower to the Agent, as amended, modified or supplemented from time to time.

"Multiemployer Plan": A "multiemployer plan" as defined in Section 3(37) of ERISA.

"<u>Net Value of an Eligible Account</u>": The face amount, net of any discount for prompt payment (and net of any other amount representing payment of finance charges, late charges, or interest (however denominated)), and net of any portion thereof which constitutes payment of sales, use or other taxes.

"<u>Net Value of Eligible Inventory</u>": The Borrower's book value at lower of weighted average cost or market, net of all reserves required by GAAP, all determined in accordance with GAAP.

"Note" and "Notes": The Revolving Notes, the Term Notes and any combination thereof.

"notices": As defined in Section 10.3.

"<u>Obligations</u>": All obligations of every nature of the Loan Parties from time to time owed to any Lender Party under the Loan Documents, any Hedging Contract or Cash Management Agreement, whether for principal, interest, fees, expenses, indemnification or otherwise; <u>provided</u>, that the Obligations of a Loan Party shall exclude any Excluded Swap Obligations with respect to such Loan Party.

"Officer's Certificate": A certificate signed by the President, Chief Financial Officer or VP Finance of the Borrower.

"Participant": As defined in Section 10.6(d).

"Pension Plan": Any Employee Benefit Plan other than a Multiemployer Plan which is subject to Section 412 of the Code or Section 302 of

ERISA.

"<u>Permitted Foreign Debt</u>": Debt incurred outside of the United States by the Borrower on behalf of a Foreign Branch or by a Foreign Branch directly; <u>provided</u> that: (a) no Event of Default is continuing or would result from the incurrence of such Debt; (b) the Borrower is in pro forma compliance with the covenants contained in <u>Section 7.1</u> hereof immediately prior to and would be immediately after the incurrence of any such Debt (without taking into account any cure periods); (c) except for Liens permitted by <u>Section 7.2(g)</u>, such Debt is not secured by any Collateral (including, without limitation, any assets of the Borrower held by a Foreign Branch or otherwise in Taiwan that may constitute Collateral); (d) any such Debt is not guaranteed by any Loan Party (other than any unsecured guarantees made by the Borrower with respect to Debt incurred by a Foreign Branch); and (e) the Agent immediately receives notice of the incurrence thereof.

"Permitted Liens": Liens permitted by Section 7.2.

"<u>Person</u>": An individual, partnership, corporation, limited liability company, business trust, joint stock company, trust, unincorporated association, joint venture, Governmental Authority or other entity of whatever nature.

"<u>Prime Rate</u>": The interest rate per annum published in the New York edition of <u>The Wall Street Journal</u> from time to time as the "Prime Rate", (rounded upward to the nearest 1/100th of 1%) such rate to change automatically effective as of the effectiveness of each change in such prime rate. If <u>The Wall Street Journal</u> ceases to publish the "Prime Rate," the Agent shall select an equivalent publication that publishes such "Prime Rate," and if such "Prime Rates" are no longer generally published or are limited, regulated or administered by a governmental or quasi-governmental body, then the Agent shall select a comparable interest rate index. The Prime Rate is a reference rate and does not necessarily represent the lowest or best rate being charged to any customer.

"Projections": As defined in Section 4.6.

"<u>Qualified ECP Guarantor</u>": In respect of any Swap Obligations, each Loan Party that has total assets exceeding \$10,000,000.00 at the time the relevant guaranty or grant of the relevant security interest becomes effective with respect to such Swap Obligation or such other Person as constitutes an "eligible contract participant" under the Commodity Exchange Act or any regulations promulgated thereunder and can cause another Person to qualify as an "eligible contract participant" at such time by entering into a keepwell under Section 1a(18)(A)(v)(II) of the Commodity Exchange Act.

"Register": As defined in Section 10.6(c).

"<u>Related Parties</u>": With respect to any Person, such Person's Affiliates and the partners, members, directors, officers, employees, agents and advisors of such Person and of such Person's Affiliates.

"<u>Required Lenders</u>": Lenders that have more than 67% in the aggregate of the outstanding Loans and unused Commitments; <u>provided</u> that so long as there are fewer than three Lenders, "Required Lenders" shall mean all Lenders who are not Defaulting Lenders.

"Revolving Commitment": The Maximum Facility A Revolving Commitment and Facility B Revolving Commitment of a Lender.

"<u>Revolving Loans</u>": The Facility A Revolving Loans and the Facility B Revolving Loans.

"Revolving Note": As defined in Section 2.1(c).

"<u>Sanctions</u>": Sanctions administered or enforced by the US Department of the Treasury's Office of Foreign Assets Control (OFAC), US Department of State, United Nations Security Council, European Union, Her Majesty's Treasury, or other relevant sanctions authority.

"Secured Obligations": As defined in the Security Documents.

"<u>Security Agreement</u>": A security agreement, in form and substance satisfactory to the Agent, for the benefit of the Lender Parties, executed and delivered by the Borrower to the Agent, for the benefit of the Lender Parties, as amended, modified or supplemented from time to time.

"<u>Security Documents</u>": The Security Agreement, the Mortgage and any other agreements granting or purporting to grant the Agent a Lien to secure, or to guaranty the Obligations or subordinating other Debt to the Obligations.

"<u>Standard Notice</u>": An irrevocable written notice provided to the Agent on a Business Day which is: (a) The same day in the case of selection of, or conversion to, or prepayment of, the Base Rate Loans; and (b) at least three (3) Business Days in advance in the case of selection of, conversion to, or renewal or prepayment of, any LIBOR Rate Loans. Standard Notice must be provided no later than 2:00 P.M., Pittsburgh time, on the last day permitted for such notice.

"<u>Subsidiary</u>": A corporation, partnership, trust, limited liability company or other business entity of which more than 50% of the shares of stock or other ownership interests having ordinary voting power (without regard to the occurrence of any contingency) to elect a majority of the board of directors or other managers of such entity are at the time owned, directly, or indirectly through one or more Subsidiaries, or both, by the Borrower.

"<u>Swap Obligation</u>": With respect to any Guarantor, any obligation to pay or perform under any agreement, contract or transaction that constitutes a "swap" within the meaning of section 1a(47) of the Commodity Exchange Act.

"<u>Term Commitment</u>": The commitment of a Lender to make Term Loans to the Borrower pursuant to <u>Section 2.2(a)</u> in an aggregate principal amounts not in excess of the amount set forth opposite such Lender's name on <u>Annex 1</u>, as such amount may be adjusted from time to time in accordance with this Agreement.

"Term Loan": As defined in Section 2.2(a).

"Term Loan Availability Period": The twelve (12) month period commencing on the Closing Date.

"Term Loan Maturity Date": June 30, 2020.

"Term Note": As defined in Section 2.2(b).

1.2 Other Definitional Provisions. As used herein and in any certificate or other document made or delivered pursuant hereto, accounting terms not defined in <u>Section 1.1</u>, and accounting terms partly defined in <u>Section 1.1</u> to the extent not defined, shall have the respective meanings given to them under GAAP. If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and the Borrower or the Agent shall so request, the Lenders and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP; <u>provided</u> that, until so amended, such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein.

ARTICLE II THE LOANS

2.1 The Revolving Loans.

(a) <u>The Facility A Revolving Commitment</u>. Each Lender severally agrees, on the terms and conditions hereinafter set forth, to make loans ("<u>Facility A Revolving Loans</u>") to the Borrower from time to time during the period from the date hereof up to but excluding the Expiration Date in an aggregate amount not to exceed at any time such Lender's Maximum Facility A Revolving Commitment; <u>provided</u> that after giving effect to any Borrowing of Facility A Revolving Loans, the total principal amount of all Facility A Revolving Loans shall not exceed the lesser of (i) the aggregate Maximum Facility A Revolving Commitment of all Lenders and (ii) the Borrowing Base at such time. Within the foregoing limits, the Borrower may borrow, repay pursuant to <u>Section 2.3(b)</u> and reborrow under this Section.

(b) <u>The Facility B Revolving Commitment</u>. Each Lender severally agrees, on the terms and conditions hereinafter set forth, to make loans ("<u>Facility B Revolving Loans</u>") to the Borrower from time to time during the period from the date hereof up to but excluding the Expiration Date in an aggregate amount not to exceed at any time such Lender's Facility B Revolving Commitment. Within the foregoing limits, the Borrower may borrow, repay pursuant to <u>Section 2.3(b)</u> and reborrow under this Section.

(c) <u>Revolving Notes</u>. The Revolving Loans made by each Lender pursuant hereto shall be evidenced by one or more promissory notes of the Borrower, substantially in the form of <u>Exhibit E</u> hereto, or otherwise in form and substance satisfactory to such Lender (as amended, modified, refinanced or restated from time to time, collectively, the "<u>Revolving Notes</u>"), payable to the order of such Lender and representing the obligation of the Borrower to pay the aggregate unpaid principal amount of all Revolving Loans made by such Lender, with interest thereon as prescribed in <u>Section 2.4</u>.

2.2 The Term Loans.

(a) <u>The Term Commitment</u>. Each Lender agrees, on the terms and conditions hereinafter set forth, to make one or more loans ("<u>Term Loans</u>") to the Borrower during the Term Loan Availability Period in an aggregate amount not to exceed such Lender's Term Commitment; <u>provided</u> that after giving effect to any Borrowing of Term Loans, the total principal amount of all Term Loans shall not exceed the lesser of (i) the total Term Commitment, and (ii) one hundred percent (100%) of the invoice amount of equipment purchased (including soft costs (i.e., freights, taxes and installation) of up to a maximum amount of \$500,000 for all such equipment) for the Borrower's U.S. operations no earlier than January 1, 2015, but no later than the end of the Term Loan Availability Period. Once repaid, Term Loans may not be reborrowed.

(b) <u>Term Notes</u>. The Term Loans made by a Lender pursuant hereto shall be evidenced by one or more promissory notes of the Borrower, substantially in the form of <u>Exhibit F</u> hereto, or otherwise in form and substance satisfactory to such Lender (as amended, modified, refinanced or restated from time to time, collectively the "<u>Term Notes</u>"), payable to the order of such Lender and representing the obligation of the Borrower to pay the unpaid principal amount of the Term Loans made by such Lender, with interest thereon as prescribed in <u>Section 2.4</u>.

2.3 Repayment.

(a) <u>Scheduled Repayments</u>.

(i) *Facility A Revolving Loans*. The aggregate principal amount of the Facility A Revolving Loans outstanding on the Expiration Date, together with accrued interest thereon, shall be due and payable in full on the Expiration Date.

(ii) *Facility B Revolving Loans*. Each Facility B Revolving Loan, together with accrued interest thereon, shall be due and payable on the date that is ninety (90) days from the end of the Advance Period in which such Borrowing was made. The aggregate principal amount of the Facility B Revolving Loans outstanding on the Expiration Date, together with accrued interest thereon, shall be due and payable in full on the Expiration Date.

(iii) *Term Loan*. The principal amount of the Term Loan shall be payable in monthly installments payable on the first day of each calendar month, each such installment to be in the principal amount (calculated as a percentage of the Term Loans outstanding at the end of the Term Loan Availability Period) set forth below opposite the period during which such payment date is to occur:

<u>Period</u>	Installment
July 1, 2016 to June 30, 2017	0.8333%
July 1, 2017 to June 30, 2018	1.6667%
July 1, 2018 to June 30, 2019	2.5000%
July 1, 2019 to Term Loan Maturity Date	3.3333%

To the extent not due and payable earlier, the Term Loans, together with accrued interest thereon, shall be payable on the Term Loan Maturity Date. For the avoidance of doubt, no payment of any principal amount of the Term Loan shall be required by this <u>Section 2.3(b)(iii)</u> prior to July 1, 2016.

(b) Mandatory Prepayments.

(i) *Borrowing Base*. If, on any date any Borrowing Base Certificate is required to be furnished pursuant to <u>Section 6.1(e)</u> or <u>(f)</u> hereof, the aggregate principal amount of the Facility A Revolving Loans outstanding hereunder exceeds the Borrowing Base, the Borrower shall immediately prepay a principal amount of the Facility A Revolving Loans in an aggregate amount not less than the amount of such excess.

(ii) *Casualty or Condemnation*. If the Borrower receives any recovery on insurance or condemnation award, it shall prepay Loans in the principal amount and at the time required by the Security Documents, which amount shall be applied <u>first</u>, to the outstanding principal installments of the Term Loans and <u>then</u>, to the Revolving Loans in the inverse order of the maturities thereof; <u>provided</u> that, notwithstanding the foregoing, so long as no Event of Default shall have occurred and be continuing, the Borrower may reinvest all or any portion of such recovery on insurance or condemnation award in similar assets so long as within 180 days following receipt of such recovery on insurance or condemnation award, such reinvestment shall have been consummated (as certified by the Borrower in writing to the Agent); <u>provided further</u>, however, that any recovery on insurance or condemnation award not so reinvested shall be immediately applied to the prepayment of the Loans as set forth in this <u>Section 2.3(b)(ii)</u>. Notwithstanding the anything contained herein to the contrary, the Agent may waive the requirement to make any payment pursuant to this <u>Section 2.3(b)(ii)</u> in its sole discretion.

(iii) *Applicability of Certain Provisions*. Prepayments required by this <u>Section 2.3(b)</u> are subject to all of the terms and conditions applicable to prepayments generally pursuant to <u>Sections 2.3(c)</u> and <u>3.2</u>, except that prepayments under this <u>Section 2.3(b)</u> may be in any principal amount. The Borrower shall not be required to make funding breakage payments under <u>Section 3.9</u> with respect to mandatory prepayments under this <u>Section 2.3(b)</u>.

(c) <u>Optional Prepayments</u>. The Borrower may at its option pay the Loans, in whole or in part, at any time and from time to time, by giving Standard Notice to the Agent, in each case specifying the date and the amount of payment; <u>provided</u>, that prepayments of a LIBOR Rate Loan at a time other than the last day of the applicable Interest Period shall be subject to payment of funding breakage amounts under <u>Section 3.9</u>. Payments which are partial prepayments of the Term Loans shall be applied to the principal installments of the Term Loans of latest maturity.

2.4 Interest Rates. The unpaid principal amount of the Loans shall bear interest for each day until due on the basis of the Base Rate, or the LIBOR Rate as selected by the Borrower or as determined under <u>Section 2.5</u>, <u>3.4</u> or <u>3.8</u> plus, in each case, the Applicable Margin. Interest with respect to each Loan shall be payable in arrears on each Interest Payment Date for such Loan.

2.5 Continuation and Conversion Options. The Borrower may elect from time to time to convert its outstanding Loans from Loans bearing interest at a rate determined by reference to one basis to Loans bearing interest at a rate determined by reference to an alternative basis by giving Standard Notice to the Agent, provided that any conversion of LIBOR Rate Loans shall only be made on the last day of an Interest Period with respect thereto. The Borrower may elect from time to time to continue its outstanding LIBOR Rate Loans upon the expiration of the Interest Period(s) applicable thereto by giving to the Agent Standard Notice and the succeeding Interest Period(s) of such continued Loan or Loans will commence on the last day of the Interest Period of the Loan to be continued. Each Standard Notice electing to convert or continue a Loan shall specify: (i) the proposed conversion/continuation date; (ii) the amount of the Loan to be converted/continued; (iii) the nature of the proposed continuation/conversion; and (iv) in the case of a conversion to, or continuation of a LIBOR Rate Loans, upon expiration of the Interest Period(s) applicable thereto, such Loans shall automatically be continued for Interest Period(s) of the same number of months as such expiring the Interest Period(s). Subject to the limitations set forth in this Section and in the definition of Interest Period, all or any part of outstanding Loans may be converted or continued as provided herein.

ARTICLE III GENERAL PROVISIONS CONCERNING THE LOANS

3.1 Use of Proceeds. The proceeds of the Revolving Loans hereunder shall be used by the Borrower for working capital and general corporate purposes. The proceeds of the Term Loans hereunder shall be used by the Borrower to finance equipment purchases.

3.2 Making the Loans. The Borrower may borrow under the Commitments by providing Standard Notice to the Agent, specifying (A) the type of Loans requested, (B) the amount of the proposed Borrowing, (C) the requested date of the Borrowing (which shall be a Business Day), (D) whether the Borrowing is to consist of LIBOR Rate Loans or a Base Rate Loan, and (E) if the Borrowing is to be a LIBOR Rate Loan, the length of the Interest Period therefor. The Agent shall promptly notify each Lender of the information contained in such Standard Notice and its Applicable Percentage of such Loans. Upon satisfaction of the applicable conditions set forth in <u>Article V</u>, each Lender will make available the proceeds of its Loan to the Agent at the Agent's Office no later than 12 o'clock Noon, Los Angeles, California time, in funds immediately available at the Agent's Office. The Agent's failure to receive Standard Notice of a particular Borrowing shall not relieve the Borrower of its obligations to repay the Borrowing and to pay interest thereon.

3.3 Transactional Amounts. Except as otherwise set forth in this Agreement, every selection of, and conversion from or to, an interest rate option, and every payment or prepayment of a Loan shall be in a principal amount of: (a) in the case of Base Rate Loans, at least \$100,000 or a higher integral multiple of \$100,000, and (b) in the case of LIBOR Rate Loans, at least \$100,000 or a higher integral multiple of \$100,000.

3.4 Post-Maturity Interest and Late Fees.

(a) <u>Default Interest</u>. Notwithstanding anything to the contrary contained in <u>Section 2.4</u>, if an Event of Default has occurred and is continuing, the unpaid principal amount of the Loans and, to the extent permitted by law, interest accrued thereon and any fees, indemnity or other amounts due hereunder shall bear interest at the Default Rate.

(b) <u>Post-Default Interest Options</u>. Notwithstanding <u>Section 2.4</u> and <u>2.5</u>, if an Event of Default has occurred and is continuing, the Agent or the Required Lenders, at its option, may refuse to permit the Borrower to select the LIBOR Rate to thereafter apply to Loans and may convert any outstanding Loan to a Base Rate Loan.

(c) <u>Late Fee</u>. To the extent permitted by Law, the Agent or the Required Lenders shall have the right to assess, and the Borrower shall pay, a late fee if any principal, interest, or fees under this Agreement are not paid within ten (10) days after their due date, and in such a case, the late charge shall be in an amount equal to the greater of Twenty Dollars (\$20.00) or five percent (5%) of the amount not timely paid.

3.5 Computation of Interest and Fees; Determinations by Lender.

(a) <u>Calculations</u>. Interest and fees shall be calculated on the basis of a 360 day year for the actual days elapsed. Any change in the interest rate resulting from a change in the Base Rate or the LIBOR Rate shall become effective as of the opening of business on the day on which such change in the Base Rate or LIBOR Rate shall become effective.

(b) <u>Determination by Agent or Lenders</u>. Each determination of an interest rate, fee, cost, indemnification or other amount by the Agent or the Lenders pursuant to any provision of this Agreement shall be conclusive and binding on the Borrower in the absence of manifest error.

3.6 Payments. The Borrower shall make each payment of principal, interest, fees, indemnity, expenses or other amount hereunder or under any Loan Document, without setoff or counterclaim, not later than 12:00 o'clock, Noon, Los Angeles, California time, on the day when due in Dollars to the Agent at the Agent's Office for the account of the Lenders (subject to <u>Section 2.6</u>), in immediately available funds, without presentment, demand, protest or notice of any kind, all of which are hereby expressly waived, and an action therefor shall immediately accrue, and without setoff, counterclaim, withholding or other deduction of any kind. The Agent shall promptly distribute to each Lender its Applicable Percentage of such payment in like funds as received. Any payment received by the Agent after 12:00 o'clock, Noon, Los Angeles, California time, on any day shall be deemed to have been received on the next succeeding Business Day.

3.7 Payment on Non-Business Days. Whenever any payment to be made hereunder or under the Notes shall be stated to be due on a day which is not a Business Day, such payment may be made on the next succeeding Business Day, and such extension of time shall be included in computing interest or fees, if any, in connection with such payment.

3.8 Inability to Determine Interest Rate; Ineffective Interest Rate. If the (a) the Agent or the Required Lenders shall have determined that by reason of circumstances affecting the Interbank LIBOR market, adequate and reasonable means do not exist for ascertaining the LIBOR Rate or (b) any Lender shall have determined that the making, maintenance or funding of a LIBOR Rate Loan has been made impractical or unlawful, then, and in any such event, the Agent, the Required Lenders or such Lender, as the case may be, may notify the Agent, and the Agent will notify the Borrower of such determination. Upon such date as shall be specified in such notice (which shall not be earlier than the date such notice is given), the obligation of the Lenders to make or maintain Loans at the LIBOR Rate shall be suspended and thereafter during such period all Loans shall be Base Rate Loans, until (i) the Agent or the Required Lenders can once again reasonably ascertain the LIBOR Rate or (ii) such impracticality or unlawfulness shall no longer exist, as applicable.

3.9 Breakage Fees. Except as otherwise expressly set forth herein, upon demand of the Agent or any Lender from time to time, the Borrower shall promptly compensate such Lender for, and hold such Lender harmless from, any loss, cost or expense incurred by it as a result of: (a) any continuation, conversion, payment or prepayment of any LIBOR Rate Loan on a day other than the last day of the Interest Period for such Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise); or (b) any failure by the Borrower to prepay, borrow, continue or convert any LIBOR Rate Loan on the date or in the amount notified by the Borrower, including, without limitation, any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Loan or from fees payable to terminate the deposits from which such funds were obtained. The Borrower shall also pay any customary administrative fees charged by the Agent or such Lender in connection with the foregoing.

3.10 Reserved.

3.11 Calculations. For purposes of calculating amounts payable by the Borrower to the Agent or any Lender under <u>Sections 3.8</u> and <u>3.9</u>, the Agent or such Lender shall be deemed to have funded each LIBOR Rate Loan by a matching deposit or other borrowing in the London interbank Eurodollar market for a comparable amount and for a comparable period, whether or not such Loan was in fact so funded. In determining such amount, the Agent or such Lender may use any reasonable averaging and attribution methods.

3.12 Deposit Account. The Borrower shall maintain its primary domestic banking services, including, without limitation depository and operating accounts, treasury management services and hedging transactions, with the Agent. The Borrower irrevocably authorizes and directs the Agent to charge its primary deposit account with the Agent for any payment due hereunder, including, without limitation, principal, interest, fees, expenses due hereunder.

3.13 Special Funding Provisions.

(a) <u>Funding by Lenders</u>; <u>Presumption by Agent</u>. Unless the Agent shall have received notice from a Lender prior to the proposed date of any Borrowing that such Lender will not make available to the Agent such Lender's share of such Borrowing, the Agent may assume that such Lender has made such share available on such date in accordance with this Agreement and may, in reliance upon such assumption, make available to the Borrower a corresponding amount. In such event, if a Lender has not in fact made its share of the applicable Borrowing available to the Agent, then the applicable Lender and the Borrower severally agree to pay to the Agent forthwith on demand such corresponding amount in immediately available funds with interest thereon, for each day from and including the date such amount is made available to the Borrower to but excluding the date of payment to the Agent, at (A) in the case of a payment to be made by such Lender, the greater of the Federal Funds Rate and the rate determined by the Agent in accordance with banking industry rules on interbank compensation and (B) in the case of a payment to be made by the Borrower, the interest rate applicable Borrowing available to the Agent. If the Borrower shall not be due until ten (10) days after such Lender failed to its share of the applicable Borrowing available to the Borrower the amount of such interest paid by the Borrower for such period. If such Lender pays its share of the applicable Borrowing to the Agent, then the amount so paid shall constitute such Lender's Loan included in such Borrowing. Any payment by the Borrower shall be without prejudice to any claim the Borrower may have against a Lender that shall have failed to make such payment to the Agent.

(b) <u>Payments by Borrower</u>; <u>Presumptions by Agent</u>. Unless the Agent shall have received notice from the Borrower prior to the date on which any payment is due to the Agent for the account of the Lenders hereunder that the Borrower will not make such payment, the Agent may assume that the Borrower has made such payment on such date in accordance herewith and may, in reliance upon such assumption, distribute to the Lenders, the amount due. In such event, if the Borrower has not in fact made such payment, then each of the Lenders severally agrees to repay the Agent forthwith on demand the amount so distributed to such Lender, in immediately available funds with interest thereon, for each day from and including the date such amount is distributed to it to but excluding the date of payment to the Agent, at the greater of the Federal Funds Rate and a rate determined by the Agent in accordance with banking industry rules on interbank compensation. A notice of the Agent to any Lender or the Borrower with respect to any amount owing under this subsection (b) shall be conclusive, absent manifest error.

(c) <u>Failure to Satisfy Conditions Precedent</u>. If any Lender makes available to the Agent funds for any Loan to be made by such Lender as provided in Article II, and such funds are not made available to the Borrower by the Agent because the conditions to the applicable Borrowing set forth in Article V are not satisfied or waived in accordance with the terms hereof, the Agent shall return such funds (in like funds as received from such Lender) to such Lender, without interest.

(d) <u>Obligations of Lenders Several</u>. The obligations of the Lenders hereunder to make Loans, and to make payments pursuant to <u>Section</u> <u>10.5(b)</u> are several and not joint. The failure of any Lender to make any Loan, or to make any payment under <u>Section 10.5(b)</u> on any date required hereunder shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so make its Loan or to make its payment under <u>Section 10.5(b)</u>.

(e) <u>Defaulting Lenders</u>. Notwithstanding any provision of this Agreement to the contrary, if a Lender becomes a Defaulting Lender, then the following provisions shall apply for so long as such Lender is a Defaulting Lender:

(i) Fees shall cease to accrue on the unfunded portion of its Commitments;

(ii) Such Defaulting Lender's unfunded Revolving Commitment and outstanding Revolving Loans shall not be included in determining "Required Lenders";

(iii) The Agent or another Lender may fund such Defaulting Lender's share of any subsequent Borrowing, in which case, the Borrower and the Defaulting Lender shall have the obligations to repay such Borrowing as are provided in <u>Section 3.13(a)</u> with respect to certain other fundings by the Agent; and

(iv) Notwithstanding anything to the contrary contained in this Agreement or any other Loan Document, the Agent, each Lender and the Borrower acknowledge and agree that the Borrower is relying on each Lender to fulfill its funding obligations under this Agreement and the other Loan Documents, that the Borrower will make significant purchase orders and sales commitments based on such reliance and the failure of any Lender to fulfill all of its funding obligations under this Agreement and the other Loan Documents may cause irreparable harm to the Borrower for which the payment of damages may not be a sufficient remedy, therefore, in addition to all other remedies available to the Borrower, each Lender expressly agrees that the Borrower shall have the right to seek specific performance, injunctive relief and all other remedies in equity and at law to compel any Defaulting Lender to fulfill its obligations under this Agreement and all other Loan Documents, including, but not limited to immediately funding to the Borrower the Defaulting Lender's proportionate share of any Loan which Borrower has requested in accordance with the terms of this Agreement.

(f) <u>Funding Source</u>. Nothing herein shall be deemed to obligate any Lender to obtain the funds for any Loan in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain funds for any Loan in any particular place or manner.

ARTICLE IV REPRESENTATIONS AND WARRANTIES

The Borrower represents and warrants to the Agent and the Lenders as follows:

4.1 Organization. The Borrower and each of its Domestic Subsidiaries is duly organized, validly existing and in good standing under the laws of the jurisdiction of its formation, and has all requisite corporate power and authority to own and operate its properties and to carry out its business. The Borrower and each Domestic Subsidiary is duly qualified and in good standing in all jurisdictions where the nature of its business or ownership of property requires such qualification.

4.2 Authorization. The execution, delivery and performance by the Borrower of the Loan Documents, and the making of Borrowings hereunder are within the Borrower's corporate powers and have been duly authorized by all necessary corporate action.

4.3 No Conflict. The execution, delivery and performance by the Borrower of the Loan Documents do not (a) violate the Borrower's or any Domestic Subsidiary's charter, by-laws, partnership agreement, operating agreement or other organizational or governing documents, (b) violate any Law applicable to the Borrower or any Domestic Subsidiary, or (c) result in a breach of or a default under, or result in or require the imposition of a Lien pursuant to any contract binding on the Borrower or any Domestic Subsidiary.

4.4 Governmental Approval. No Governmental Approval is required for the due execution, delivery and performance by the Borrower of the Loan Documents.

4.5 Validity. The Loan Documents are the binding obligations of the Borrower and its Domestic Subsidiaries, enforceable in accordance with their respective terms, except as such enforceability may be limited by bankruptcy, insolvency or other similar laws of general application and equitable principles relating to or affecting creditors' rights.

4.6 Financial Matters.

(a) <u>Financial Statements</u>. The balance sheets of the Borrower and its consolidated Subsidiaries as at December 31, 2014 and March 31, 2015, and the related statements of income, cash flows and changes in stockholders' equity (or comparable statements) of the Borrower and its consolidated Subsidiaries for the fiscal year and fiscal quarter then ended, copies of which have been furnished to the Lenders, fairly present the financial condition of the Borrower and its consolidated Subsidiaries as at such dates and their results of the operations and cash flow for the respective periods ended on such dates, all in accordance with GAAP (except in the case of unaudited statements, for year-end adjustments and the absence of footnotes). Since December 31, 2014, there has been no Material Adverse Effect. The Borrower and its Subsidiaries do not have any contingent obligations or liabilities, for taxes or otherwise, except those that are disclosed in the financial statements referred to above or on <u>Schedule 4.6</u>.

(b) <u>Projections</u>. The Borrower has furnished to the Lenders projections prepared by the Borrower and approved by the Borrower's Board of Directors demonstrating the projected consolidated financial condition and results of operations of the Borrower and its consolidated Subsidiaries for the period commencing on January 1, 2015 and ending on December 31, 2015, which projections are accompanied by a written statement of the assumptions and estimates underlying such projections (the "<u>Projections</u>"). The Projections, as of the date hereof, are reasonable, are made in good faith and represent the Borrower's best judgment as to such matters; it being recognized by the Lenders that such Projections as to future events are not to be viewed as fact and that actual results during the period or periods covered by the Projections may differ from such projected results. To the best knowledge of the Borrower, no facts or circumstances exist which would lead the Borrower to believe that the Projections will not be attained or exceeded.

4.7 Corporate Structure and Ownership. <u>Schedule 4.7</u> sets forth the names of the record and beneficial owners of all Equity Interests of each Subsidiary and the amount thereof owned by each of them. All of such Equity Interests are duly authorized, validly issued and are fully paid and nonassessable. Except as set forth on <u>Schedule 4.7</u>, there are no voting arrangements, restrictions on transfer or other arrangements that pertain to the Equity Interests of the Borrower or any Subsidiary.

4.8 Partnerships. Neither the Borrower nor any Subsidiary of the Borrower is a partner of a partnership or a party to a joint venture or otherwise has an obligation to make capital contributions to, or be generally liable for or on account of, the debts or liabilities of any other Person.

4.9 Insurance. The properties of the Borrower and its Domestic Subsidiaries are insured with financially sound and reputable insurance companies not Affiliates of the Borrower, in such amounts, with such deductibles and covering such risks as are customarily carried by companies engaged in similar businesses and owning similar properties in similar locations, and the Borrower maintains the insurance required by <u>Section 6.5</u>.

4.10 Litigation. Except as set forth on <u>Schedule 4.10</u> hereto, there is no pending or threatened action or proceeding affecting the Borrower or any of its Domestic Subsidiaries before any Governmental Authority, which, in the case of any such action or proceeding commenced or threatened after the Closing Date, individually or in the aggregate, has had or could reasonably be expected to have a Material Adverse Effect.

4.11 Employee Benefit Plans. The Borrower and each of its ERISA Affiliates is in compliance in all material respects with any applicable provisions of ERISA and the regulations and published interpretations thereunder with respect to all Employee Benefit Plans. No ERISA Event has occurred or is reasonably expected to occur with respect to any Pension Plan. Neither the Borrower nor any of its ERISA Affiliates has or presently contributes to a Multiemployer Plan. No assets of an Employee Benefit Plan will be used to repay or secure any Loan or be involved in any way with, and no "prohibited transaction" as defined in ERISA or the Code shall occur as a result of, the transactions contemplated by this Agreement.

4.12 Environmental Matters. The Borrower has not received any written communication from any Governmental Authority alleging potential liability or responsibility for violation of any Environmental Law or release or injury to the environment or responsibility for remediation actions. Except as set forth on <u>Schedule 4.12</u>, the Borrower and its Domestic Subsidiaries are in compliance with all Environmental Laws and no event or condition has occurred or is occurring with respect to Borrower or any of its Domestic Subsidiaries relating to any Environmental Law that has resulted in or could reasonably be expected to result in such claims, or is or could reasonably be expected to be the subject of any investigation, proceeding, settlement, except violations and claims that, individually or in the aggregate, have not had and could not reasonably be expected to have a Material Adverse Effect. The Borrower and its Domestic Subsidiaries have all Governmental Approvals relating to environmental matters necessary for the ownership and operation of their respective properties and businesses as presently owned and operated and as presently proposed to be owned and operated, except for those the absence of which, individually, or in the aggregate, has not had and could not reasonably be expected to have a Effect. Neither the Borrower nor its Domestic Subsidiaries has transported or arranged for the transport of any materials subject to Environmental Laws to any environmental clean-up site.

4.13 Title to Properties; Liens. The Borrower and its Domestic Subsidiaries have (a) good, sufficient and legal title to (in the case of fee interests in real property), (b) valid leasehold interests in (in the case of leasehold interests in real or personal property), or (c) good title to (in the case of all other personal property), all of their respective properties and assets reflected in the financial statements referred to in <u>Section 4.6</u> or in the most recent financial statements delivered pursuant to <u>Section 6.1</u>, in each case except for assets disposed of since the date of such financial statements in the ordinary course of business or as otherwise permitted under <u>Section 7.8</u>. All such properties and assets are free and clear of Liens, other than Permitted Liens.

4.14 Payment of Taxes. Except to the extent permitted by <u>Section 6.4</u>, all tax returns and reports of the Borrower and its Domestic Subsidiaries required to be filed by any of them have been timely filed, and all taxes shown on such tax returns to be due and payable and all assessments, fees and other governmental charges upon the Borrower and its Domestic Subsidiaries and upon their respective properties, assets, income, businesses and franchises that are due and payable have been paid when due and payable. The Borrower knows of no proposed tax assessment against the Borrower or any of its Domestic Subsidiaries.

4.15 Governmental Regulation. Neither the Borrower nor any of its Subsidiaries is subject to regulation under the Federal Power Act, the Interstate Commerce Act or the Investment Company Act of 1940 or under any other federal or state statute or regulation which may limit its ability to incur Debt or which may otherwise render all or any portion of the Obligations unenforceable.

4.16 Governmental Approval, Intellectual Property, Etc. Except as disclosed in <u>Schedule 4.16</u>, (a) the Borrower and its Domestic Subsidiaries own or possess all Governmental Approvals and Intellectual Property necessary for the operation of their businesses, without known conflict with the rights of others; (b) no product or process of the Borrower or its Subsidiaries violates or infringes any Governmental Approval or Intellectual Property owned by any other Person; and (c) there is no violation by any Person of any right of the Borrower or any of its Subsidiaries with respect to any Intellectual Property owned or used by the Borrower or any of its Subsidiaries except, with respect to clauses (a) and (b), for matters that, individually or in the aggregate, have not had and could not reasonably be expected to have a Material Adverse Effect. <u>Schedule 4.16</u> lists all of the Borrower's and Domestic Subsidiaries' registered Intellectual Property.

4.17 Labor Disputes and Casualties. Neither Borrower nor any Subsidiary is affected by any fire, explosion, accident, strike, lockout, or other labor dispute, drought, storm, hail, earthquake, embargo, act of public enemy, or other casualty (whether or not covered by insurance) which, individually or in the aggregate, has had or could be reasonably expected to have a Material Adverse Effect.

4.18 Compliance. Neither the Borrower nor any Subsidiary is in default in the performance of any agreement or instrument to which it may be a party or by which its properties may be bound, or in violation of any Law, which defaults and violations, individually or in the aggregate, have had or could reasonably be expected to have a Material Adverse Effect.

4.19 Margin Stock. Neither the Borrower nor any Subsidiary is engaged in, and does not have as one of its substantial activities, the business of extending or obtaining credit for the purpose of purchasing or carrying "margin stock" (as that term is defined in Regulation U of the Board of Governors of the Federal Reserve System), and no proceeds of any Borrowing have been or will be used for such purpose or for the purpose of purchasing or carrying any shares of margin stock. None of the Collateral consists of margin stock.

4.20 Personal Property Collateral Matters.

(a) <u>Names and Organization</u>. The Borrower and each Subsidiary's name as it appears in official filings in the state of its organization, type of organization, jurisdiction of organization, organization number provided by the applicable Government Authority, and chief executive office are set forth on <u>Schedule 4.20</u>. Neither the Borrower nor any Domestic Subsidiary (or predecessor by merger or otherwise) has, within the four-month period preceding the date hereof, (or, in the case of an additional Domestic Subsidiary under <u>Section 6.10</u>, the date it becomes a Guarantor) had a different name from the name of such Person listed on the signature pages hereof, except as set forth on <u>Schedule 4.20</u>.

(b) <u>First Priority Lien</u>. The Security Agreement creates a valid security interest in the Collateral in favor of the Agent, for the benefit of the Lender Parties, securing the Secured Obligations (as defined therein), which security interest has been duly perfected (except to the extent that a security interest in Equity Interests of any Foreign Subsidiary cannot perfected by possession and control) and is prior to all other Liens, except for Permitted Liens. For the avoidance of doubt, "Collateral" shall not include Excluded Collateral or any assets of a Foreign Subsidiary. All filings and other actions necessary or desirable to perfect and protect such security interest in favor of the Agent have been duly made and taken, except (i) for the filing of UCC financing statements made by the Agent required in order to perfect the security interests in the Collateral, (ii) to the extent that a security interest of any Foreign Subsidiary or any foreign deposit account or securities account is not perfected by possession and control.

(c) <u>Possession or Control of Certain Collateral</u>. Except as set forth in <u>Schedule 4.20</u>, the Borrower and its Domestic Subsidiaries have exclusive possession and control of the Equipment and Inventory (in each case as defined in the Security Agreement) located in the United States (except for Inventory at a location for which either (x) a warehouseman's or bailee's agreement in form and substance satisfactory to the Agent, in its sole discretion, has been entered into or (y) an adequate rent reserve in an amount determined by the Agent, in its sole discretion, has been established; provided that the foregoing representation need not be true with respect to Inventory that is designated to be sold by Amazon or Cisco prior to the entry into a warehouseman's or bailee's agreement with respect to the location thereof so long as the Borrower is in compliance with <u>Section 6.11</u>). The Borrower and its Subsidiaries have delivered to the Agent, for the benefit of the Lender Parties, possession of all originals of all promissory notes or other instruments, stock certificates, chattel paper and negotiable documents constituting Collateral. None of the Accounts (as defined in the Security Agreement) is evidenced by a promissory note or other instrument, chattel paper or negotiable document. The Borrower and its Domestic Subsidiaries only maintain deposit accounts and securities accounts with the Agent or accounts covered by control agreements approved by the Agent, other than deposit accounts and securities accounts located in Taiwan and set forth on <u>Schedule 4.20</u> so long as the Borrower is in compliance with the requirement set forth in <u>Section 3.12</u> hereof.

4.21 USA PATRIOT Act, OFAC and Other.

(a) No Loan Party, any of its Subsidiaries or any of the Affiliates or respective officers, directors, brokers or agents of such Loan Party, Subsidiary or Affiliate (i) has violated any Anti-Terrorism Laws or (ii) has engaged in any transaction, investment, undertaking or activity that conceals the identity, source or destination of the proceeds from any category of prohibited offenses designated by the Organization for Economic Co-operation and Development's Financial Action Task Force on Money Laundering.

(b) No Loan Party, any of its Subsidiaries or any of the Affiliates or respective officers, directors, employees, brokers or agents of such Loan Party, Subsidiary or Affiliate is a Person that is, or is owned or controlled by Persons that are: (i) the subject of any Sanctions, or (ii) located, organized or resident in a country or territory that is, or whose government is, the subject of Sanctions.

(c) No Loan Party, any of its Subsidiaries or any of the Affiliates or respective officers, directors, brokers or agents of such Loan Party, Subsidiary or Affiliate acting or benefiting in any capacity in connection with the Loans (i) conducts any business or engages in making or receiving any contribution of goods, services or money to or for the benefit of any Person, or in any country or territory, that is the subject of any Sanctions, (ii) deals in, or otherwise engages in any transaction related to, any property or interests in property blocked pursuant to any Anti-Terrorism Law or (iii) engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law.

4.22 Solvency. The Borrower and each of its Subsidiaries are and, upon the incurrence of any Obligations by Borrower on any date on which this representation is made or restated, will be, solvent within the meaning of applicable Laws relating to fraudulent conveyances.

4.23 Disclosure. No financial or other information, exhibit or report furnished to the Agent or the Lenders by or on behalf of the Borrower or any of its Subsidiaries for use in connection with the transactions contemplated by this Agreement contains any untrue statement of a material fact or omits to state a material fact (known to the Borrower, in the case of any document not furnished by it) necessary in order to make the statements contained therein not misleading in light of the circumstances in which the same were made.

ARTICLE V CONDITIONS OF LENDING

5.1 Conditions Precedent to Initial Loans. The obligation of the Lenders to make the initial Loans is subject to the following conditions

(a) Loan Documents. The Agent shall have received the following, in form and substance satisfactory to the Lenders:

(i) The Notes executed by the Borrower; and

precedent:

(ii) Copies of all Loan Documents (not otherwise specifically identified in this <u>Section 5.1</u>) executed by the Borrower.

(b) <u>Corporate Action</u>. The Agent shall have received the following, each dated the Closing Date:

(i) Copies of the Certificate of Incorporation of the Borrower, certified as of a recent date by the Secretary of State of its state of organization and a good standing certificate (or equivalent) from such state;

(ii) Copies of (A) the bylaws of the Borrower, and (B) resolutions of the Board of Directors or other authorizing documents of the Borrower, in form and substance satisfactory to the Agent, approving the Loan Documents and the Borrowings hereunder, certified by the Secretary or an Assistant Secretary of the Borrower;

(iii) An incumbency certificate executed by the Secretary or an Assistant Secretary of the Borrower, certifying the names and signatures of the officers of the Borrower authorized to sign the Loan Documents; and

(iv) An Officer's Certificate certifying as to the matters set forth in Section 5.2(a), (b) and (c).

(c) Financial Matters. The Borrower shall have provided to the Lenders the financial statements and Projections referred to in Section 4.6.

(d) Other. The Agent shall have received:

(i) Evidence that all insurance required to be maintained pursuant to the Loan Documents has been obtained and is in effect;

(ii) Evidence that the Existing Line of Credit Agreement and all commitments thereunder have been or are concurrently being terminated, and all outstanding obligations thereunder have been or are concurrently being paid in full or terminated; and

(iii) A favorable opinion of counsel to the Borrower, covering such corporate, perfection and enforceability matters as the Agent may reasonably request.

(e) Security Matters and Documents. The Agent shall have received:

(i) A copy of the Security Agreement executed by the Borrower, together with: (A) acknowledgement copies (or other evidence of filing satisfactory to the Agent) of proper financing statements duly filed under the Uniform Commercial Code (or any equivalent or similar legislation) of all jurisdictions as may be necessary or, in the Agent's opinion, desirable to effectively perfect the interests in the personal property and fixtures granted under the security agreement(s); (B) control agreements satisfactory to the Agent with respect to all deposit or securities accounts included in the Collateral, including, without limitation such accounts with Comerica Securities; (C) possession of all certificated securities (with undated stock powers) and instruments included in the Collateral, including, without limitation, the Global Intercompany Note; and (D) subject to <u>Section 6.11</u>, evidence satisfactory to the Agent that all other filings, recordings, landlord consents and waivers and other actions the Agent deems necessary or advisable to establish, preserve and perfect the Liens granted to the Agent in personal property shall have been made or obtained;

(ii) UCC, tax, and lien searches from all domestic locations requested by Agent, together with lien termination documents satisfactory to Agent terminating all liens shown on such searches that are not Permitted Liens; and

(iii) Copies of the Mortgage executed by the Borrower, together with opinions from local counsel with respect to the enforceability and validity of the Mortgage as the Agent may request.

(f) <u>Fees, Expenses, etc</u>. All fees, expenses and other compensation required to be paid to the Lender Parties pursuant hereto or pursuant to any other written agreement on or prior to the Closing Date shall have been paid or received.

(g) <u>Deposit Account.</u> The Borrower shall have established a deposit account with the Agent which will be subject to the provisions of <u>Section 3.12</u>.

(h) <u>Borrowing Base Certificate and Collateral Report</u>. The Borrower shall furnish to the Agent a certificate Borrowing Base Certificate for the month most recently ended prior to the Closing Date, together with a report, in form and level of detail reasonably satisfactory to the Agent, including (i) accounts receivable aging and reconciliations, (ii) accounts payable aging and (iii) inventory report and reconciliations.

(i) <u>General</u>. All corporate and legal proceedings and all instruments and documents in connection with the transactions contemplated by this Agreement shall be reasonably satisfactory in content, form and substance to the Lenders and their counsel, and the Lenders' counsel shall have received any and all further information and documents which the Lender or such counsel may reasonably have requested in connection therewith.

5.2 Conditions Precedent to Each Borrowing. The obligation of the Lenders to make any Loan (including the initial Borrowing) shall be subject to the following additional conditions precedent:

(a) <u>Representations</u>. The representations and warranties contained in <u>Article IV</u> or any other Loan Document are correct when made and on and as of the date of such Borrowing as though made on and as of such date.

(b) <u>No Default</u>. No (i) Default of which the Borrower has knowledge or (ii) Event of Default has occurred and is continuing, or would result from such Borrowing.

(c) <u>Material Adverse Effect</u>. Since December 31, 2014, there shall not have occurred, or to the knowledge of the Borrower any Material Adverse Effect.

(d) Standard Notice. Standard Notice of such Borrowing shall have been delivered to the Agent.

(e) <u>Invoices</u>. Solely with respect to Borrowings of Term Loans, the Agent shall have received copies of invoices for equipment purchased (including soft costs to the extent permitted by <u>Section 2.2(a)</u>) in amounts permitted hereunder or otherwise satisfactory to the Agent to support the request for such Borrowing.

Each request for a Loan submitted by the Borrower, under this Agreement shall be deemed to be a representation and warranty that the foregoing conditions have been satisfied on and as of the date of the Borrowing.

ARTICLE VI COVENANTS

So long as any Obligation shall remain unpaid or the Lenders shall have any Commitment hereunder, the Borrower will, unless the Required Lenders shall otherwise consent in writing:

6.1 Financial Information. Furnish to the Agent:

(a) as soon as available, but in any event within ninety (90) days after the end of each fiscal year of the Borrower, the audited consolidated balance sheet of the Borrower and its consolidated Subsidiaries as at the end of such fiscal year and the related consolidated statements of income, cash flows and changes in stockholders' equity (or comparable statement) for such year, setting forth in each case in comparative form the figures for the previous year, accompanied by an unqualified report and opinion thereon of independent certified public accountants acceptable to the Agent;

(b) as soon as available, but in any event within forty-five (45) days after the end of each fiscal quarter of the Borrower, an unaudited consolidated balance sheet of the Borrower and its consolidated Subsidiaries as at the end of such month and the related unaudited consolidated statements of income, cash flows and changes in stockholders' equity (or comparable statement) for such fiscal quarter and year to date, setting forth in each case in comparative form for the previous fiscal year, certified by the chief financial officer of the Borrower as fairly presenting the financial condition of the Borrower and its consolidated Subsidiaries and their results of operation, cash flow and changes in financial position (subject to year end adjustments);

(c) together with each delivery of financial statements pursuant to clause (b) above, an Officers' Certificate in the form of <u>Exhibit A</u> certifying as to the matters set forth therein and demonstrating in reasonable detail compliance at the end of such accounting periods with the restrictions contained in <u>Section 7.1</u>;

(d) as soon as practicable, and in any event within sixty (60) days after the end of each fiscal year, a budget and projections (as were approved by the Board of Directors of the Borrower) by fiscal quarter for the next four fiscal quarters, including projected consolidated balance sheets and statements of income and retained earnings (or comparable statements) and cash flow of the Borrower and its consolidated Subsidiaries, all in form and detail acceptable to the Agent;

(e) <u>Borrowing Base Certificates</u>. (i) Within thirty (30) days of the end of each month when the aggregate amount of Revolving Loans outstanding is (x) in excess of \$10,000,000 or (y) in less than \$10,000,000, but the Current Ratio is less than 1.25 to 1.00, or (ii) on the date of each request for a Loan when, after giving effect to such Borrowing, any of the foregoing circumstances would exist, the Borrower shall furnish to the Agent a Borrowing Base Certificate together with a report, in form and level of detail reasonably satisfactory to the Agent, including (A) accounts receivable aging and reconciliations; (B) accounts payable aging and (C) inventory report and reconciliations;

(f) <u>Quarterly Collateral Reports</u>. Within thirty (30) days of the end of each fiscal quarter of the Borrower, the Borrower shall deliver to the Agent a Borrowing Base Certificate together with a report, in form and level of detail reasonably satisfactory to the Agent, including (i) accounts receivable aging and reconciliations, (ii) accounts payable aging and (iii) inventory report and reconciliations; and

(g) promptly upon request, any other financial statements, reports or information with respect to the Borrower or any Subsidiary reasonably requested by the Agent or any Lender.

All financial statements delivered pursuant to this <u>Section 6.1</u> are to be complete and correct in all material respects and to be prepared in reasonable detail acceptable to the Agent and in accordance with GAAP.

6.2 Notices and Information. Deliver to the Agent:

(a) promptly upon the Borrower obtaining knowledge (i) of the occurrence of a Default or Event of Default, (ii) that any Person has given any notice to the Borrower or any Subsidiary of the Borrower or taken any other action with respect to a claimed cross-default of the type referred to in <u>Section 8.1(f)</u>, (iii) of the institution of, or any materially adverse development in, any litigation involving an alleged liability (including possible forfeiture of property) of the Borrower or any of its Domestic Subsidiaries greater than \$500,000, in the aggregate, (iv) of any material casualty to its assets resulting in a loss in excess of \$500,000, in the aggregate, or (v) of a condition or events that could reasonably be expected to cause a Material Adverse Effect, an Officer's Certificate specifying the nature and period of existence of any such condition or event, and what action the Borrower, is taking with respect thereto;

(b) promptly upon any officer of the Borrower becoming aware of the occurrence of or forthcoming occurrence of any (i) ERISA Event, or (ii) "prohibited transaction," as such term is defined in Section 4975 of the Code or Section 406 of ERISA, in connection with any Employee Benefit Plan or any trust created thereunder, an Officer's Certificate specifying the nature thereof, what action the Borrower has taken, is taking or proposes to take with respect thereto, and, when known, any action taken or threatened by the Internal Revenue Service, the Department of Labor, or the Pension Benefit Guaranty Corporation with respect thereto;

(c) with reasonable promptness following receipt thereof by the Borrower, copies of (i) all notices received by the Borrower or any of its ERISA Affiliates of the Pension Benefit Guaranty Corporation's intent to terminate any Pension Plan or to have a trustee appointed to administer any Pension Plan; (ii) each Schedule B (Actuarial Information) to the annual report (Form 5500 Series) filed by the Borrower or any of its ERISA Affiliates with the Internal Revenue Service with respect to each Pension Plan; and (iii) all notices received by the Borrower or any of its ERISA Affiliates from a Multiemployer Plan sponsor concerning the imposition or amount of withdrawal liability pursuant to Section 4202 of ERISA;

(d) promptly, and in any event within thirty (30) days following receipt thereof by the Borrower, a copy of any notice, summons, citation, directive, letter or other form of written communication from any Governmental Authority or court in any way concerning any alleged violation on the part of the Borrower or any of its Domestic Subsidiaries in connection with any substance defined as toxic or hazardous by any applicable Environmental Law or any waste or by product thereof, or concerning the filing of a Lien upon, against or in connection with the Borrower, its Domestic Subsidiaries, or any of their leased or owned real or personal property, in connection with a Hazardous Substance Superfund or a Post-Closure Liability Fund as maintained pursuant to Section 9507 of the Code; and

(e) promptly, and in any event within ten (10) days after request, such other information and data with respect to the Borrower or any of its Subsidiaries as from time to time may be reasonably requested by the Agent or any Lender.

6.3 Corporate Existence, Etc. At all times preserve and keep in full force and effect its and its Subsidiaries' corporate existence, rights, franchises and licenses material to its business and those of each of its Subsidiaries.

6.4 Payment of Obligations. Pay and discharge as the same shall become due and payable, all its obligations and liabilities, including (a) all tax liabilities, assessments and governmental charges or levies upon it or its properties or assets, unless the same are being contested in good faith by appropriate proceedings diligently conducted and adequate reserves in accordance with GAAP are being maintained by the Borrower or such Domestic Subsidiary; (b) all lawful claims which, if unpaid, would by Law become a Lien upon its property; and (c) all Debt, as and when due and payable, but subject to any subordination provisions contained in any instrument or agreement evidencing such Debt.

6.5 Maintenance of Properties. Maintain or cause to be maintained in good repair, working order and condition all material properties used or useful in the business of the Borrower and its Domestic Subsidiaries and from time to time will make or cause to be made all appropriate repairs, renewals and replacements thereof.

6.6 Insurance. Maintain or cause to be maintained, with financially sound and reputable insurers, insurance with respect to its properties and business and the properties and business of its Domestic Subsidiaries against loss or damage of the kinds customarily insured against by entities of established reputation engaged in the same or similar businesses and similarly situated, of such types and in such amounts as are customarily carried under similar circumstances by such other businesses and as is reasonably acceptable to the Agent and all additional insurance required by the Security Documents, including, as applicable and without limitation (i) "all-risk" fire and extended coverage hazard insurance in an amount not less than 100% of the full insurable replacement value of the real property of the Borrower and its Domestic Subsidiaries and its contents, (ii) comprehensive general public liability insurance and (iii) business interruption insurance, all such policies of insurance shall insure the Lenders as their interest may appear, shall bear a long-form lender's loss payable endorsement, shall list the Agent as an additional insureds with respect to liability coverages and shall require thirty (30) days' notice of cancellation or material change endorsements in favor of the Agent.

6.7 Inspection. Permit any authorized representatives designated by any Lender and at the expense of such Lender to visit and inspect any of the properties of the Borrower or any of its Subsidiaries, including its and their financial and accounting records, and to make copies and take extracts therefrom, and to discuss its and their affairs, finances and accounts with its and their officers, members, employees, representatives and independent public accountants, all at such reasonable times during normal business hours and as often as may be reasonably requested; <u>provided</u>, that when an Event of Default exists, the foregoing shall be at the expense of the Borrower. Without limiting the generality of the foregoing, the Borrower will, and will cause each other Loan Party to, do all things reasonably necessary to permit the Lender to conduct, at Borrower's expense, (i) one collateral field examinations each year covering the Borrower's and its Domestic Subsidiaries accounts receivable, accounts payable and inventory, and (ii) any number of such collateral field examinations if an Event of Default has occurred and is continuing or exists.

6.8 Compliance with Laws, Etc. Exercise, and cause each of its Domestic Subsidiaries to exercise, all due diligence in order to comply with the requirements of all applicable Laws, including, without limitation, all Environmental Laws, noncompliance with which has had or could reasonably be expected to cause, either individually or in the aggregate, a Material Adverse Effect.

6.9 Books and Records. Maintain proper records and accounts in which full, true and correct entries in conformity with GAAP, consistently applied shall be made of all financial transactions and matters involving the assets and business of the Borrower and its Domestic Subsidiaries.

6.10 Additional Subsidiaries. Notify the Agent at the time that any Person becomes a Subsidiary, and, in the case of any Domestic Subsidiary, promptly thereafter (and in any event within thirty (30) days) cause such Person to (a) become a Guarantor by executing and delivering to the Agent a Guaranty, (b) grant to the Agent a Lien on its assets by executing and delivering a supplement to the Security Agreement and such mortgages as the Agent may request; and (c) deliver to the Agent documents comparable to those delivered pursuant to <u>Section 5.1</u> with respect to other Loan Parties and other collateral, all in form and substance reasonably satisfactory to the Agent.

6.11 Post-Closing Matters. Take all necessary actions, and cause each of its Subsidiaries to take all necessary actions, to satisfy the requirements set forth on <u>Schedule 6.11</u>, within such periods as specified therein.

ARTICLE VII NEGATIVE COVENANTS

So long as any Obligation shall remain unpaid or the Lenders shall have any Commitment hereunder, the Borrower will not, without the written consent of the Required Lenders:

7.1 Financial Covenants

(a) <u>Minimum Consolidated EBITDA</u>. Permit Consolidated EBITDA for the period of four fiscal quarters then ended to be less than: (i) \$11,000,000, at the end of any fiscal quarter ending on or after January 1, 2015 and on or before December 31, 2015, (ii) \$12,500,000, at the end of any fiscal quarter ending on or after January 1, 2016 and on or before December 31, 2017, or (iii) the amount determined by the Agent on an annual basis, based on the most recent projections delivered pursuant to <u>Section 6.1(d)</u>, but in any event, no less than \$12,500,000, at the end of any fiscal quarter thereafter.

(b) <u>Leverage Ratio</u>. Permit the ratio of Consolidated Funded Debt at the end of any fiscal quarter to Consolidated EBITDA for the period of four fiscal quarters then ended to exceed: (i) 5.00 to 1.00, at the end of any fiscal quarter ending on or after January 1, 2015 and on or before December 31, 2017, (ii) 4.50 to 1.00, at the end of any fiscal quarter ending on or after January 1, 2018 and on or before December 31, 2019, or (iii) 4.00 to 1:00, at the end of any fiscal quarter thereafter.

(c) <u>Current Ratio</u>. As of the end of any fiscal quarter of the Borrower, permit the Current Ratio to be less than 1.25 to 1.00.

(d) <u>Capital Expenditures</u>. Permit Capital Expenditures of the Borrower and its Domestic Subsidiaries (including any Capital Leases) to exceed: (i) \$30,000,000, in the fiscal year ending on or before December 31, 2015, (ii) \$20,000,000, in the fiscal year ending December 31, 2016, or (iii) the amount reasonably determined by the Agent on an annual basis, based on the most recent projections delivered pursuant to <u>Section 6.1(d)</u>, in any fiscal year ending thereafter; <u>provided</u>, that solely for the purpose of this <u>Section 7.1(d)</u>, equipment purchased or acquired with net cash proceeds of the issuance Equity Interests of the Borrower or its Domestic Subsidiaries after the Closing Date shall not count against the foregoing limit so long as, with respect to such purchase or acquisition, (x) the aggregate consideration therefor does not exceed 50% of such net cash proceeds, (y) immediately before and after giving effect thereto, no Event of Default exists or is continuing, and (z) the Borrower has provided updated financial projects for the following four fiscal quarter period demonstrating that no violation of the covenants set forth in <u>Section 7.1</u> is anticipated as a result thereof. Notwithstanding anything contained in this subsection (d) or elsewhere in this Agreement to the contrary, the aggregate amount of consideration that is permitted by clause (x) above and the Investments permitted by <u>Section 7.7(j)</u> related to the net proceeds of any sale or issuance of Equity Interests after the Closing Date shall not exceed, in the aggregate, 60% of such proceeds.

7.2 Liens, Etc. Create or suffer to exist, or permit any of its Domestic Subsidiaries to create, incur or suffer to exist, any Lien upon or with respect to any of its assets or properties (other than Excluded Collateral), whether now owned or hereafter acquired, or assign, or permit any of its Domestic Subsidiaries to assign, any right to receive income (other than from Excluded Collateral), to or in favor of any Person, except (a) Liens in favor of the Agent, on behalf of the Lenders under the Loan Documents; (b) Liens in favor of East West Bank permitted by the Intercreditor Agreement securing Debt permitted by Section 7.3(c); (c) Liens reflected on the financial statements referred to in Section 4.6 and other Liens existing on the date hereof and set forth in Schedule <u>7.2</u>; (d) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like liens arising in the ordinary course of business which are not overdue by more than 30 days; (e) easements, rights of way, restrictions and similar encumbrances affecting real property which, in the aggregate are not substantial in amount, and which do not materially detract from the value of, or materially interfere with the use of, the property; (f) purchase money liens upon or in any property acquired or held by the Borrower or any Subsidiary in the ordinary course of business to secure the purchase price of such property or to secure indebtedness incurred solely for the purpose of financing the acquisition of such property; and (g) Liens securing Permitted Foreign Debt in an aggregate amount not to exceed \$8,000,000 at any time.

7.3 Debt. Create or suffer to exist, or permit any of its Domestic Subsidiaries to create or suffer to exist, any Debt, other than (a) Debt reflected on the Borrower's financial statements referred to in Section 4.6 which is not being repaid with the proceeds of the Loans, and other Debt existing on the date hereof and set forth on Schedule 7.3 hereto and any refinancings, refundings, renewals or extensions thereof; provided, that the amount of such Debt is not increased and its terms are not less advantageous to the Borrower or its Subsidiaries; (b) Debt owed to the Lender Parties under the Loan Documents; (c) Debt owed to East West Bank pursuant to the Existing EWB Loan Documents and permitted by the Intercreditor Agreement; (d) Debt relating to Liens permitted under Sections 7.2 (f); (e) guarantees permitted by Section 7.7; (f) Debt of a Domestic Subsidiary to another wholly-owned Domestic Subsidiary or to the Borrower; (g) Capital Leases in an aggregate amount not to exceed \$5,000,000 at any time outstanding; provided, that the Borrower has provided updated financial projects for the following four fiscal quarter period demonstrating that no violation of the covenants set forth in Section 7.1 is anticipated as a result of the incurrence of any such Capital Lease; (h) obligations (contingent or otherwise) of the Borrower or any of its Domestic Subsidiaries existing or arising under any Hedging Contract, provided that (i) such obligations are (or were) entered into by such Person in the ordinary course of business for the purpose of directly mitigating risks associated with liabilities, commitments, investments, assets, or property held or reasonably anticipated by such Person, or changes in the value of securities issued by such Person, and not for purposes of speculation or taking a "market view" and (ii) such Hedging Contract does not contain any provision exonerating the non-defaulting party from its obligation to make payments on outstanding transactions to the defaulting party; (i) Permitted Foreign Debt; and (j) convertible debt of the Borrower, provided that (i) the Borrower is in pro forma compliance with the covenants contained in Section 7.1 hereof immediately prior to and would be immediately after the incurrence of any such Debt (without taking into account any cure periods), and (ii) any such convertible Debt is subordinated to Obligations hereunder on terms acceptable to the Agent in its sole discretion.

7.4 Lease Obligations. Create or suffer to exist, or permit any of its Domestic Subsidiaries to create or suffer to exist, any obligations for the payment of rent for any property under leases or agreements to lease (other than Capital Leases) which would cause the direct or contingent liabilities of the Borrower and its Subsidiaries, on a consolidated basis, in respect of all such obligations to exceed \$1,000,000 payable in any fiscal year of the Borrower.

7.5 Equity Payments, Etc. Declare or pay any dividends, purchase or otherwise acquire for value any of its Equity Interests, or make any distribution of assets to its equity holders as such, or permit any of its Subsidiaries to purchase or otherwise acquire for value any Equity Interests of the Borrower, except that the Borrower may declare and deliver dividends and distributions payable in Equity Interests of the Borrower.

7.6 Fundamental Changes. (a) Change its corporate structure; (b) consolidate with or merge into any other corporation or entity, or acquire a substantial portion of the assets, business or Equity Interests of another Person; or (c) liquidate, windup or dissolve; except, a Subsidiary may merge with and into the Borrower.

7.7 Loans, Investments, Contingent Liabilities. Make or permit to remain outstanding, or permit any Domestic Subsidiary to make or permit to remain outstanding, any loan or advance to, or guarantee, induce or otherwise become contingently liable, directly or indirectly, in connection with the obligations, stock or dividends of, or own, purchase or acquire any stock, obligations or securities of or any other interest in, or make any capital contribution to, any other Person ("Investments"), except that the Borrower and its Domestic Subsidiaries may: (a) allow to remain outstanding Investments reflected on the Borrower's financial statements referred to in Section 4.6 and other Investments existing on the date hereof and set forth on Schedule 7.7; (b) own, purchase or acquire certificates of deposit issued by a Lender or any U.S. bank having capital and surplus in excess of \$500,000,000, commercial paper rated Standard & Poor's A-1 or Moody's P-1, direct obligations of the United States of America or its agencies, and obligations guaranteed by the United States of America; (c) continue to own the existing capital stock of the Borrower's Subsidiaries; (d) endorse negotiable instruments for deposit or collection or similar transactions in the ordinary course of business; (e) acquire and own stock, obligations or securities received from customers in connection with debts created in the ordinary course of business owing to the Borrower or a Subsidiary; (f) make or permit to remain outstanding advances to officers, directors and employees of the Borrower and its Subsidiaries in any aggregate amount not to exceed \$1,000,000 at any time outstanding for travel, entertainment, relocation and similar ordinary expenses; (g) reserved; (h) Investments representing Debt permitted by Section 7.3(f); (i) make or permit to remain outstanding loans and advances to any other Persons or enter into or permit to remain outstanding guarantees in connection with the obligations of any other Persons, in an aggregate amount for all such loans, advances and guarantees not exceeding \$1,000,000; and (j) Investments in Foreign Branches and Foreign Subsidiaries in the form of cash or cash equivalents in an aggregate amount not to exceed at any time \$5,000,000 plus 60% of the net proceeds of any sale or issuance of Equity Interests in the Borrower after the Closing Date. Notwithstanding anything contained in this Section 7.7 or elsewhere in this Agreement to the contrary, the aggregate amount of consideration that is permitted by subsection (j) above and the Capital Expenditures permitted to be made under Section 7.1(d)(x) related to the net proceeds of any sale or issuance of Equity Interests after the Closing Date shall not exceed, in the aggregate, 60% of all such proceeds. All loans and advances from the Borrower to any Subsidiary, among Subsidiaries or from any Subsidiary to the Borrower, shall be represented by the Global Intercompany Note, and such note shall be pledged by each Loan Party to the Agent in accordance with the Security Agreement.

7.8 Asset Sales. Convey, sell, lease, transfer or otherwise dispose of, or permit any Subsidiary to convey, sell, lease, transfer or otherwise dispose of, in one transaction or a series of transactions, all or any part of its or its Subsidiary's business, properties or assets outside the ordinary course of business (including Equity Interests of a Subsidiary, but excluding assets subject to Liens permitted by <u>Section 7.2(g)</u>), whether now owned or hereafter acquired. Notwithstanding anything herein to the contrary, the Borrower shall not convey, sell, lease, transfer or otherwise dispose of any of the following assets to a Foreign Branch or Foreign Subsidiary: (i) any equipment for which an invoice was submitted pursuant to <u>Section 5.2(e)</u>, (ii) any assets included in the calculation of the Borrowing Base as set forth in the most recently delivered Borrowing Base Certificate, or (iii) any Intellectual Property constituting Collateral.

7.9 Transactions with Affiliates. Except as set forth on <u>Schedule 7.9</u> or transactions engaged in in the ordinary course of business and consistent with the Borrower's then current transfer pricing policy, enter into or permit to exist, or permit any of its Subsidiaries to enter into or permit to exist, any transaction (including the purchase, sale, lease or exchange of any property or the rendering of any service) with any Affiliate of the Borrower on terms that are less favorable to Borrower or that Subsidiary, as the case may be, than those that might be obtained at the time from Persons who are not such an Affiliate.

7.10 Conduct of Business. Engage in any business, or permit any of its Domestic Subsidiaries to engage in any business, other than the businesses engaged in by the Borrower and its Domestic Subsidiaries on the date hereof and similar or directly related businesses.

7.11 Fiscal Year. Change the Borrower's fiscal year from a year ending December 31.

7.12 Security Matters.

(a) <u>Name and Organization</u>. Change the Borrower's, a Domestic Subsidiary's or a first-tier Foreign Subsidiary's name, identity or corporate structure or organizational number or reorganize, reincorporate or take any other action that results in a change of the jurisdiction of organization of the Borrower or such a Subsidiary, without giving the Agent thirty (30) days' prior written notice thereof, <u>provided</u>, the Borrower's jurisdiction shall at all times remain within the United States.

(b) <u>Perfection</u>. Permit any other Person to maintain possession or control of any Equipment or Inventory (in each case as defined in the Security Agreement) of the Borrower or a Domestic Subsidiary (other than Inventory or Equipment of a Foreign Branch located in Taiwan) unless, subject to <u>Section 6.11</u>, the Agent has received a waiver from such Person satisfactory to the Agent (or for Inventory to be sold by Amazon or Cisco, at a location for which an adequate rent reserve in an amount determined by the Agent, in its sole discretion, has been established); permit any certificated security or instrument to be included in the Collateral, unless they have been delivered to the Agent (with appropriate endorsements); establish, or permit any Domestic Subsidiary to establish, any deposit or securities account, unless the Agent has received a control agreement satisfactory to the Agent (or the Agent otherwise has a perfected first priority security interest in such deposit or securities account), other than deposit accounts and securities accounts located in Taiwan and set forth on <u>Schedule 4.20</u>.

7.13 Limitation on Other Restrictions on Liens. Enter into, or become subject to, or permit any Domestic Subsidiary to enter into, or become subject to, any agreement or instrument that would prohibit the grant of any Lien on any of its properties, except the Loan Documents and restrictions on granting Liens in property subject to Liens permitted by <u>Sections 7.2(f)</u>.

7.14 Limitation on Other Restrictions on Amendment of the Loan Documents. Enter into, or become subject to, or permit any Subsidiary to enter into, or become subject to, any agreement or instrument that would prohibit or require the consent of any Person to any amendment, modification or supplement to any of the Loan Documents.

7.15 Limitations on Modifications of Certain Agreements and Instruments. Amend, modify or supplement, or permit any Domestic Subsidiary or first-tier Foreign Subsidiary to, amend, modify or supplement, its charter or governing documents, or amend, modify or supplement any, terminate, or waive or release any of its rights or remedies under, the Existing EWB Loan Documents

7.16 Agreements Related to Negative Covenants. Enter into or suffer to exist, or permit any Domestic Subsidiary to enter into or suffer to exist, any agreement to do any act prohibited by this <u>Article VII</u>, unless the Obligations hereunder will be repaid in full as a result thereof.

7.17 Compliance with Anti-Terrorism Regulations.

(a) (i) Violate any Anti-Terrorism Laws (ii) engage in any transaction, investment, undertaking or activity that conceals the identity, source or destination of the proceeds from any category of prohibited offenses designated by the Organization for Economic Co-operation and Development's Financial Action Task Force on Money Laundering or (iii) permit any of their respective Affiliates to violate these laws or engage in these actions.

(b) (i) Use, directly or indirectly, the proceeds of the Loans, or lend, contribute or otherwise make available such proceeds to any subsidiary, joint venture partner or other Person, (x) to fund any activities or business of or with any Person, or in any country or territory, that, is, or whose government is, the subject of Sanctions at the time of such funding, or (y) in any other manner that would result in a violation of Sanctions by any Person (including any Person participating in the Loans, whether as underwriter, advisor, investor, or otherwise).

(c) (i) Deal in, or otherwise engage in any transaction related to, any property or interests in property blocked pursuant to any Anti-Terrorism Law, (ii) engage in or conspire to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempt to violate, any of the prohibitions set forth in any Anti-Terrorism Law or (iii) permit any of their respective Affiliates to do any of the foregoing.

ARTICLE VIII EVENTS OF DEFAULT

8.1 Events of Default. If any of the following events ("Events of Default") shall occur and be continuing:

(a) The Borrower shall fail to pay, when due, any amount of principal of any Loan, any interest on any Loan, any fee hereunder, or any other amount payable hereunder or under any other Loan Document; or

(b) The Borrower shall fail to perform or observe any term, covenant or agreement contained in Article <u>VII</u>, other than <u>7.10</u> or <u>7.11</u> or other than as set forth in <u>Section 8.1(c) and (d)</u> below; or

(c) The Borrower shall fail to perform or observe any term, covenant or agreement contained in <u>Section 7.4</u>, <u>7.7</u> (so long as all Investments giving rise to such failure is less than \$750,000 in the aggregate), <u>7.8</u> (so long as all conveyances, sales, leases, transfers and other dispositions giving rise to such failure are less than \$750,000 in the aggregate), <u>7.9</u>, <u>7.13</u>, <u>7.14</u> or <u>7.15</u> and any such failure continues for two (2) Business Days after the Borrower obtains knowledge or the Agent gives notice thereof; or

(d) The Borrower shall fail to perform or observe any term, covenant or agreement contained in this Agreement or any other Loan Document, other than those referred to in <u>Section 8.1(a)</u>, (b) or (c) and any such failure shall remain unremedied for thirty (30) days after the Borrower obtains knowledge or the Agent gives notice thereof; <u>provided</u>, however, that, if, in the Agent's reasonable discretion, such failure cannot be remedied or cured within thirty (30) days, so long as Borrower immediately commences such steps which the Agent deems, in its reasonable discretion, sufficient to remedy or cure such failure and thereafter continuously and diligently pursues the same to the Agent's satisfaction, in its reasonable discretion, then Borrower shall have such additional time as to cure or remedy such failure as the Agent, in its sole discretion, may from time to time agree in writing; or

(e) Any representation or warranty made by any Loan Party, in any Loan Document shall prove to have been incorrect in any material respect when made or deemed made; or

(f) The Borrower and its Domestic Subsidiaries shall (i) fail to pay any principal of, or premium or interest on, any Debt owed to a Lender Party, when due (whether by scheduled maturity, required prepayment, acceleration, demand or otherwise), or (ii) fail to perform or observe any term, covenant or condition on its part to be performed or observed under any agreement or instrument relating to any Debt owed to a Lender Party, when required to be performed or observed, and the effect of such default or other event is to cause, or to permit the holder of such Debt to cause, such Debt to be demanded or to become due or to be repurchased, prepaid, defeased or redeemed or any offer therefor to be made, prior to its stated maturity; or

(g) (i) The Borrower or any of its Subsidiaries shall commence any case, proceeding or other action (A) under any existing or future Law, relating to bankruptcy, insolvency, reorganization or relief of debtors, seeking to have an order for relief entered with respect to it, or seeking to adjudicate it a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to it or its debts, or (B) seeking appointment of a receiver, trustee, custodian or other similar official for it or for all or any substantial part of its assets, or the Borrower or any of its Subsidiaries shall make a general assignment for the benefit of its creditors; or (ii) there shall be commenced against the Borrower or any of its Subsidiaries any case, proceeding or other action of a nature referred to in clause (i) above which (A) results in the entry of an order for relief or any such adjudication or appointment or (B) remains undismissed or undischarged for a period of sixty (60) days; or

(h) One or more judgments, attachments or decrees shall be entered against the Borrower or any of its Subsidiaries, individually or in the aggregate in excess of 25% of the amount of unrestricted cash and cash equivalents of the Borrower and its Subsidiaries on a consolidated basis immediately before and after such judgment, attachment or decree is entered and provided that such balance of unrestricted cash and cash equivalents is maintained until any such judgment, attachment or decree is actually paid by the Borrower or any of its Domestic Subsidiaries, and all such judgments, attachments or decrees shall not have been vacated, discharged, or stayed or bonded pending appeal within sixty (60) days from the entry thereof; or

(i) There shall occur one or more ERISA Events that might reasonably be expected to result in liability of the Borrower, any of its Subsidiaries or any of their respective ERISA Affiliates during the term of this Agreement; or there shall exist an amount of unfunded benefit liabilities (as defined in Section 4001(a)(18) of ERISA), individually or in the aggregate for all Pension Plans (excluding for purposes of such computation any Pension Plans with respect to which assets exceed benefit liabilities), which exceeds the amount set forth in the financial statements delivered pursuant to <u>Section 4.6</u>; or

(j) Any guaranty relating to the Loans, for any reason other than satisfaction in full of all Obligations, ceases to be in full force and effect or is declared null and void, or any guarantor denies that it has any further liability under such guaranty or gives notice to such effect; or

(k) Any Security Document shall cease to be in full force and effect or any Lien created by or purported to be created by the Security Documents ceases to be valid, enforceable and perfected first priority liens except to the extent expressly permitted by the Loan Documents; or

(l) A Change in Control shall occur; or

(m) Any "Event of Default" shall occur under the Existing EWB Loan Documents;

THEN, (i) upon the occurrence of any Event of Default described in clause (f) above, the Commitments shall immediately terminate and all Loans hereunder with accrued interest thereon, and all other Obligations under this Agreement, the Notes and the other Loan Documents shall automatically become due and payable; (ii) upon the occurrence of any other Event of Default, the Agent may (with the consent of the Required Lenders, and shall upon the direction of the Required Lenders), by notice to the Borrower, declare the Commitments to be terminated forthwith, whereupon the Commitments shall immediately terminate, and/or, by notice to the Borrower, declare the Loans hereunder, together with accrued interest thereon, and all other Obligations under this Agreement, the Notes and the other Loan Documents to be due and payable forthwith, whereupon the same shall immediately become due and payable and (iii) upon the occurrence of any Event of Default, the Agent may (with the consent of the Required Lenders, and shall upon the direction of the Required Lenders) exercise the remedies available to it under the other Loan Documents, and at law or in equity.

8.2 Application of Funds. After the exercise of remedies under <u>Section 8.1</u>, any amounts received on account of Obligations shall be applied by Agent in the following order:

<u>First</u>, to payment of that portion of the Obligations constituting fees, indemnities, expenses and other amounts payable to the Agent in its capacity as such;

<u>Second</u>, to payment of that portion of the Obligations constituting fees, indemnities and other amounts (other than principal and interest) payable to the Lenders, ratably among them in proportion to the amounts described in this clause <u>Second</u> payable to them;

Third, to payment of that portion of the Obligations constituting accrued and unpaid interest on the Loans and other Obligations, ratably among the Lenders in proportion to the respective amounts described in this clause <u>Third</u> payable to them;

<u>Fourth</u>, to payment of that portion of the Obligations constituting unpaid principal of the Loans, ratably among the Lenders in proportion to the respective amounts described in this clause <u>Fourth</u> held by them; and

Last, the balance, if any, after all of the Obligations have been indefeasibly paid in full, to the Borrowers or as otherwise required by Law.

ARTICLE IX AGENCY

9.1 Appointment and Authority. Each of the Lenders hereby irrevocably appoints East West Bank to act on its behalf as the Agent hereunder and under the other Loan Documents and authorizes the Agent to take such actions on its behalf and to exercise such powers as are delegated to the Agent to take such actions on its behalf and to exercise such powers as are delegated to the Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article are solely for the benefit of the Lender Parties and neither the Borrower nor any other Loan Party shall have rights as a third party beneficiary of any such provisions.

9.2 Rights as a Lender. The Person serving as the Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Agent hereunder and without any duty to account therefor to the Lenders.

9.3 Exculpatory Provisions. The Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Agent:

(a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or Event of Default has occurred and is continuing;

(b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents), provided, that the Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Agent to liability or that is contrary to any Loan Document or applicable Law; and

(c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Agent or any of its Affiliates in any capacity.

The Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Agent shall believe in good faith shall be necessary), under the circumstances as provided in <u>Sections 10.1</u> and <u>8.1</u> or (ii) in the absence of its own gross negligence or willful misconduct. The Agent shall be deemed not to have knowledge of any Default or Event of Default unless and until notice describing such Default or Event of Default is given to the Agent by the Borrower or a Lender.

The Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or the occurrence of any Default or Event of Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or document or (v) the satisfaction of any condition set forth in <u>Article V</u> or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Agent.

9.4 Reliance by Agent. The Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing delivered or available to it in any format so long as it is believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan that by its terms must be fulfilled to the satisfaction of a Lender, the Agent may presume that such condition is satisfactory to such Lender unless the Agent shall have received notice to the contrary from such Lender prior to the making of such Loan. The Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

9.5 Delegation of Duties. The Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub agents appointed by the Agent. The Agent and any such sub agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article shall apply to any such sub agent and to the Related Parties of the Agent and any such sub agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Agent.

9.6 Resignation of Agent. The Agent may at any time give notice of its resignation to the Lenders and the Borrower. Upon receipt of any such notice of resignation, the Required Lenders shall have the right, in consultation with the Borrower, to appoint a successor, which shall be a bank with an office in the United States, or an Affiliate of any such bank. If no such successor shall have been so appointed by the Required Lenders and shall have accepted such appointment within 30 days after the retiring Agent gives notice of its resignation, then the retiring Agent may on behalf of the Lenders, appoint a successor Agent meeting the qualifications set forth above, provided, that if the Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (1) the retiring Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents (except that in the case of any collateral security held by the Agent on behalf of the Lenders under any of the Loan Documents, the retiring Agent shall continue to hold such collateral security until such time as a successor Agent is appointed) and (2) all payments, communications and determinations provided to be made by, to or through the Agent shall instead be made by or to each Lender and directly, until such time as the Required Lenders appoint a successor Agent as provided for above in this paragraph. Upon the acceptance of a successor's appointment as Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Agent, and the retiring Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this paragraph). The fees payable by the Borrower to a successor Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article and Section 10.4 and 10.5 shall continue in effect for the benefit of such retiring Agent, its sub agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Agent was acting as Agent.

9.7 Non-Reliance on Agent and Other Lenders. Each Lender acknowledges that it has, independently and without reliance upon the Agent or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender also acknowledges that it will, independently and without reliance upon the Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

ARTICLE X MISCELLANEOUS

10.1 Amendments, Etc. No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent to any departure by the Borrower or any other Loan Party therefrom, shall be effective unless in writing signed by the Required Lenders and the Borrower or the applicable Loan Party, as the case may be, and acknowledged by the Agent, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, that no such amendment, waiver or consent shall:

(a) extend or increase the Commitment of any Lender (or reinstate any Commitment terminated pursuant to <u>Section 8.2</u>) without the written consent of such Lender;

(b) postpone any date fixed by this Agreement or any other Loan Document for any payment or mandatory prepayment of principal, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under any other Loan Document without the written consent of each Lender directly affected thereby;

(c) reduce the principal of, or the rate of interest specified herein on, any Loan, or any fees or other amounts payable hereunder or under any other Loan Document without the written consent of each Lender directly affected thereby, it being understood that any waiver of or reduction in the post-default rate of interest shall not be deemed a reduction of the rate of interest;

(d) change <u>Section 3.6</u> or <u>Section 10.13</u> in a manner that would alter the pro rata sharing of payments required thereby without the written consent of each Lender;

(e) change any provision of this Section or the definition of "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender; or

(f) release any Guarantor from its guaranty or release a substantial portion of the Collateral, without the written consent of each Lender;

provided further, that no amendment, waiver or consent shall, unless in writing and signed by the Agent in addition to the Lenders required above, affect the rights or duties of the Agent under this Agreement or any other Loan Document.

10.2 No Implied Waiver; Remedies Cumulative. No delay or failure of any Lender or the Agent in exercising any right or remedy under this Agreement shall operate as a waiver thereof; nor shall any single or partial exercise of any such right or remedy preclude any other or further exercise thereof or the exercise of any other right or remedy. The rights and remedies of the Lender under this Agreement are cumulative and not exclusive of any other rights or remedies available hereunder, under any other agreement, at law, or otherwise.

10.3 Notices. All notices and other communications (collectively, "notices") under this Agreement shall be in writing (including facsimile transmission) and shall be sent by first-class mail, by nationally-recognized overnight courier, by personal delivery, by facsimile transmission, or by e-mail, in all cases with charges prepaid. All notices shall be sent to a party at its address specified on the signature page hereof, or to such other address as shall have been designated by the applicable party by notice to the other party hereto. Any properly given notice shall be deemed given or made upon the earliest of: (i) if delivered by hand or by courier, when signed for by or on behalf of the relevant party ; (ii) if delivered by mail, four Business Days after deposit in the mails, or (iii) if delivered by facsimile or e-mail, when sent and receipt has been confirmed by telephone; <u>provided</u>, (x) that notices to the Lender pursuant to Article II shall not be effective until actually received by the Agent and (y) any notice from the Agent or any Lender alleging a default or breach of any Loan Document or the occurrence of any Default or Event of Default shall not be delivered solely by facsimile or e-mail. The Lenders and the Agent may rely on any notice, including any notice of Borrowing (whether or not made in a manner contemplated by this Agreement), purportedly made by or on behalf of the Borrower, and the Lenders shall have no duty to verify the identity or authority of the Person giving such notice.

10.4 Expenses. The Borrower shall pay (i) all reasonable out of pocket expenses incurred by the Agent and its Affiliates (including the reasonable fees, charges and disbursements of counsel for the Agent), in connection with the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated), (ii) all reasonable out-of-pocket expenses incurred by the Lenders (including the reasonable fees, charges and disbursements of counsel for Lenders), in connection with the preparation, negotiation, execution and delivery of this Agreement and the other Loan Documents or any amendments, modifications or waivers of the provisions hereof (whether or not the transactions contemplated hereby or thereby shall be consummated), and (iii) all out-of-pocket expenses incurred by the Agent or any Lender (including the fees, charges and disbursements of counsel for the Agent and a reasonable estimate of the allocated cost of in-house counsel for the Agent and in-house counsel for the Agent and each Lender and a reasonable estimate of the allocated cost of in-house counsel for the Agent and in-house counsel for each Lender (to the extent not duplicative of outside counsel)), in connection with the enforcement or protection of its rights (A) in connection with this Agreement and the other Loan Documents, including its rights under this Section, or (B) in connection with the Loans made hereunder, including all such out-of-pocket expenses incurred of such Loans.

10.5 Indemnity.

(a) <u>Indemnity by Borrower</u>. The Borrower agrees to defend, indemnify, pay and hold the Lender Parties and their Related Parties, harmless from and against any and all claims, liabilities, losses, damages, costs and expenses (whether or not any of the foregoing Persons is a party to any litigation), and costs of investigation, document production, attendance at a deposition, or other discovery, with respect to or arising out of this Agreement or the Loan Documents or any use of proceeds hereunder or any exercise by the Agent or the Lenders of their rights and remedies under this Agreement and the other Loan Documents or any claim, demand, action or cause of action being asserted against the Borrower or any of its Subsidiaries, including without limitation any violation of any Environmental Law or other Law or any environmental claim based upon the management, use, control, ownership or operation of property of the Borrower (or any other Loan Party) (collectively, the "<u>Indemnified Liabilities</u>"), provided that the Borrower shall have no obligation hereunder to an indemnified party, with respect to Indemnified Liabilities arising from the gross negligence or willful misconduct of such indemnified party. This covenant shall survive termination of this Agreement and payment of the outstanding Notes.

(b) <u>Reimbursements by Lender</u>. To the extent that the Borrower for any reason fails to indefeasibly pay any amount required under <u>Section 10.5(a)</u> or <u>Section 10.4</u> to be paid by it to the Agent or any Related Party of any of the foregoing, each Lender severally agrees to pay to the Agent or such Related Party, as the case may be, such Lender's Applicable Percentage (determined as of the time that the applicable unreimbursed expense indemnity payment is sought) of such unpaid amount; <u>provided</u>, that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Agent in its capacity as such or against any Related Party in connection with such capacity. The obligations of the Lenders under this subsection (b) are subject to the provisions of <u>Section 3.13(d)</u>.

(c) <u>Payments</u>. All amounts due under this Section shall be payable promptly after demand therefor.

10.6 Assignments and Participations.

(a) <u>Successors and Assigns Generally</u>. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that neither the Borrower nor any other Loan Party may assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Agent and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of subsection (b) of this Section, (ii) by way of participation in accordance with the provisions of subsection (d) of this Section, or (iii) by way of pledge or assignment of a security interest subject to the restrictions of subsection (f) of this Section (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section and, to the extent expressly contemplated hereby, the Related Parties of each of the Lender Parties) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans at the time owing to it); <u>provided</u>, that

(i) except in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the Loans at the time owing to it or in the case of an assignment to a Lender, the aggregate amount of the Commitment (which for this purpose includes Loans outstanding thereunder) or, if the Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date), shall not be less than \$2,500 unless the Agent, and so long as no Event of Default has occurred and is continuing, the Borrower, otherwise consents (each such consent not to be unreasonably withheld or delayed);

(ii) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loans or the Commitment assigned;

(iii) the consent of the Agent is required for any assignment of a Commitment or the Loans, unless the Person that is the proposed assignee would otherwise qualify as an Eligible Assignee;

(iv) the parties to each assignment shall execute and deliver to the Agent an Assignment and Assumption, together with a processing and recordation fee in the amount of \$3,500, and the Eligible Assignee, if it shall not be a Lender, shall deliver to the Agent an Administrative Questionnaire; and

(v) no such assignment shall be made to the Borrower, any of the Borrower's Affiliates, Subsidiaries or Related Persons or to a natural person.

Subject to acceptance and recording thereof by the Agent pursuant to subsection (c) of this Section, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of <u>Sections 3.9</u>, <u>10.4</u>, and <u>10.5</u> with respect to facts and circumstances occurring prior to the effective date of such assignment. Upon request, the Borrower (at its expense) shall execute and deliver Notes to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with subsection (d) of this Section.

(c) <u>Register</u>. The Agent, acting solely for this purpose as an agent of the Borrower (and such agency being solely for tax purposes), shall maintain at the Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amounts of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "<u>Register</u>"). The entries in the Register shall be conclusive, and the Borrower, the Lender Parties may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by each of the Borrower and any Lender at any reasonable time and from time to time upon reasonable prior notice.

(d) <u>Participations</u>. Any Lender may at any time, without the consent of, or notice to, the Borrower or the Agent, sell participations to any Person (other than a natural person, or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each, a "<u>Participant</u>") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans); <u>provided</u>, that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Agent, and the Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; <u>provided</u>, that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the first proviso to <u>Section 10.1</u> that affects such Participant. Subject to subsection (e) of this Section, the Borrower agrees that each Participant shall be entitled to the benefits of <u>Section 3.9</u> to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (b) of this Section. To the extent permitted by Law, each Participant also shall be entitled to the benefits of <u>Section 10.13</u> as though it were a Lender.

(e) <u>Limitations upon Participant Rights</u>. A Participant shall not be entitled to receive any greater payment under <u>Section 3.9</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent.

(f) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Notes, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; provided that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.

10.7 Entire Agreement. This Agreement, together with the Exhibits and the Schedules hereto, the other Loan Documents, and any separate letter agreement with respect to fees payable to the Agent constitutes the entire agreement of the parties hereto with respect to the subject matters hereof and supersedes all prior and contemporaneous understandings and agreements.

10.8 Survival. All representations and warranties of the Borrower contained in or made in connection with this Agreement or in any other Loan Documents shall survive, and shall not be waived by, the execution and delivery of this Agreement, any investigation by or knowledge of the Agent, any extension of credit, or any other event or circumstance whatever, but, except as expressly set forth herein, shall terminate upon the indefeasible payment in full in cash of all Obligations.

10.9 Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, and all such counterparts shall constitute but one and the same agreement.

10.10 Severability. In case any one or more of the provisions contained in this Agreement should be invalid, illegal or unenforceable in any respect, the validity , legality and enforceability of the remaining provisions contained herein shall not be in any way be affected or impaired thereby.

effect.

10.11 Headings. Section headings in this Agreement are included for convenience of reference only and shall not be given any substantive

10.12 Setoff. In the event that any obligation of the Borrower now or hereafter existing under this Agreement or any other Loan Document shall have become due and payable, each Lender is hereby authorized by the Borrower, at any time and from time to time, without notice, (a) to set off against, and to appropriate and apply to the payment of, the Obligations and liabilities of the Borrower under the Loan Documents (whether matured or unmatured, fixed or contingent or liquidated or unliquidated) any and all amounts owing by such Lender to the Borrower (whether payable in Dollars or any other currency, whether matured or unmatured, and, in the case of deposits, whether general or special, time or demand and however evidenced) and (b) pending any such action, to the extent necessary, to hold such amounts as collateral to secure such Obligations and liabilities and to return as unpaid for insufficient funds any and all checks and other items drawn against any deposits so held as such Lender in its sole discretion may elect. The Borrower hereby grants to each Lender under this <u>Section 10.12</u> are in addition to other rights and remedies (including other rights of set-off) which such Lender may have. The Borrower agrees that, to the fullest extent permitted by Law, any Affiliate of each Lender, and any holder of a participation in any obligation of the Borrower under this <u>Section 10.12</u> regardless of whether such Affiliate or participant otherwise would be deemed a creditor of the Borrower. Each Lender agrees to notify the Borrower and the Agent promptly after any such setoff and application; <u>provided</u>, that the failure to give such notice shall not affect the validity of such setoff and application.

10.13 Sharing of Payments By Lenders. If any Lender shall, by exercising any right of setoff or counterclaim or otherwise, obtain payment in respect of any principal of or interest on any of its Loans or other Obligations hereunder resulting in such Lender's receiving payment of a proportion of the aggregate amount of its Loans and accrued interest thereon or other such obligations greater than its pro rata share thereof as provided herein, then the Lender receiving such greater proportion shall (a) notify the Agent and each other Lender of such fact, and (b) purchase (for cash at face value) participations in the Loans and such other obligations of the other Lenders, or make such other adjustments as shall be equitable, so that the benefit of all such payments shall be shared by the Lenders ratably in accordance with the aggregate amount of principal of and accrued interest on their respective Loans and other amounts owing them, provided that:

(i) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations shall be rescinded and the purchase price restored to the extent of such recovery, without interest; and

(ii) the provisions of this paragraph shall not be construed to apply to (x) any payment made by the Borrower pursuant to and in accordance with the express terms of this Agreement or (y) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans to any assignee or participant, other than to the Borrower or any Domestic Subsidiary thereof (as to which the provisions of this paragraph shall apply).

Each Loan Party consents to the foregoing and agrees, to the extent it may effectively do so under applicable Law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against such Loan Party rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of each Loan Party in the amount of such participation.

10.14 Limitation on Payments. The parties hereto intend to conform to all applicable Laws limiting the maximum rate of interest that may be charged or collected by the Agent or any Lender from the Borrower. Accordingly, notwithstanding any other provision hereof, the Borrower shall not be required to make any payment to or for the account of the Agent or any Lender, and such Person shall refund any payment made by the Borrower, to the extent that such requirement or such failure to refund would violate or conflict with mandatory and nonwaivable provisions of applicable Law limiting the maximum amount of interest which may be charged or collected by the Agent or any Lender from the Borrower. To the fullest extent permitted by law, in any action, suit or proceeding pertaining to this Agreement, the burden of proof, by clear and convincing evidence, shall be on the Borrower to demonstrate that this <u>Section 10.14</u> applies to limit any obligation of the Borrower under this Agreement or to require the Agent or any Lender to make any refund, or claiming that this Agreement conflicts with any applicable Law limiting the maximum rate of interest that may be charged or collected by the Agent or any Lender form the Borrower, as to each element of such claim.

10.15 Disclosure of Information to Affiliates, Confidentiality.

(a) <u>Confidentiality</u>. Each of the Lender Parties agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its and its Affiliates' respective partners, directors, officers, employees, agents, advisors and other representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any regulatory authority purporting to have jurisdiction over it (including any self-regulatory authority), (c) to the extent required by applicable Laws or regulations or by any subpoena or similar legal process, (d) to any other party hereto, (e) in connection with the exercise of any remedies hereunder or under any other Loan Document or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights or obligations under this Agreement or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower and its obligations, (g) with the consent of the Borrower or (h) to the extent such Information (x) becomes publicly available other than as a result of a breach of this Section or (y) becomes available to the Agent or any Lender or any of their respective Affiliates on a nonconfidential basis from a source other than the Borrower and its Affiliates.

(b) <u>Information</u>. For purposes of this Section, "Information" means all information received from the Borrower, any of its Subsidiaries or any of their respective Affiliates relating to the Borrower or any of its Subsidiaries or any of their respective businesses, other than any such information that is available to the Agent or any Lender on a nonconfidential basis prior to disclosure by the Borrower, any of its Subsidiaries or any of their respective Affiliates, <u>provided</u>, that, in the case of information received from the Borrower or any of its Subsidiaries after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

The Agent and each Lender acknowledges that (a) the Information may include material non-public information concerning the Borrower or a Subsidiary, as the case may be, (b) it has developed compliance procedures regarding the use of material non-public information and (c) it will handle such material non-public information in accordance with applicable Law, including United States Federal and state securities Laws.

(c) <u>Other Permitted Disclosures</u>. Anything in this Agreement to the contrary notwithstanding, Agent may use the name, logos, and other insignia of any Borrower and the Loan Parties and the Commitments provided hereunder in any "tombstone" or comparable advertising, on its website or in other marketing materials of Agent.

10.16 Governing Law. THIS AGREEMENT AND THE NOTES SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT GIVING EFFECT TO ITS CHOICE OF LAW PRINCIPLES.

10.17 Waiver of Jury Trial. EACH OF THE PARTIES TO THIS AGREEMENT HEREBY AGREES TO WAIVE ITS RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS AGREEMENT OR ANY OF THE OTHER LOAN DOCUMENTS OR ANY DEALINGS BETWEEN THEM RELATING TO THE SUBJECT MATTER OF THIS LOAN TRANSACTION OR THE LENDER/BORROWER RELATIONSHIP THAT IS BEING ESTABLISHED.

10.18 Consent to Jurisdiction; Venue. All judicial proceedings brought against the Borrower with respect to this Agreement and the Loan Documents may be brought in any state or federal court of competent jurisdiction in sitting in Los Angeles County, California, and by execution and delivery of this Agreement, the Borrower accepts for itself and in connection with its properties, generally and unconditionally, the nonexclusive jurisdiction of the aforesaid courts, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Agreement. The Borrower irrevocably waives any right it may have to assert the doctrine of <u>forum non conveniens</u> or to object to venue to the extent any proceeding is brought in accordance with this Section.

10.19 USA Patriot Act Notice. The Agent and each Lender hereby notifies the Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56), as amended, it is required to obtain, verify and record information that identifies the Borrower, which information includes the name and address of the Borrower and other information that will allow the Agent and each Lender to identify the Borrower in accordance with such Act.

10.20 Keepwell. Provided that the Borrower continues to be a Qualified ECP Guarantor, the Borrower hereby unconditionally and irrevocably undertakes to provide such funds or other support as may be needed from time to time by each other Loan Party to honor all of its obligations under the applicable Guaranty or other Loan Document in respect of Swap Obligations; <u>provided</u>, that the Borrower shall only be liable under this <u>Section</u> <u>10.20</u> for the maximum amount of such liability that can be hereby incurred without rendering its obligations under this <u>Section</u> <u>10.20</u>, or otherwise under this Agreement, voidable under applicable law relating to fraudulent conveyance or fraudulent transfer, and not for any greater amounts. The obligations of the Borrower under this <u>Section</u> <u>10.20</u> shall remain in full force and effect until the Obligations and all other amounts payable under this Agreement have been paid in cash and performed in full, and all commitments to extend credit under the Loans Documents have terminated. The Borrower intends that this <u>Section</u> <u>10.20</u> constitute, and this <u>Section</u> <u>10.20</u> shall be deemed to constitute, a "keepwell, support, or other agreement" for the benefit of each such other Loan Party for all purposes of Section <u>1a(18)(A)(v)(II)</u> of the Commodity Exchange Act.

10.21 Limitation of Liability. TO THE FULLEST EXTENT PERMITTED BY LAW, NO CLAIM MAY BE MADE BY THE BORROWER AGAINST THE AGENT, ANY LENDER OR ANY OF THEIR RELATED PARTIES FOR ANY SPECIAL, INDIRECT, CONSEQUENTIAL OR PUNITIVE DAMAGES IN RESPECT OF ANY CLAIM ARISING FROM OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR ANY STATEMENT, COURSE OF CONDUCT, ACT, OMISSION OR EVENT IN CONNECTION WITH ANY OF THE FOREGOING (WHETHER BASED ON BREACH OF CONTRACT, TORT OR ANY OTHER THEORY OF LIABILITY); AND THE BORROWER HEREBY WAIVES, RELEASES AND AGREES NOT TO SUE UPON ANY CLAIM FOR ANY SUCH DAMAGES, WHETHER OR NOT ACCRUED AND WHETHER OR NOT KNOWN OR SUSPECTED TO EXIST; PROVIDED THAT THE ABOVE LIMITATION ON LIABILITY SHALL NOT APPLY TO CLAIMS FOR SPECIAL, INDIRECT, CONSEQUENTIAL OR PUNITIVE DAMAGES IN RESPECT TO ANY SUCH CLAIM THAT IS MADE AS A DIRECT RESULT OF THE AGENT'S OR A LENDER'S GROSS NEGLIGENCE OR WILLFUL MISCONDUCT.

THE BORROWER ACKNOWLEDGES THAT IT HAS BEEN REPRESENTED BY LEGAL COUNSEL IN CONNECTION WITH THE EXECUTION AND DELIVERY OF THIS AGREEMENT AND THAT IT UNDERSTANDS THE PROVISIONS OF THIS AGREEMENT.

APPLIED OPTOELECTRONICS, INC.

By<u>: /s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

Address: 13115 Jess Pirtle Blvd. Sugar Land, TX 77478 Attention: Jessica Hung Fax: 281-295-1888 E-mail: jhung@ao-inc.com

EAST WEST BANK, as Agent and a Lender

By: <u>/s/ Kelvin Chan</u> Name: Kelvin Chan Title: Managing Director

Address: East West Bank Attn: Kelvin Chan, Technology Banking Group - West Region 2350 Mission College Blvd., Suite 988 Santa Clara, CA 95054 Telephone: (408) 330-2088 Fax: (408) 588-9688 E-Mail: kelvin.chan@eastwestbank.com

With a copy to: East West Bank Attn: Mercie Martinez, Loan Servicing Department 9300 Flair Drive, 6th Floor El Monte, CA 91731 Telephone: (626) 371-8746 Fax: (626) 927-2088 E-Mail: Mercedes.Martinez@eastwestbank.com

Signature Page to Credit Agreement

COMERICA BANK, as a Lender

By: <u>/s/ Alice Yang</u> Name: Alice Yang Title: Senior Vice President

Address: Comerica Bank 2900 North Loop West, 9th Floor Houston, TX 77092 Attn: Alice Yang Telephone: (713) 507-2502 Fax: _____ E-Mail aliceyang@comerica.com

Signature Page to Credit Agreement

ANNEX 1

Commitments

Lender	Maximum Facility A Revolving Commitment	Facility B Revolving Commitment	Term Commitment
East West Bank	\$12,000,000	\$3,000,000	\$6,000,000
Comerica Bank	\$8,000,000	\$2,000,000	\$4,000,000

Annex 1

ANNEX 2

Agent's Office

East West Bank Attn: Mercie Martinez, Loan Servicing Department 9300 Flair Drive, 6th Floor El Monte, CA 91731 Telephone: (626) 371-8746 Fax: (626) 927-2088 E-Mail: Mercedes.Martinez@eastwestbank.com

<u>Secondary Contact</u>: Cristal Li Telephone: (626) 371-8726 Fax: (626) 927-2088 Email: Part&synd@eastwestbank.com

Annex 2

EXHIBIT A

[FORM OF COMPLIANCE CERTIFICATE] COMPLIANCE CERTIFICATE

THE UNDERSIGNED HEREBY CERTIFY THAT:

(1) I am the duly elected [Title] of Applied Optoelectronics, Inc., a Delaware corporation (the "Borrower");

(2) I have reviewed the terms of the Credit Agreement, dated as of June 30, 2015(as amended, modified or supplemented, the "<u>Credit Agreement</u>", the terms defined therein and not otherwise defined in this Certificate (including Attachment No. 1 annexed hereto and made a part hereof), being used in this Certificate as therein defined), by and among the Borrower, the Lenders party thereto and East West Bank, as Agent, and the terms of the other Loan Documents, and I have made, or have caused to be made under my supervision, a review in reasonable detail of the transactions and condition of Borrower and its Subsidiaries during the accounting period covered by the attached financial statements; and

(3) The examination described in paragraph (2) above did not disclose, and I have no knowledge of, the existence of any condition or event which constitutes an Event of Default during or at the end of the accounting period covered by the attached financial statements or as of the date of this Certificate[, except as set forth below].

(4) Attached as Attachment 1 is a calculation of the financial covenants set forth in <u>Section 7.1</u> of the Credit Agreement demonstrating compliance with such covenants for the most recently completed fiscal quarter [year] or at such quarter [year] end.

[Set forth [below] [in a separate attachment to this Certificate] are all exceptions to paragraph (3) above listing, in detail, the nature of the condition or event, the period during which it has existed and the action which the Borrower has taken, is taking, or proposes to take with respect to each such condition or event:

A-1

The foregoing certifications, together with the computations set forth in Attachment No. 1 annexed hereto and made a part hereof and the financial statements delivered with this Certificate in support hereof, are made and delivered this [______ day of ______, ___] pursuant to Section 6.1(c) of the Credit Agreement.

APPLIED OPTOELECTRONICS, INC.

By:	
Name:	
Title:	

[Add Attachment 1]

EXHIBIT B

ASSIGNMENT AND ASSUMPTION

This Assignment and Assumption (the "<u>Assignment and Assumption</u>") is dated as of the Effective Date set forth below and is entered into by and between the Assignor identified in item 1 below (the "<u>Assignor</u>") and the Assignee identified in item 2 below (the "<u>Assignee</u>"). Capitalized terms used but not defined herein shall have the means given to them in the Credit Agreement identified below (as amended, the "<u>Credit Agreement</u>"), receipt of a copy of which is hereby acknowledged by the Assignee. The Standard Terms and Conditions set forth in Annex 1 attached hereto are hereby agreed to and incorporated herein by referenced and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Standard Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by the Agent as contemplated below (i) all of the Assignor's rights and obligations in its capacity as a Lender under the Credit Agreement and any other documents or instruments delivered pursuant thereto to the extent related to the amount and percentage interest identified below of all of such outstanding rights and obligations of the Assignor under the respective facilities identified below (including without limitation any guarantees included in such facilities) and (ii) to the extent permitted to be assigned under applicable Law, all claims, suits, causes of action and any other right of the Assignor (in its capacity as a Lender) against any Person, whether known or unknown, arising under or in connection with the Credit Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby or in any way based on or related to any of the foregoing, including, but not limited to, contract claims, tort claims, malpractice claims, statutory claims and all other claims at law or in equity related to the rights and obligations sold and assigned pursuant to clause (i) above (the rights and obligations sold and assigned by the Assignor to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

B-1

6. Assigned Interest[s]:

Assignor	Assignee	Facility Assigned	Aggregate Amount of Commitment/Loans for all	Amount of Commitment/ Loans Assigned	Percentage Assigned f Commitment/ Loans ^{2]}
			Lenders ¹	_	Communent Louis
		Facility A Revolving	\$	\$	%
		Loans/Commitment			
		Facility A Revolving	\$	\$	%
		Loans/Commitment			
		Term Loan/Commitment	\$	\$	%

1 Amount to be adjusted by the counterparties to take into account any payments or prepayments made between the Trade Date and the Effective Date.

B-2	в	-2
-----	---	----

² Set forth, to at least 9 decimals, as a percentage of the Commitment/Loans of all Lenders thereunder.

Effective Date: ______, 20___ [TO BE INSERTED BY AGENT AND WHICH SHALL BE THE EFFECTIVE DATE OF RECORDATION OF TRANSFER IN THE REGISTER THEREFOR.]

The terms set forth in this Assignment and Assumption are hereby agreed to:

ASSIGNOR [NAME OF ASSIGNOR]

By:____ Title:

ASSIGNEE [NAME OF ASSIGNEE]

By:___ Title:

Consented to and³ Accepted:

EAST WEST BANK, as Agent

By____ Title:

³ To be added only if the consent of the Agent is required by the terms of the Credit Agreement.

B-3

CREDIT AGREEMENT, dated as of June 30, 2015, by and among Applied Optoelectronics, Inc., the Lenders referred to therein and East West Bank, as Agent

STANDARD TERMS AND CONDITIONS FOR ASSIGNMENT AND ASSUMPTION

1. <u>Representations and Warranties</u>.

1.1 <u>Assignor</u>. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any Lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Credit Agreement or any other Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral thereunder, (iii) the financial condition of the Borrower, any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Document.

1.2. <u>Assignee</u>. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it meets all the requirements to be an assignee under Section 10.6(b)(iii), (v) and (vi) of the Credit Agreement (subject to such consents, if any, as may be required under Section 10.6(b)(iii) of the Credit Agreement), (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement as a Lender thereunder and, to the extent of [the][the relevant] Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it is sophisticated with respect to decisions to acquire assets of the type represented by the Assigned Interest and either it, or the person exercising discretion in making its decision to acquire the Assigned Interest, is experienced in acquiring assets of such type, (v) it has received a copy of the Credit Agreement, and has received or has been accorded the opportunity to receive copies of the most recent financial statements delivered pursuant to Section 6.1 thereof, as applicable, and such other documents and information as it deems appropriate to make its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest, (vi) it will, independently and without reliance upon the Agent or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender.

C-1

1.3. <u>Payments</u>. From and after the Effective Date, the Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assignor for amounts which have accrued to but excluding the Effective Date and to the Assignee for amounts which have accrued from and after the Effective Date.¹ Notwithstanding the foregoing, the Agent shall make all payments of interest, fees or other amounts paid or payable in kind from and after the Effective Date to the Assignee.

1.4. <u>General Provisions</u>. This Assignment and Assumption shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment and Assumption may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment and Assumption by telecopy shall be effective as delivery of a manually executed counterpart of this Assignment and Assumption. This Assignment and Assumption shall be governed by, and construed in accordance with, the law of the Commonwealth of Pennsylvania [confirm that choice of law provision parallels the Credit Agreement].

C-2

¹ The Administrative Agent should consider whether this method conforms to its systems. In some circumstances, the following alternative language may be appropriate:

[&]quot;From and after the Effective Date, the Administrative Agent shall make all payments in respect of [the][each] Assigned Interest (including payments of principal, interest, fees and other amounts) to [the][the relevant] Assignee whether such amounts have accrued prior to, on or after the Effective Date. The Assignor[s] and the Assignee[s] shall make all appropriate adjustments in payments by the Administrative Agent for periods prior to the Effective Date or with respect to the making of this assignment directly between themselves."

EXHIBIT C

NOTICE OF LOAN REQUEST

[Date]_____, 20____

East West Bank, as Agent One Oxford Centre 9300 Flair Drive, 6th Floor El Monte, CA 91731 Attention: <u>Loan Servicing Department</u>

Ladies and Gentlemen:

The undersigned, [_____], refers to the Credit Agreement, dated as of June 30, 2015 among the undersigned, the LENDERS party thereto and EAST WEST BANK, as Agent (together with its respective successors and assigns, the "<u>Agent</u>"), (as the same may from time to time be amended, restated or otherwise modified, the "<u>Credit Agreement</u>", the terms defined therein being used herein as therein defined), and hereby gives you notice, pursuant to <u>Section 3.2</u> of the Credit Agreement that the undersigned hereby requests a [Facility [A][B] Revolving][Term] Loan under the Credit Agreement, and in connection therewith sets forth below the information relating to the Loan (the "<u>Proposed Loan</u>") as required by <u>Section 3.2</u> of the Credit Agreement:

(a) The Business Day of the Proposed Loan is _____, 20__.

(b) The amount of the Proposed Loan is \$_____

(c) The Proposed Loan is to be a [LIBOR Rate Loan] [Base Rate Loan].

The undersigned hereby certifies on behalf of Borrower that the following statements are true on the date hereof, and will be true on the date of the Proposed Loan:

(i) the representations and warranties contained in <u>Article IV</u> of the Credit Agreement or in any other Loan Document (whether made by the Borrower or another Loan Party) are correct in all material respects on and as of the date of the Proposed Loan, before and after giving effect to the Proposed Loan, as though made on and as of such date (except to the extent such representations and warranties relate solely to an earlier date);

(ii) no (i) Default of which the Borrower has knowledge or (ii) Event of Default has occurred and is continuing, or would result from the Proposed Loan; and

(iii) the conditions set forth in Section 2.[1][2] and Section 5.2 of the Credit Agreement have been satisfied.

C-3

Very truly yours,

APPLIED OPTOELECTRONICS, INC.

By:	
Name:	
Title:	

\sim	1
ι.	-4

EXHIBIT D

NOTICE OF BORROWING BASE CERTIFICATE

[Date]_____, 20____

East West Bank, as Agent One Oxford Centre 9300 Flair Drive, 6th Floor El Monte, CA 91731 Attention: <u>Loan Servicing Department</u>

Reference is made to the Credit Agreement dated as of June 30, 2015, by and among Applied Optoelectronics, Inc., the Lenders party thereto and East West Bank, as Agent, as amended, restated, supplemented or otherwise modified from time to time (the "<u>Credit Agreement</u>"). Capitalized terms used but not otherwise defined herein shall have the meanings given to them in the Credit Agreement.

The undersigned, [NAME OF SIGNATORY], the [chief financial officer/chief executive officer/treasurer], hereby certifies and warrants that the following schedule and supporting information accurately state the Borrowing Base of the Borrower as of the date hereof:

A. Net Value of Eligible Accounts	\$
B. Reserves against Eligible Accounts	\$
C. Available Eligible Accounts (line A <u>minus</u> line B)	\$
D. Advance rate	85%
E. AR Availability (line C <u>multiplied by</u> line D)	\$
F. Net Value of Eligible Inventory	\$
G. Reserves against Eligible Inventory	\$
H. Available Eligible Inventory (line F <u>minus</u> line G)	\$
I. Advance rate	35%
J. Inventory Availability (line H <u>multiplied by</u> line I)	\$
K. Lesser of (i) \$5,000,000 or (ii) line E <u>multiplied by</u> 50%	\$
L. Lesser of lines J and K	\$
M. Borrowing Base (line E <u>plus</u> line L)	\$

D-1

N. Maximum Facility A Revolving Commitment	\$10,000,000
O. Lesser of lines M and N	\$
P. Aggregate principal amount of Facility A Revolving Loans outstanding	\$
Q. Availability (line O <u>minus</u> line P)	\$

Attached hereto are schedules showing the Eligible Accounts and Eligible Inventory. The undersigned hereby certifies and warrants that the information provided therein is true, complete and correct in all respects as of the date hereof.

D-2

The foregoing certifications, together with the schedules annexed hereto and made a part hereof, are made and delivered this [______ day of _____] pursuant to <u>Section 6.1[(e)][(f)]</u> of the Credit Agreement.

APPLIED OPTOELECTRONICS, INC.

By:	-
Name:	
Title:	_

EXHIBIT E

APPLIED OPTOELECTRONICS, INC.

[FORM OF] REVOLVING CREDIT NOTE

\$[Amount in Numbers]

Sugar Land, TX [Date]

FOR VALUE RECEIVED, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "<u>Borrower</u>"), promises to pay to the order of [______] (the "<u>Lender</u>") the principal amount of [_____] DOLLARS (§[____]) or, if less, the aggregate amount of Facility [A][B] Revolving Loans (as defined in the Credit Agreement referred to below) made by the Lender to the Borrower pursuant to the Credit Agreement referred to below outstanding on the Expiration Date (as defined in such Credit Agreement) or in such amounts on such earlier dates as may be provided in such Credit Agreement.

The Borrower also promises to pay interest on the unpaid principal amount hereof from the date hereof until paid at the rates and at the times determined in accordance with the provisions of the Credit Agreement.

All payments of principal and interest in respect of this Note shall be made in lawful money of the United States of America in same day funds at the office of the Lender described in the Credit Agreement.

This Note is a Revolving Note referred to in, and is entitled to the benefits of, the Credit Agreement, dated as of June 30, 2015 (as amended, modified or supplemented, the "<u>Credit Agreement</u>") between the Borrower, the Agent and the Lenders, which among other things provides for the acceleration of the maturity hereof upon the occurrence of certain events and for repayments in certain circumstances and upon certain terms and conditions. Terms defined in the Credit Agreement have the same meanings herein.

This Note is secured by the Security Documents.

The Borrower hereby expressly waives presentment, demand, notice, protest and all other demands and notices in connection with the delivery, acceptance, performance, default or enforcement of this Note and the Agreement, and an action for amounts due hereunder or thereunder shall immediately accrue.

The Lender and the Borrower intend to conform to all applicable laws limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower. Accordingly, notwithstanding any other provision hereof, the Borrower shall not be required to make any payment to or for the account of the Lender, and the Lender shall refund any payment made by the Borrower, to the extent that such requirement or such failure to refund would violate or conflict with mandatory and nonwaivable provisions of applicable law limiting the maximum amount of interest which may be charged or collected by the Lender from the Borrower. To the fullest extent permitted by law, in any action, suit or proceeding pertaining to this Note, the burden of proof, by clear and convincing evidence, shall be on the Borrower to demonstrate that this Paragraph applies to limit any obligation of the Borrower under this Note or to require the Lender to make any refund, or claiming that this Note conflicts with any applicable law limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower, as to each element of such claim.

E-1

This Note shall be governed by, and construed in accordance with, the laws of the State of New York without giving effect to its choice of law principles.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer, as of the date and the place first above written.

APPLIED OPTOELECTRONICS, INC.

By: _____ Title:_____

EXHIBIT F

APPLIED OPTOELECTRONICS, INC.

[FORM OF] TERM NOTE

\$[Amount in Numbers]

Sugar Land, TX [Date]

FOR VALUE RECEIVED, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "<u>Borrower</u>"), promises to pay to the order of [______] (the "<u>Lender</u>") the principal amount of [______] DOLLARS (\$[_____]), or, if less, the aggregate amount of Term Loans (as defined in the Credit Agreement referred to below) made by the Lender to the Borrower pursuant to the Credit Agreement referred to below outstanding on the Term Loan Maturity Date (as defined in the Credit Agreement) or in such amounts on such earlier dates as are provided for in the Credit Agreement.

The Borrower also promises to pay interest on the unpaid principal amount hereof from the date hereof until paid at the rates and at the times determined in accordance with the provisions of the Credit Agreement.

All payments of principal and interest in respect of this Note shall be made in lawful money of the United States of America in same day funds at the office of the Lender described in the Credit Agreement.

This Note is a Term Note referred to in, and is entitled to the benefits of, the Credit Agreement, dated as of June 30, 2015 (as amended, modified or supplemented, the "<u>Credit Agreement</u>") between the Borrower, the Agent and the Lenders, which among other things provides for the acceleration of the maturity hereof upon the occurrence of certain events and for repayments in certain circumstances and upon certain terms and conditions. Terms defined in the Credit Agreement have the same meanings herein.

This Note is secured by the Security Documents.

The Borrower hereby expressly waives presentment, demand, notice, protest and all other demands and notices in connection with the delivery, acceptance, performance, default or enforcement of this Note and the Agreement, and an action for amounts due hereunder or thereunder shall immediately accrue.

The Lender and the Borrower intend to conform to all applicable laws limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower. Accordingly, notwithstanding any other provision hereof, the Borrower shall not be required to make any payment to or for the account of the Lender, and the Lender shall refund any payment made by the Borrower, to the extent that such requirement or such failure to refund would violate or conflict with mandatory and nonwaivable provisions of applicable law limiting the maximum amount of interest which may be charged or collected by the Lender from the Borrower. To the fullest extent permitted by law, in any action, suit or proceeding pertaining to this Note, the burden of proof, by clear and convincing evidence, shall be on the Borrower to demonstrate that this Paragraph applies to limit any obligation of the Borrower under this Note or to require the Lender to make any refund, or claiming that this Note conflicts with any applicable law limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower, as to each element of such claim.

F-1

This Note shall be governed by, and construed in accordance with, the laws of the State of New York without giving effect to its choice of law principles.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer, as of the date and the place first above written.

APPLIED OPTOELECTRONICS, INC.

By: _____ Title:_____

EXHIBIT G

[FORM OF]

GLOBAL INTERCOMPANY NOTE

June 30, 2015

FOR VALUE RECEIVED, each of Applied Optoelectronics, Inc., a Delaware corporation (the "Borrower"), and each of its direct and indirect Subsidiaries (collectively, the "Group Members") which is a party to this Global Intercompany Note (this "Note"), promises to pay to the order of such other Group Member as makes loans to such Group Member (each Group Member which borrows money pursuant to this Note is referred to herein as a "Payor" and each Group Member which makes loans and advances pursuant to this Note is referred to herein as a "Payee"), the aggregate unpaid principal amount of all loans and advances (including trade payables) heretofore and hereafter made by such Payee to such Payor and any other Debt now or hereafter owing by such Payor to such Payee as shown in the books and records of such Payee (the "Intercompany Debt"). All such Intercompany Debt shall be payable at the times, in the locations and in the currency specified in the documents and records relating thereto; provided that, if any of the time, place or currency of payment shall be not be so specified elsewhere, such amounts shall be payable on demand, in immediately available funds at the chief executive office of the Borrower in the United States of America and in lawful money of the United States of America, as applicable, and provided further that, at any time that an Event of Default (as defined in the Credit Agreement defined below) has occurred and is continuing, at all times following the instruction to the Borrower from the Agent (as defined below) such Intercompany Debt shall thereafter be payable on demand, in immediately available funds at the chief executive office of the Parent in the United States of America and in lawful money of the United States of America. The failure to show in books and records any such Intercompany Debt or any error in showing in books and records such Intercompany Debt shall not affect the obligations of any Payor hereunder. Each Payor promises also to pay interest on the unpaid principal amount of all such Intercompany Debt in like money at said location from the date of the incurrence thereof until paid at such rate per annum as shall be agreed upon from time to time by such Payor and such Payee (provided that such rate shall not exceed the maximum lawful interest rate then in effect).

1. Relationship with Other Documents

This Note is the Intercompany Note referred to in the Credit Agreement (as amended, restated, extended, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), dated as of June 30, 2015, among the Borrower, the Lenders (as defined in the Credit Agreement) from time to time party thereto, and East West Bank, National Association, as Agent (as defined in the Credit Agreement) for Lenders. Capitalized terms used but not otherwise defined herein shall have the meanings given them in the Credit Agreement. This Note is subject to the terms of the Loan Documents, and shall be pledged by each Payee that is a Loan Party to the Agent for the benefit of the Secured Parties, as security for such Payee's Obligations under the Loan Documents. Each Payor hereby acknowledges and agrees that the Agent and the other Secured parties may exercise all the rights of the Loan Parties provided in the Loan Documents with respect to this Note.

This Note amends and restates any original notes between and among any Group Members existing on the date hereof (collectively, the "<u>Original</u> <u>Notes</u>"), supersedes and replaces the Original Notes in their entirety, evidences the indebtedness previously evidenced by the Original Notes and does not constitute a payment or novation of such indebtedness. For the avoidance of doubt, all indebtedness incurred and all loans, advances and other extensions of credit from and after the date hereof between any Group Members shall be evidenced by this Note.

This Note shall evidence all Debt owed by any Group Member to any other Group Member, notwithstanding any other loan agreement, promissory note, document or instrument which may govern the terms thereof. To the extent that the terms of any such other agreements, promissory notes, documents or instruments are inconsistent with this Note, this Note shall govern. For so long as any Obligations remain outstanding, this Note may not be and shall not be deemed to be replaced, superseded or in any way modified, without the consent of the Agent.

2. <u>Pledge of Note</u>.

Each party to this Note acknowledges, agrees and covenants that: (a) all right title and interest in this Note of each Payee that is a Loan Party and the Debt of each Payor evidenced hereby owing to any Loan Party shall be pledged to the Agent, for the benefit of the Secured Parties as security for the Obligations of such Loan Party; (b) the original of this Note shall be delivered to the Agent as further security for the Obligations; (c) for so long as any Obligations or Commitments remain outstanding, no party hereto shall agree to amend, modify or terminate this Note, or provide any security or collateral for any Debt evidenced hereby, in each case without the prior written consent of the Agent; (d) subject to clause (c) above, each Payor may continue to deal with each Payee in the ordinary course of business in relation to the intercompany Debt evidenced by this Note until it receives written notice to the contrary from the Agent, and thereafter such Payor will cease to have any right to deal with such Payee in relation to the Note and therefore from that time such Payor should deal only with the Agent; (e) it is authorized to disclose information in relation to the intercompany Debt owed to it hereunder to the Agent on request; (f) if the Agent so requires by notice in writing, it shall pay all monies to which a Loan Party is entitled under the Note directly to the Agent; (and not to the related Payee) until the Agent otherwise agrees in writing; (g) the provisions of this section may only be revoked with the written consent of the Agent; and (h) it has not assigned its rights under this Note to any Person (other than the Agent) or created any other interest (whether by way of security or otherwise) in this Note in favor of any Person (other than the Agent).

3. Subordination of Indebtedness Owed by Loan Parties or Non-Loan Party Borrowers.

G-2

As used herein, "Senior Indebtedness" means in respect of each Payor that is a Loan Party, all Obligations of such Payor. Anything in this Note to the contrary notwithstanding, each Payee agrees that any and all claims of such Payee with respect to the Intercompany Debt evidenced by this Note against any Payor that is a Loan Party (each a "Subordinated Payor"), or against any of their respective properties, shall be subordinate and junior in right of payment, to the extent and in the manner hereinafter set forth, to all of the Senior Indebtedness, as applicable, until all of such Senior Indebtedness has been performed and indefeasibly paid in full in immediately available funds and the Commitments have been terminated, on the following terms:

(i) in the event of any insolvency, reorganization or like proceedings in connection therewith, relative to any Subordinated Payor or to its creditors, as such, or to its property, and in the event of any proceedings for voluntary liquidation, dissolution or other winding up of such Subordinated Payor (except if expressly permitted by the Credit Agreement), whether or not involving insolvency or bankruptcy, then (x) the holders of Senior Indebtedness shall be indefeasibly paid in full in cash in respect of all amounts constituting Senior Indebtedness before any Payee is entitled to receive (whether directly or indirectly), or make any demands for, any payment on account of this Note and (y) until the holders of Senior Indebtedness are indefeasibly paid in full in cash in respect of all amounts constituting Senior Indebtedness, any payment or distribution to which such Payee would otherwise be entitled (other than (x) equity interests or (y) debt securities of such Subordinated Payor that are subordinated, to at least the same extent as this Note, to the payment of all Senior Indebtedness then outstanding (such securities being hereinafter referred to as "<u>Restructured Debt Securities</u>")) shall be made to the holders of Senior Indebtedness;

(ii) no payment or distribution of any kind or character with respect to this Note shall be made by or on behalf of any Subordinated Payor or any other Person on any Subordinated Payor's behalf, to any non-Loan Party if any Event of Default has occurred and is continuing or would result from such payment; provided, however, that a Subordinated Payor may make payments or distributions to a non-Loan Party if such Subordinated Payor receives written notice approving such payment from the Agent; and

(iii) if any payment or distribution of any character, whether in cash, securities or other property in respect of this Note shall (despite these subordination provisions) be received by any Payee in violation of clause (i) or (ii) before all Senior Indebtedness shall have been indefeasibly paid in full in cash, except as expressly permitted by the Credit Agreement and the Indenture, such Payees shall have no right to possession of asset and such payment or distribution shall be held in trust for the benefit of, and shall be paid over or delivered to, the Administrative Agent for the benefit of the applicable Secured Parties, to the extent necessary to pay all Senior Indebtedness of the relevant Subordinated Payor in full in cash.

To the fullest extent permitted by law, no present or future holder of Senior Indebtedness shall be prejudiced in its right to enforce the subordination of this Note by any act or failure to act on the part of any Payor or by any act or failure to act on the part of such holder or any trustee or agent for such holder. Each Payee and each Payor hereby agree that the subordination of this Note is for the benefit of the Agent and the other Secured Parties are obligees under this Note to the same extent as if their names were written herein as such and each of the Agent, on behalf of itself and the other Secured Parties may proceed to enforce the subordination provisions herein.

G-3

Nothing contained in the subordination provisions set forth above is intended to or will impair, as between each Payor and each Payee, the obligations of such Payor, which are absolute and unconditional, to pay to such Payee the principal of and interest on this Note as and when due and payable in accordance with its terms, or is intended to or will affect the relative rights of such Payee and other creditors of such Payor other than the holders of Senior Indebtedness.

4. Additional Parties.

From time to time after the date hereof, additional Subsidiaries of the Group Members may become parties hereto by executing a counterpart signature page to this Note. Upon delivery of such counterpart signature page to the Payees, notice of which is hereby waived by the other Payors, each additional Group Member shall be a Payor and a Payee and shall be as fully a Payor and a Payee party hereto as if such additional Group Member were an original Payor and an original signatory hereof. So long as this Note is pledged in favor of the Agent, the original of any such counterpart shall be delivered to the Agent. Each Payor expressly agrees that its obligations arising hereunder shall not be affected or diminished by the addition or release of any other Payor hereunder. This Note shall be fully effective as to any Payor that is or becomes a party hereto regardless of whether any other Person becomes or fails to become or ceases to be a Payor hereunder.

5. Miscellaneous.

Each Payor and any endorser hereby waives presentment, demand, protest or notice of any kind in connection with this Note. All payments under this Note shall be made without offset, counterclaim or deduction of any kind.

Each Payee is hereby authorized to record all loans and advances made by it to any Payor (all of which shall be evidenced by this Note), and all repayments or prepayments thereof, in its books and records, such books and records constituting prima facie evidence of the accuracy of the information contained therein.

THIS NOTE AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS NOTE SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF TEXAS.

This Note may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed signature page of this Note by facsimile transmission or other electronic transmission shall be effective as delivery of a manually executed counterpart hereof.

[The Remainder of this Page Intentionally Left Blank]

IN WITNESS WHEREOF, each Payor and Payee has caused this Note to be duly executed and delivered by its proper and duly authorized officer as of the date set forth above.

Applied Optoelectronic, Inc., a Delaware corporation

By

Name: Title:

Prime World International Holdings, Ltd., a company registered in the British Virgin Islands

By

Name: Title:

Global Technology, Inc., a company registered in China

By

Name: Title:

[Signature Page to Global Intercompany Note]

G-5

ALLONGE

This allonge is intended to be attached to and made part of that certain Global Intercompany Note, dated as of June 30, 2015, by and among Applied Optoelectronics, Inc., a Delaware corporation, and each of its subsidiaries (as further amended, restated, amended and restated, supplemented, restructured or otherwise modified from time to time, the "<u>Note</u>"), and, when so attached, shall constitute an endorsement thereof. Capitalized terms used herein but not otherwise defined herein shall have the meanings given to them in the Note.

The undersigned are the Loan Parties (as defined in the Credit Agreement) on the date of the Note. From time to time after the date hereof, additional Loan Parties may become a signatory to this allonge by executing a counterpart signature page to this allonge. Upon delivery of such counterpart signature page, each such additional Loan Party shall be a signatory to this allonge as if such additional Loan Party were an original signatory hereof. Each Loan Party expressly agrees that its obligations arising under the Note and hereunder shall not be affected or diminished by the addition or release of any other Loan Party under the Note or hereunder. This allonge shall be fully effective as to any Loan Party that is or becomes a signatory hereto regardless of whether any other Person becomes or fails to become or ceases to be a Loan Party under the Note or hereunder.

FOR VALUE RECEIVED, each of the undersigned does hereby sell, assign and transfer all of its right, title and interest in and to the Note, PAYABLE TO THE ORDER OF

Dated: _____

Applied Optoelectronic, Inc., a Delaware corporation

By

Name: Title:

G-6

SECURITY AGREEMENT

THIS SECURITY AGREEMENT (this "<u>Agreement</u>"), dated as of June 30, 2015 and, by and between APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "<u>Borrower</u>"), and each Additional Grantor that may become a party hereto after the date hereof in accordance with Section 6.16 hereof (each of the Borrower and each Additional Grantor being a "<u>Grantor</u>" and collectively the "<u>Grantors</u>") and EAST WEST BANK, as Agent for and representative of the Lender Parties (as defined below) (in such capacity, the "<u>Agent</u>"). If there are no Guarantors that are party to this Agreement, the term "Grantors" shall refer only to the Borrower.

RECITALS

A. The Borrower, the Lenders party thereto from time to time (the "<u>Lenders</u>"), and the Agent have entered into a Credit Agreement, dated as of the date hereof (as amended, modified or supplemented from time to time, the "<u>Credit Agreement</u>").

NOW, THEREFORE, intending to be legally bound hereby, the Grantors agree as follows:

ARTICLE I DEFINITIONS

1.1 Definitions. Capitalized terms not otherwise defined herein shall have the meanings given in the Credit Agreement. In addition to the other terms defined elsewhere in this Agreement, as used herein the following terms shall have the following meanings:

"Additional Grantor": Any Person that becomes a party hereto after the date hereof as an additional Grantor by executing a Supplement.

"Excluded Collateral": Any or all (i) Equipment purchased by a Grantor on behalf of a Foreign Branch prior to the Closing or by a Foreign Branch directly and kept on the books and records of a Foreign Branch to be used and located in Taiwan, (ii) with the consent of the Agent, Equipment of a Grantor located in the United States to be transferred to and kept on the books and records of a Foreign Branch and relocated to Taiwan, (iii) Inventory produced and located in Taiwan and kept on the books and records of a Foreign Branch, and (iv) Inventory of a Grantor located in the United States that is sold or otherwise transferred to, and kept on the books and records of, a Foreign Branch in compliance with Section 7.9 of the Credit Agreement and relocated to Taiwan. For the avoidance of doubt, to the extent that any Grantor loans or advances any amount to a Foreign Branch for the purpose of acquiring Excluded Collateral, such loan or advance must be in compliance with Section 7.7 of the Credit Agreement.

"Lender Parties": The Agent and the Lenders.

"<u>Pledged Debt</u>": Indebtedness from time to time owed to a Grantor, the instruments and certificates evidencing such Indebtedness and all interest, cash or other property received, receivable or otherwise distributed in respect of or exchanged therefor.

"<u>Pledged Equity</u>": (i) All Equity Interests now or hereafter owned by a Grantor, including all securities convertible into, and rights, warrants, options and other rights to purchase or otherwise acquire, any of the foregoing, including those owned on the date hereof and set forth on <u>Schedule</u> <u>1</u> annexed hereto, (ii) the certificates or other instruments representing any of the foregoing and any interest of a Grantor in the entries on the books of any securities intermediary pertaining thereto, (iii) all distributions, dividends and other property received, receivable or otherwise distributed in respect of or exchanged therefor, (iv) all capital and interests in profits, losses and assets, at any time represented by any Pledged Equity; (v) all other payments due or to become due in respect of any Pledged Equity, whether under any partnership agreement, liability company agreement or otherwise, whether as contractual obligations, damages, insurance proceeds or otherwise; and (vi) all of its claims, rights, powers, privileges, authority, options, security interests, liens and remedies, if any, under any partnership agreement, limited liability company agreement or otherwise in respect of such Pledged Equity.

"<u>Secured Obligations</u>": All Obligations, and (a) with respect to the Borrower, (i) all obligations from time to time of the Borrower to any Lender Party, including those under or in connection with any Loan Document, Hedging Contracts and Cash Management Agreements; and (b) with respect to each other Grantor, all obligations from time to time of such Grantor to the Lender Parties under or in connection with the Guaranty or any other Loan Document; including, in each case, all obligations to pay principal, interest, fees, expenses, indemnities or other amounts, in each case whether such obligations are direct or indirect, secured or unsecured, joint or several, absolute or contingent, due or to become due, whether for payment or performance, now existing or hereafter arising (including interest and other obligations arising or accruing after the commencement of any bankruptcy, insolvency, reorganization, dissolution or similar proceeding with respect to the Borrower or any other Grantor or any other Person, or which would have arisen or accrued but for the commencement of such proceeding, even if such obligation or the claim therefor is not enforceable or allowable in such proceeding).

"Securities Collateral": The Pledged Equity, the Pledged Debt and any other Investment Property in which Grantor has an interest.

"<u>Supplement</u>": A supplement to this Agreement, in substantially the form set forth as Exhibit A attached hereto, entered into pursuant to Section 6.16 hereof.

"UCC": The Uniform Commercial Code as in effect in the State of New York.

ARTICLE II SECURITY

2.1 Grant of Security. Each Grantor hereby assigns to Agent, and hereby grants to Agent, for the benefit of the Lender Parties, a security interest in, all of such Grantor's right, title and interest in and to all of the personal property of such Grantor including the following, in each case whether now or hereafter existing, whether tangible or intangible, whether now owned or hereafter acquired and wherever the same may be located (the "<u>Collateral</u>"):

- (a) all Accounts;
- (b) all Chattel Paper;
- (c) all Money and all Deposit Accounts, together with all amounts on deposit from time to time in such Deposit Accounts;
- (d) all Documents;
- (e) all General Intangibles (including patents, trademarks, service marks, copyrights, and other intellectual property), Payment Intangibles and Software;
- (f) all Goods, including Inventory, Equipment and Fixtures;
- (g) all Instruments;
- (h) all Investment Property;

- (i) all Letter-of-Credit Rights and other Supporting Obligations;
- (j) all Records;
- (k) all Commercial Tort Claims; and
- (l) all Proceeds and Accessions with respect to any of the foregoing Collateral.

Each category of Collateral set forth above shall have the meaning set forth in the UCC, it being the intention of the Grantors that the description of the Collateral set forth above be construed to include the broadest possible range of assets.

2.2 Security for Obligations. This Agreement secures, and the Collateral is collateral security for, the prompt payment in full when due, whether at stated maturity, by required prepayment, declaration, acceleration, demand or otherwise, of all Secured Obligations of each Grantor.

2.3 Exclusions.

(a) <u>Security Interest Prohibited</u>. Notwithstanding anything herein to the contrary, in no event shall the Collateral include, and each Grantor shall not be deemed to have granted a security interest in, any of such Grantor's rights or interests in or under, any license, contract, permit, Instrument, security or franchise or any of its rights or interests thereunder to the extent, but only to the extent, that such a grant would, under the terms of such license, contract, permit, Instrument, security or franchise, result in a breach of the terms of, or constitute a default under, such license, contract, permit, Instrument, security or franchise (other than to the extent that any such term would be rendered ineffective pursuant to the UCC or any other applicable law or principles of equity); <u>provided</u>, that immediately upon the ineffectiveness, lapse or termination of any such provision the Collateral shall include, and such Grantor shall be deemed to have granted a security interest in, all such rights and interests as if such provision had never been in effect.

(b) <u>Stock of Foreign Subsidiaries</u>. Notwithstanding the foregoing, the Collateral shall not include any Equity Interests issued by a Person if such Person is a controlled foreign corporation (as such term is defined in Section 957(a) of the Internal Revenue Code of 1986, as amended) to the extent that creation of a security interest by a Grantor in such Equity Interests could reasonably be expected to result in material adverse tax consequences to such Grantor or the Borrower, it being acknowledged and agreed that the creation of a security interest in Equity Interests possessing more than 65% of the voting power of all classes of Equity Interests of such Person entitled to vote would result in such material adverse tax consequences.

(c) <u>Prohibited by Law</u>. Notwithstanding anything herein to the contrary, in no event shall the Collateral include, and each Grantor shall not be deemed to have granted a security interest in, any of such Grantor's rights or interests in any property to the extent that such grant of a security interest is prohibited by any requirements of any Law or requires a consent not obtained of any Governmental Authority pursuant to such requirement (other than to the extent that any such requirement would be rendered ineffective pursuant to the UCC or any other applicable Law (including the United States Bankruptcy Code) or principles of equity); <u>provided</u>, that immediately upon the ineffectiveness, lapse or termination of any such requirement the Collateral shall include, and the relevant Grantor shall be deemed to have granted a security interest in, all such rights and interests as if such provision had never been in effect. In the event that any asset of any Grantor is excluded from the Collateral by virtue of the foregoing, such Grantor agrees to use all reasonable efforts to obtain all requisite consent to enable Grantor to provide a security interest in such asset pursuant thereto as promptly as practicable.

(d) <u>Excluded Collateral</u>. Notwithstanding anything herein to the contrary, in no event shall the Collateral include, and each Grantor shall not be deemed to have granted a security interest in, any of such Grantor's rights or interests in any Excluded Collateral.

ARTICLE III PERFECTION AND OTHER COLLATERAL MATTERS

3.1 Perfection.

(a) <u>Generally</u>. Each Grantor agrees that from time to time, at the expense of Grantors, such Grantor will promptly execute and deliver all further instruments and documents, and take all further action, that may be necessary or desirable, or that the Agent may request, in order to perfect and protect any security interest granted or purported to be granted hereby or to enable the Agent to exercise and enforce its rights and remedies hereunder with respect to any Collateral. Without limiting the generality of the foregoing, each Grantor will:

(i) (A) execute (if necessary) and file such financing or continuation statements, or amendments thereto, (B) execute and deliver, and cause to be executed and delivered, agreements establishing that the Agent has control of electronic Chattel Paper, Deposit Accounts, Investment Property and Letter-of-Credit Rights of such Grantor, (C) deliver to the Agent all certificates or Instruments representing or evidencing Investment Property, accompanied by duly executed endorsements or instruments of transfer or assignment in blank, all in form and substance satisfactory to the Agent and (D) deliver such other instruments or notices, in each case, as may be necessary or desirable, or as the Agent may request, in order to perfect and preserve the security interests granted or purported to be granted hereby;

(ii) furnish to the Agent from time to time statements and schedules further identifying and describing the Collateral and such other reports in connection with the Collateral as the Agent may reasonably request, all in reasonable detail;

(iii) at any reasonable time, upon request by Agent, exhibit, to the extent reasonable, the Collateral to and allow inspection of the Collateral by the Agent or any Lender, or persons designated by the Agent or any Lender;

(iv) at the Agent's request, appear in and defend any action or proceeding that may affect such Grantor's title to or the Agent's security interest in all or any part of the Collateral;

(v) use commercially reasonable efforts to obtain any necessary consents of third parties to the creation and perfection of a security interest in favor of the Agent with respect to any Collateral; and

(vi) at the request of the Agent, take any and all actions required to perfect its security interest in titled vehicles.

(b) <u>Financing Statements, etc.</u> Each Grantor hereby authorizes the Agent to file one or more financing or continuation statements, and amendments thereto, relating to any Collateral without the signature of the Grantor where permitted by law, including financing statements that indicate that the Collateral is "all assets" of such Grantor or words to similar effect. A photocopy or other reproduction of this Agreement or any financing statement covering any Collateral shall be sufficient as a financing statement where permitted by law.

(c) <u>Waiver by Landlords, Mortgagees, etc.</u> To the extent that any Collateral at any time is located on premises that are leased from, or otherwise belonging to, any Person other than a Grantor, or that is subject to a mortgage or other lien in favor of any Person other than a Grantor, the Grantors shall provide the Agent with a waiver agreement in form and substance satisfactory to the Agent, duly executed by such landlord, mortgagee or other Person. Any such executed waiver agreement with respect to any location leased on the Closing Date shall be provided to the Agent within thirty (30) days the Closing Date (or such longer time as the Agent may agree in its sole discretion).

(d) <u>Commercial Tort Claims</u>. Each Grantor agrees that it will promptly notify the Agent in writing of any Commercial Tort Claim with claims for damages (whether stated or anticipated by any Grantor), individually or in the aggregate, in excess of \$750,000 constituting Collateral and from time to time, at the expense of Grantors, such Grantor will promptly execute and deliver all further instruments and documents, and take all further action, that may be necessary, or that the Agent may reasonably request, in order to perfect and protect any security interest granted or purported to be granted hereby with respect to such Commercial Tort Claim. The Grantors hereby represent that, as of the Closing Date, no Grantor has any Commercial Tort Claims.

3.2 Matters Relating to Accounts. Except as otherwise provided in this Section, each Grantor shall continue to collect, at its own expense, all amounts due or to become due to such Grantor under the Accounts. In connection with such collections, each Grantor may take (and, upon the occurrence and during the continuance of an Event of Default at the Agent's direction, shall take) such action as such Grantor or the Agent may deem necessary or advisable to enforce collection of amounts due or to become due under the Accounts; <u>provided</u>, that the Agent shall have the right at any time, upon the occurrence and during the continuation of an Event of Default and upon written notice to such Grantor of its intention to do so, to:

(a) notify the account debtors or obligors under any Accounts of the assignment of such Accounts to the Agent and to direct such account debtors or obligors to make payment of all amounts due or to become due to such Grantor thereunder directly to the Agent,

(b) notify each Person maintaining a lockbox or similar arrangement to which account debtors or obligors under any Accounts have been directed to make payment to remit all amounts representing collections on checks and other payment items from time to time sent to or deposited in such lockbox or other arrangement directly to the Agent,

(c) enforce collection of any such Accounts at the expense of Grantors, and

(d) adjust, settle or compromise the amount or payment thereof, in the same manner and to the same extent as such Grantor might have done.

After receipt by such Grantor of the notice from the Agent referred to in the proviso to the preceding sentence, (i) all amounts and proceeds (including checks and other Instruments) received by such Grantor in respect of the Accounts shall be received in trust for the benefit of the Agent on behalf of the Lender Parties hereunder, shall be segregated from other funds of such Grantor and shall be forthwith paid over or delivered to the Agent in the same form as so received (with any necessary endorsement), and (ii) such Grantor shall not, without the written consent of the Agent, adjust, settle or compromise the amount or payment of any Account, or release wholly or partly any account debtor or obligor thereof, or allow any credit or discount thereon.

3.3 Matters Relating to Securities Collateral.

(a) <u>Form of Securities Collateral</u>. If any Securities Collateral is not a security pursuant to Section 8-103 of the UCC, Grantor shall not take any action that, under such Section, converts such Securities Collateral into such a security without causing the issuer thereof to promptly comply with Section 3.1(b).

(b) <u>Voting and Distributions</u>. (i) So long as no Event of Default, shall have occurred and be continuing, (A) each Grantor shall be entitled to exercise any and all voting and other consensual rights pertaining to the Securities Collateral for any purpose not prohibited by the terms of this Agreement or the Credit Agreement; <u>provided</u>, such Grantor shall not exercise or refrain from exercising any such right if the Agent shall have notified Grantor that, in the Agent's judgment, such action would have a material adverse effect on the value of the Securities Collateral or any part thereof; and (B) each Grantor shall be entitled to receive and retain any and all dividends, other distributions, principal and interest paid in respect of the Securities Collateral in compliance with the Credit Agreement.

(ii) Upon the occurrence and during the continuation of an Event of Default (A) upon written notice from the Agent to Grantor, all rights of Grantor to exercise the voting and other consensual rights which it would otherwise be entitled to exercise pursuant hereto shall cease, and all such rights shall thereupon become vested in the Agent who shall have the sole right to exercise such voting and other consensual rights; (B) all rights of Grantor to receive the dividends, other distributions, principal and interest payments which it would otherwise be authorized to receive and retain shall cease, and all such rights shall become vested in the Agent who shall thereupon have the sole right to receive and hold as Collateral such dividends, other distributions, principal and interest payments; and (C) all dividends, principal, interest payments and other distributions which are received by Grantor contrary to the provisions of clause (B) shall be received in trust for the benefit of the Agent, shall be segregated from other funds of Grantor and shall forthwith be paid over to the Agent as Collateral in the same form as so received (with any necessary endorsements).

(iii) In order to permit the Agent to exercise the voting and other consensual rights which it may be entitled to exercise pursuant hereto and to receive all dividends and other distributions which it may be entitled to receive hereunder, (A) Grantor shall promptly execute and deliver (or cause to be executed and delivered) to the Agent all such proxies, dividend payment orders and other instruments as the Agent may from time to time reasonably request, and (B) without limiting the effect of clause (A) above, Grantor hereby grants to the Agent an irrevocable proxy to vote the Pledged Equity and to exercise all other rights, powers, privileges and remedies to which a holder of the Pledged Equity would be entitled, which proxy shall be effective, automatically and without the necessity of any further action.

(c) <u>Subsequently Acquired Collateral</u>. If any Grantor shall acquire (by purchase, dividend, distribution or otherwise) any additional Securities Collateral at any time or from time to time after the date hereof, such Securities Collateral shall automatically (and without any further action being required to be taken) be subject to the pledge and security interests created pursuant to Section 2.1 and, furthermore, such Grantor will promptly thereafter (i) take (or cause to be taken) all action with respect to such Securities Collateral in accordance with the procedures set forth in Section 3.1, and (ii) deliver to the Agent (A) a certificate executed by a principal executive officer of such Grantor listing such Securities Collateral and certifying that the same has been duly pledged in favor of the Agent, on behalf of the Lender Parties, and (B) supplements to Schedule 1 hereto as are necessary to cause such annexes to be complete and accurate at such time.

(d) Grantors agree that they will at all times own and control 100% of the Equity Interests of the Borrower's Subsidiaries.

(e) <u>Transfer Taxes</u>. Each pledge of Securities Collateral under Section 2.1 or 3.4 shall be accompanied by any transfer tax stamps or similar items and all related fees or taxes required in connection with the pledge of such Securities Collateral.

3.4 Matters Relating to Intellectual Property Collateral.

(a) A true and complete list of all registered Intellectual Property and applications for any Intellectual Property owned, held (whether pursuant to a license or otherwise) or used by any Grantor, in whole or in part, is set forth on <u>Schedule 2</u> annexed hereto; and after reasonable inquiry, nor Grantor is aware of any pending or threatened claim by any third party that any of the Intellectual Property owned, held or used by such Grantor is invalid or unenforceable.

(b) If any Grantor, either itself or through any agent, employee, licensee or designee, files an application for the registration of any patent, trademark or copyright with the United States Patent and Trademark Office, the United States Copyright Office or any similar office or agency in the United States, the applicable Grantor shall, within 30 days of such filing, give the Agent written notice thereof, and, upon request of the Agent, such Grantor shall execute and deliver any and all patent security agreements, copyright security agreements or trademark security agreements with respect to such registered Intellectual Property as the Agent may request to evidence the Agent's Lien on such patent, trademark or copyright, and the General Intangibles of such Grantor relating thereto or represented thereby.

(c) Each Grantor shall take all actions necessary or reasonably requested by the Agent to maintain and pursue each application, to obtain the relevant registration and to maintain the registration of each of the patents, trademarks and copyrights that are owned by the applicable Grantor and that such Grantor reasonably determines are material to the conduct of its business or operations (now or hereafter existing), including the filing of applications for renewal, affidavits of use, affidavits of noncontestability and opposition and interference and cancellation proceedings.

(d) In the event any Grantor determines that any of the patent, trademark or copyright Collateral owned by such Grantor is infringed upon, or misappropriated or diluted by a third party, the applicable Grantor shall notify the Agent promptly after becoming aware of such infringement, misappropriation or dilution. The applicable Grantor shall, unless such Grantor shall reasonably determine that such patent, trademark or copyright Collateral is not material to the conduct of its business or operations, promptly exercise enforcement remedies against such infringement, misappropriation or dilution and to recover any and all damages for such infringement, misappropriation or dilution, and shall take such other actions as the Agent shall deem reasonably appropriate under the circumstances to protect such patent, trademark or copyright Collateral.

(e) In addition to, and not by way of limitation of, the granting of a security interest in the Collateral pursuant hereto, each Grantor, effective upon the occurrence and during the continuance of an Event of Default, hereby assigns, transfers and conveys to the Agent the nonexclusive right and license to use all trademarks, tradenames, copyrights, patents or technical processes (including, without limitation, any Intellectual Property constituting Collateral) owned or used by Grantor that relate to the Collateral, together with any goodwill associated therewith, all to the extent necessary to enable the Agent to realize on the Collateral in accordance with this Agreement and to enable any transferee or assignee of the Collateral to enjoy the benefits of the Collateral. This right shall inure to the benefit of all successors, assigns and transferees of the Agent and its successors, assigns and transferees, whether by voluntary conveyance, operation of law, assignment, transfer, foreclosure, deed in lieu of foreclosure or otherwise. Such right and license shall be granted free of charge, without requirement that any monetary payment whatsoever be made to any Grantor. If and to the extent that any Grantor is permitted to license any Intellectual Property constituting Collateral, the Agent shall promptly enter into a non disturbance agreement or other similar arrangement, at such Grantor's request and expense, with such Grantor and any licensee of any such Intellectual Property permitted hereunder in form and substance reasonably satisfactory to the Agent pursuant to which (i) the Agent shall agree not to disturb or interfere with such licensee's rights under its license agreement with such Grantor so long as such licensee is not in default thereunder, and (ii) such licensee shall acknowledge and agree that such Intellectual Property licensed to it is subject to the security interest created in favor of the Agent and the other terms of this Agreement.

ARTICLE IV CERTAIN RIGHTS OF AGENT

4.1 Agent Appointed Attorney-in-Fact. Each Grantor hereby irrevocably appoints the Agent as such Grantor's attorney-in-fact, with full authority in the place and stead of such Grantor and in the name of such Grantor, the Agent or otherwise, from time to time in the Agent's discretion to take any action and to execute any instrument that the Agent may deem necessary or advisable to accomplish the purposes of this Agreement, including, without limitation:

(a) to obtain and adjust insurance required to be maintained by such Grantor;

(b) to ask for, demand, collect, sue for, recover, compound, receive and give acquittance and receipts for moneys due and to become due under or in respect of any of the Collateral;

(c) to receive, endorse and collect any drafts or other Instruments, Documents, Chattel Paper and other documents in connection with clauses (a) and (b) above;

(d) to file any claims or take any action or institute any proceedings that the Agent may deem necessary or desirable for the collection of any of the Collateral or otherwise to enforce or protect the rights of the Agent with respect to any of the Collateral;

(e) to pay or discharge Liens (other than Liens permitted under this Agreement or the Credit Agreement) levied or placed upon or threatened against the Collateral, the legality or validity thereof and the amounts necessary to discharge the same to be determined by the Agent in its sole discretion, any such payments made by the Agent to become obligations of such Grantor to the Agent, due and payable immediately without demand;

(f) to sign and endorse any invoices, freight or express bills, bills of lading, storage or warehouse receipts, drafts against debtors, assignments, verifications and notices in connection with Accounts and other documents relating to the Collateral;

(g) generally to sell, transfer, pledge, make any agreement with respect to or otherwise deal with any of the Collateral as fully and completely as though the Agent were the absolute owner thereof for all purposes; and

(h) to do, at the Agent's option and Grantors' expense, at any time or from time to time, all acts and things that the Agent deems necessary to protect, preserve or realize upon the Collateral and the Agent's security interest therein in order to effect the intent of this Agreement, all as fully and effectively as such Grantor might do;

provided, that except for actions set forth in Paragraphs (e) and (h), such power may be exercised only so long as an Event of Default has occurred and is continuing. All third parties are entitled to rely conclusively on a representation by the Agent that it is entitled to exercise such power of attorney.

4.2 Standard of Care. The powers conferred on the Agent hereunder are solely to protect its interest in the Collateral and shall not impose any duty upon it to exercise any such powers. Except for the exercise of reasonable care in the custody of any Collateral in its possession and the accounting for moneys actually received and spent by it hereunder, the Agent shall have no duty as to any Collateral or as to the taking of any necessary steps to preserve rights against prior parties or any other rights pertaining to any Collateral. The Agent shall be deemed to have exercised reasonable care in the custody and preservation of Collateral in its possession if such Collateral is accorded treatment substantially equal to that which the Agent accords its own property.

4.3 Agent May Perform. If the Grantor fails to perform any obligation under or in connection with this Agreement, the Agent may (but shall have no duty to) itself perform or cause performance of such obligation, and the expenses of the Agent incurred in connection therewith shall be payable by the Grantor pursuant to Section 6.4. The Agent may from time to time take any other action which the Agent deems necessary or appropriate for the maintenance, preservation or protection of any of the Collateral or of its security interest therein.

ARTICLE V <u>REMEDIES</u>

5.1 Remedies. If any Event of Default has occurred and is continuing, the Agent may exercise in respect of the Collateral, in addition to all other rights and remedies provided for herein or in any other Loan Document or otherwise available to it, all the rights and remedies of an Agent on default under the UCC (whether or not the UCC applies to the affected Collateral), and also may:

(i) require each Grantor to, and each Grantor hereby agrees that it will at its expense and upon request of the Agent forthwith, assemble all or part of the Collateral as directed by the Agent and make it available to the Agent at a place to be designated by the Agent that is reasonably convenient to both parties,

(ii) enter onto the property where any Collateral is located and take possession thereof with or without judicial process,

(iii) prior to the disposition of the Collateral, store, process, repair or recondition the Collateral or otherwise prepare the Collateral for disposition in any manner to the extent the Agent deems appropriate,

(iv) take possession of any Grantor's premises or place custodians in exclusive control thereof, remain on such premises and use the same and any of such Grantor's equipment for the purpose of completing any work in process, taking any actions described in the preceding clause (iii) and collecting any Secured Obligation,

(v) sell the Collateral or any part thereof in one or more parcels at public or private sale, at any of the Agent's offices or elsewhere, for cash, on credit or for future delivery, at such time or times and at such price or prices and upon such other terms as the Agent may deem commercially reasonable,

(vi) exercise dominion and control over and refuse to permit further withdrawals from any Deposit Account maintained with the Agent or any Agent and provide instructions directing the disposition of funds in Deposit Accounts not maintained with the Agent, and

(vii) provide entitlement orders with respect to Security Entitlements and other Investment Property constituting a part of the Collateral and, without notice to any Grantor, transfer to or register in the name of the Agent or any of its nominees any or all of the Collateral constituting Investment Property.

The Agent may be the purchaser of any or all of the Collateral at any such sale and the Agent shall be entitled, for the purpose of bidding and making settlement or payment of the purchase price for all or any portion of the Collateral sold at any such public sale, to use and apply any of the Secured Obligations as a credit on account of the purchase price for any Collateral payable by the Agent at such sale. Each Grantor hereby waives any claims against the Agent arising by reason of the fact that the price at which any Collateral may have been sold at such a private sale was less than the price which might have been obtained at a public sale, even if the Agent accepts the first offer received and does not offer such Collateral to more than one offeree.

(b) Securities Collateral. Grantor recognizes that, by reason of certain prohibitions contained in applicable securities laws, the Agent may be compelled, with respect to any sale of all or any part of the Securities Collateral conducted without prior registration or qualification of such Securities Collateral under the such securities laws, to limit purchasers to those who will agree, among other things, to acquire the Securities Collateral for their own account, for investment and not with a view to the distribution or resale thereof and who meet certain financial criteria. Each Grantor acknowledges that any such private placement may be at prices and on terms less favorable than those obtainable through a sale without such restrictions (including an offering made pursuant to a registration statement under such securities laws) and, notwithstanding such circumstances, each Grantor agrees that any such private placement shall not be deemed, in and of itself, to be commercially unreasonable and that the Agent shall have no obligation to delay the sale of any Securities Collateral for the request, Grantor shall and shall cause each issuer of any Securities Collateral to be sold hereunder from time to time to furnish to the Agent all such information as Agent may request in order to determine the amount of Securities Collateral that may be sold by the Agent in exempt transactions under such securities laws, as the same are from time to time to time in effect.

(c) Intellectual Property Collateral. Anything contained herein to the contrary notwithstanding, upon the occurrence and during the continuation of an Event of Default, (i) the Agent shall have the right (but not the obligation) to bring suit, in the name of any Grantor, the Agent or otherwise, to enforce any Intellectual Property, in which event such Grantor shall, at the request of the Agent, do any and all lawful acts and execute any and all documents reasonably required by the Agent in aid of such enforcement and such Grantor shall promptly, upon demand, reimburse and indemnify the Agent as provided in Section 6.4 hereof, in connection with the exercise of its rights under this Section 5.1, and, to the extent that the Agent shall elect not to bring suit to enforce any Intellectual Property as provided in this Section, such Grantor agrees to use all reasonable measures, whether by action, suit, proceeding or otherwise, to prevent the infringement of any of the Intellectual Property by others and for that purpose agrees to use its commercially reasonable judgment in maintaining any action, suit or proceeding against any Person so infringing reasonably necessary to prevent such infringement; (ii) upon written demand from the Agent, each Grantor shall execute and deliver to the Agent an assignment or assignments of the Intellectual Property and such other documents as are necessary or appropriate to carry out the intent and purposes of this Agreement; (iii) each Grantor agrees that such an assignment and/or recording shall be applied to reduce the Secured Obligations outstanding only to the extent that the Agent receives cash proceeds in respect of the sale of, or other realization upon, the Intellectual Property; and (iv) within five business days after written notice from the Agent, each Grantor shall make available to the Agent, to the extent within such Grantor's power and authority, such personnel in such Grantor's employ as the Agent may reasonably designate, by name, title or job responsibility, to permit such Grantor to continue, directly or indirectly, to produce, advertise and sell the products and services sold or delivered by Grantor under or in connection with the Intellectual Property, such persons to be available to perform their prior functions on the Agent's behalf and to be compensated by the Agent at such Grantor's expense on a per diem, pro rata basis consistent with the salary and benefit structure applicable to each as of the date of such Event of Default.

5.2 Application of Proceeds. Except as expressly provided elsewhere in this Agreement, all proceeds received by the Agent in respect of any sale of, collection from, or other realization upon all or any part of the Collateral may, in the discretion of the Agent, be held by the Agent as collateral for the Secured Obligations, or then or at any time thereafter applied in accordance with Section 8.2 of the Credit Agreement. If and when all Secured Obligations shall have been completely and finally satisfied in cash in full and all commitments to extend credit under the Loan Documents shall have terminated, any surplus of such cash or cash proceeds held by the Agent shall be paid over to the Grantors or as otherwise required by law. The Grantors shall remain liable, jointly and severally, for any deficiency.

ARTICLE VI MISCELLANEOUS

6.1 Amendments, etc. No amendment to or waiver of any provision of this Agreement, nor any consent to any departure by any Grantor herefrom, shall in any event be effective unless in a writing manually signed by the Agent, <u>provided</u>, that this Agreement may be modified by the execution of a Supplement by an Additional Grantor in accordance with Section 6.16 hereof and Grantors hereby waive any requirement of notice of or consent to any such amendment. Any such waiver or consent shall be effective only in the specific instance and for the specific purpose for which it was given.

6.2 No Implied Waiver; Remedies Cumulative. No failure of the Agent to exercise any right or remedy under this Agreement shall operate as a waiver thereof; nor shall any single or partial exercise of any such right or remedy preclude any other or further exercise thereof or the exercise of any other right or remedy. The rights and remedies of the Agent under this Agreement are cumulative and not exclusive of any other rights or remedies available hereunder, under any other agreement, at law, or otherwise.

6.3 Notices. Except to the extent, if any, otherwise expressly provided herein, all notices and other communications (collectively, "notices") under this Agreement shall be made and be effective, in the case of the Borrower or any Lender Party, in accordance with the Credit Agreement, and in the case of any other Grantor, in accordance with its Guaranty. The Agent may rely on any notice (whether or not made in a manner contemplated by this Agreement) purportedly made by or on behalf of a Grantor, and the Agent shall have no duty to verify the identity or authority of the Person giving such notice.

6.4 Indemnity and Expenses.

(a) <u>Indemnity</u>. Without limiting the application of Sections 10.4 and 10.5 of the Credit Agreement, Grantors, jointly and severally, agree to indemnify the Agent from and against any and all claims, losses, liabilities and expenses (including reasonable attorneys' fees) arising out of or resulting from this Agreement (including, without limitation, enforcement of this Agreement), except claims, losses, liabilities and expenses resulting solely from the gross negligence or willful misconduct of the Agent.

(b) <u>Expenses</u>. The Grantors, jointly and severally, agree to pay to the Agent, upon demand, the amount of all reasonable expenses, including the reasonable fees and expenses of its counsel and of any experts and agents, which the Agent may incur in connection with (i) the administration of this Agreement, (ii) the custody, preservation, use or operation of, or the sale of, collection of or other realization upon, any Collateral, (iii) the exercise or enforcement of any of the rights of the Agent hereunder, or (iv) the failure by any Grantor to perform or observe any of the provisions hereof.

6.5 Entire Agreement. This Agreement, together with the Exhibits and the Schedules hereto, and the other Loan Documents, constitutes the entire agreement of the parties hereto with respect to the subject matter hereof and supersedes all prior and contemporaneous understandings and agreements.

6.6 Survival. The obligations of the Grantors under Section 6.4 shall survive the termination of this Agreement and all other events and conditions whatsoever. All representations and warranties of each Grantor contained in or made in connection with this Agreement shall survive, and shall not be waived by, the execution and delivery of this Agreement, any investigation by or knowledge of any Lender Party, any extension of credit, termination of this Agreement or any other event or circumstance whatsoever.

6.7 Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, and all such counterparts shall constitute but one and the same agreement.

6.8 Severability. In case any one or more of the provisions contained in this Agreement should be invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not be in any way be affected or impaired thereby.

6.9 Headings. Section headings in the Agreement are included for convenience of reference only and shall not be given any substantive effect.

6.10 Successors and Assigns. This Agreement shall be binding upon each Grantor, its successors and assigns, and shall inure to the benefit of and be enforceable by the Agent and its successors and assigns. Without limitation of the foregoing, and subject to and in accordance with the terms of the Credit Agreement, each Lender (and any successive assignee or transferee) from time to time may assign or otherwise transfer or create a participation interest in all or any portion of its rights or obligations under the Loan Documents (including all or any portion of any commitment to extend credit), or any Secured Obligations, to any other Person, and such Secured Obligations (including any Secured Obligations resulting from extension of credit by such other Person under or in connection with the Loan Documents) shall be and remain Secured Obligations entitled to the benefit of this Agreement, and to the extent of its interest in such Secured Obligations such other Person shall be vested with all the benefits in respect thereof granted to the Lender Parties in this Agreement or otherwise.

6.11 Binding Effect. This Agreement shall be binding upon and inure to the benefit of Grantors, the Agent and their respective successors and permitted assigns, except that Grantors shall not have the right to assign its rights hereunder or any interest herein without the prior written consent of the Agent.

6.12 <u>Governing Law</u>. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT GIVING EFFECT TO ITS CHOICE OF LAW PRINCIPLES, EXCEPT TO THE EXTENT THAT THE UCC PROVIDES THAT THE PERFECTION OF THE SECURITY INTEREST HEREUNDER, OR REMEDIES HEREUNDER, IN RESPECT OF ANY PARTICULAR COLLATERAL ARE GOVERNED BY THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF NEW YORK.

6.13 <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES TO THIS AGREEMENT HEREBY AGREES TO WAIVE ITS RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS AGREEMENT OR ANY OF THE OTHER LOAN DOCUMENTS OR ANY DEALINGS BETWEEN THEM RELATING TO THE SUBJECT MATTER OF THIS LOAN TRANSACTION OR THE AGENT/DEBTOR RELATIONSHIP THAT IS BEING ESTABLISHED.

6.14 Consent to Jurisdiction; Venue. All judicial proceedings brought against a Grantor with respect to this Agreement and the Loan Documents may be brought in any state or federal court of competent jurisdiction in Los Angeles County, California, and by execution and delivery of this Agreement, each Grantor accepts for itself and in connection with its properties, generally and unconditionally, the nonexclusive jurisdiction of the aforesaid courts, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Agreement. Each Grantor irrevocably waives any right it may have to assert the doctrine of forum non conveniens or to object to venue to the extent any proceeding is brought in accordance with this Section.

6.15 LIMITATION OF LIABILITY. TO THE FULLEST EXTENT PERMITTED BY LAW, NO CLAIM MAY BE MADE BY ANY GRANTOR AGAINST ANY LENDER PARTY OR ANY RELATED PARTY THEREOF FOR ANY SPECIAL, INDIRECT, CONSEQUENTIAL OR PUNITIVE DAMAGES IN RESPECT OF ANY CLAIM ARISING FROM OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR ANY STATEMENT, COURSE OF CONDUCT, ACT, OMISSION OR EVENT IN CONNECTION WITH ANY OF THE FOREGOING (WHETHER BASED ON BREACH OF CONTRACT, TORT OR ANY OTHER THEORY OF LIABILITY); AND EACH GRANTOR HEREBY WAIVES, RELEASES AND AGREES NOT TO SUE UPON ANY CLAIM FOR ANY SUCH DAMAGES, WHETHER OR NOT ACCRUED AND WHETHER OR NOT KNOWN OR SUSPECTED TO EXIST.

6.16 Additional Grantors. The initial Grantors hereunder shall be the Borrower and such of the Grantors as are signatories hereto on the date hereof. From time to time after the date hereof, additional Persons may become Additional Grantors, by executing a Supplement. Upon delivery of any such Supplement to the Agent, notice of which is hereby waived by Grantors, each such Additional Grantor shall be a Grantor and shall be as fully a party hereto as if such Additional Grantor were an original signatory hereto. Each Grantor expressly agrees that its obligations arising hereunder shall not be affected or diminished by the addition or release of any other Grantor hereunder, nor by any election of the Agent not to cause any entity to become an Additional Grantor hereunder. This Agreement shall be fully effective as to any Grantor that is or becomes a party hereto regardless of whether any other Person becomes or fails to become or ceases to be a Grantor hereunder.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, Grantors and the Agent have caused this Agreement to be duly executed and delivered by their respective officers thereunto duly authorized as of the date first written above.

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

EAST WEST BANK, as Agent

By: <u>/s/Stefan J. Murry</u> Name: Kelvin Chan Title: Managing Director

Signature Page to Security Agreenment

SCHEDULE 1 TO SECURITY AGREEMENT

SCHEDULE 2 TO SECURITY AGREEMENT

[FORM OF SUPPLEMENT]

SUPPLEMENT (this "<u>Supplement</u>"), dated as of ______, 20__, is delivered pursuant to Section 6.16 of the Security Agreement referred to below. The undersigned hereby agrees that this Supplement may be attached to the Security Agreement, dated as of June 30, 2015, among the Grantors signatory thereto and East West Bank, as Agent ("<u>Agent</u>"), as it may heretofore have been and as it may hereafter be further amended, modified or supplemented from time to time, the "<u>Security Agreement</u>". Capitalized terms used herein not otherwise defined herein shall have the meanings given in the Security Agreement. The undersigned by executing and delivering this Supplement hereby becomes a Grantor under the Security Agreement in accordance with Section 6.16 thereof and agrees to be bound by all of the terms thereof with the same force and effect as if originally named as a Grantor therein, and each reference to a "Grantor" in the Security Agreement shall be deemed to include the undersigned. Without limiting the generality of the foregoing, the undersigned hereby:

(a) authorizes Agent to add the information set forth on the Schedules of this Agreement to the correlative Schedules attached to the Security Agreement or the Credit Agreement;

(b) represents and warrants that the representations and warranties made by it as a Grantor under the Security Agreement are true and correct on and as of the date hereof;

(c) grants to and creates in favor of Agent, for the benefit of the Lender Parties, as security for the payment of the Secured Obligations a security interest in and lien on all of the undersign's right, title and interest in and to the Collateral.

Except as expressly supplemented hereby, the Security Agreement shall remain in full force and effect in accordance with its terms.

THIS SUPPLEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF CALIFORNIA, EXCLUSIVE OF CHOICE OF LAW PRINCIPLES.

This Supplement may be executed by the parties hereto in any number of counterparts, each of which shall be deemed to be an original, and all such counterparts shall constitute but one and the same agreement.

The address, facsimile number and e-mail address for all notices to be delivered to the undersigned pursuant to the Security Agreement is as set forth on the signature page hereof.

Supp-1

[NAME OF ADDITIONAL GRANTOR]

By:	
Name:	
Title:	
[Address:	
-	

Telephone No.:	
Fax No.:	
E-mail:	

ACCEPTED:

EAST WEST BANK, as Agent

By:_____ Name:_____ Title:_____

Supp-2

Schedule 1

PATENT SECURITY AGREEMENT

This **PATENT SECURITY AGREEMENT** (this "Agreement"), dated as of June 30, 2015 and, by and between, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "Borrower") and each Additional Grantor that may become a party hereto after the date hereof (each of the Borrower and each Additional Grantor being a "Grantor" and collectively the "Grantors") and EAST WEST BANK, as Agent for and representative of the Lender Parties (in such capacity, the "Agent"). If there are no Guarantors that are party to this Agreement, the term "Grantors" shall refer only to the Borrower.

WHEREAS, the Grantor has granted a security interest in and continuing Lien on the Collateral (including the patents subject of this Agreement) to the Agent pursuant to the Security Agreement, and desires to enter into this Agreement for purposes of affirming such grant to the Agent, and providing a short-form medium to facilitate the recording of such security interest in the applicable governmental offices.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Grantor agrees as follows:

Section 1. <u>Affirmation and Grant of Security</u>. The Grantor hereby affirms its grant to the Agent of, and grants to the Agent, a security interest in and continuing lien on all of such Grantor's right, title and interest in, to and under the patents listed on **Schedule I** to secure the Secured Obligations.

Section 2. <u>Recordation</u>. The Grantor authorizes and requests that the Commissioner of Patents and Trademarks and any other applicable government officer record this Agreement.

Section 3. <u>Grants, Rights and Remedies</u>. This Agreement has been entered into in conjunction with the provisions of the Security Agreement. The Grantor does hereby acknowledge and confirm that the grant of the security interest hereunder to, and the rights and remedies of, the Agent with respect to the Collateral are more fully set forth in the Security Agreement, the terms and provisions of which are incorporated herein by reference as if fully set forth herein.

Section 4. <u>Miscellaneous</u>. The Grantor agrees that neither this Agreement nor any term hereof may be changed, waived, discharged or terminated, except by an instrument in writing signed by the party (including, if applicable, any party required to evidence its consent to or acceptance of this Agreement) against whom enforcement of such change, waiver, discharge or termination is sought. This Agreement may be executed in any number of counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.

Section 5. <u>Governing Law</u>. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT GIVING EFFECT TO ITS CHOICE OF LAW PRINCIPLES, EXCEPT TO THE EXTENT THAT THE UCC PROVIDES THAT THE PERFECTION OF THE SECURITY INTEREST HEREUNDER, OR REMEDIES HEREUNDER, IN RESPECT OF ANY PARTICULAR COLLATERAL ARE GOVERNED BY THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF NEW YORK.

IN WITNESS WHEREOF, the Grantor has caused this Patent Security Agreement to be duly executed and delivered by its officer thereunto duly authorized as of the date first above written.

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

(Signature Page to Patent Security Agreement)

SCHEDULE I

ISSUED PATENTS

Patent No.	Country	Title	Issue Date
8,995,484	USA	Temperature controlled multi-channel transmitter optical	03-31-2015
		subassembly and optical transceiver module including same	
8,831,433	USA	Temperature controlled multi-channel transmitter optical	09-09-2014
		subassembly and optical transceiver module including same	
8,891,974	USA	Distortion compensation circuit including tunable phase path	11-18-2014
8,805,191	USA	Optical transceiver including optical fiber coupling assembly to	08-12-2014
		increase usable channel wavelengths	
8,655,183	USA	Optical transceiver that maintains a bend diameter of an internal	02-18-2014
		optical fiber and method of assembling same	
9,002,214	USA	Wavelength-selectable laser device and apparatus and system	04-07-2015
		including same	
8,818,208	USA	Laser mux assembly for providing a selected wavelength	08-26-2014
8,606,116	USA	System and method for distortion compensation in response to	12-10-2013
		frequency detection	
8,670,675	USA	System and method for distortion compensation including	03-11-2014
		configureable delay	
8,718,489	USA	Distortion compensation circuit including one or more phase	05-06-2014
		invertible distortion paths	
8,899,846	USA	Receptacle diplexer	12-02-2014
8,320,773	USA	Reducing cross-modulation in multichannel modulated optical	11-27-2012
		systems	
8,091,876	USA	Fixture for securing optoelectronic packages for wire and/or	01-10-2012
		component bonding	
8,045,164	USA	Position finding system and method for use in aligning laser	10-25-2011
		device with an optical fiber	

8,406,631	USA	Network timing	03-26-2013
7,978,985 USA		Clipping correction system and method for correcting clipped	07-12-2011
		signals in a receiver	
8,165,474	USA	Reducing cross modulation in multichannel modulated optical	04-24-2012
		systems with anti-clipping	
8,787,772	USA	Laser package including semiconductor laser and memory device	07-22-2014
		for storing laser parameters	
8,073,340	USA	Distortion compensation circuit including one or more phase	12-06-2011
		invertible distortion paths	
8,121,493	USA	1	02-21-2012
		time dependent series of distortion signal	
7,766,686	USA	Pluggable form factor release mechanism	08-03-2010
8,121,170	USA	Gain-coupled distributed feedback semiconductor laser including	02-21-2012
		first-order and second-order gratings	
7,925,170	USA	Predistortion circuit including distortion generator diodes with	04-12-2011
		adjustable diode bias	
7,573,923	USA	Laser drive circuit and method providing high limit clipping	08-11-2009
		corresponding to low limit clipping in a laser	
8,358,937	USA	Systems and methods for reducing clipping in multichannel	01-22-2013
		modulated optical systems	
8,165,475	USA	Systems and methods for reducing clipping in multichannel	04-24-2012
		modulated optical systems	
7,627,012	USA	Distributed feedback semiconductor laser including wavelength	12-01-2009
		monitoring section	
7,542,503	USA	Distributed feedback laser with improved optical field uniformity	06-02-2009
		and mode stability	
7,478,955	USA	1 0 5	01-20-2009
7,010,013	USA	Assembly with tapered, threaded ferrule housing for improved	03-07-2006
		alignment of fiber with laser	
6,859,481	USA	Optically-pumped multiple-quantum well active region with	02-22-2005
		improved distribution of optical pumping power	
6,652,707	USA	Method and apparatus for demounting workpieces from adhesive	11-25-2003
		film	
6,795,478	USA	VCSEL with antiguide current confinement layer	09-21-2004

7,026,178	USA	Method for fabricating a VCSEL with ion-implanted current-	04-11-2006
		confinement structure	
6,697,413 USA Ti		Tunable vertical-cavity surface-emitting laser with tuning	02-24-2004
		junction	
6,788,466	USA	Multiple reflectivity band reflector	09-07-2004
6,763,053	USA	Laser having multiple reflectivity band reflector	07-13-2004
6,765,939	USA	Multiple reflectivity band reflector with non-uniform profile and	07-20-2004
		laser system employing same for laser wavelength monitoring	
6,735,224	USA	Planar lightwave circuit for conditioning tunable laser output	05-11-2004
6,763,046	USA	Method and system employing multiple reflectivity band reflecto	r07-13-2004
		for laser wavelength monitoring	
6,611,543	USA	Vertical-cavity surface-emitting laser with metal mirror and	08-26-2003
		method of fabrication of same	
6,549,556	USA	Vertical-cavity surface-emitting laser with bottom dielectric	04-15-2003
		distributed bragg reflector	
8,126,329	USA	Quad-port optical module with pass-through and add/drop	02-28-2012
		configuration	
7,519,260	USA	Coaxial optoelectronic device separation apparatus and method	04-14-2009
8,079,222	USA	Thermoelectric cooler controller	12-20-2011
8,055,138	USA	Photosensor operating point	11-08-2011
7,583,719	USA	Complex-coupled distributed feedback semiconductor laser	09-01-2009
7,468,286	,468,286 USA System and method for securing optoelectronic packages for		12-23-2008
		mounting components at a mounting angle	
6,263,002	USA	Tunable fiber fabry-perot surface-emitting lasers	07-17-2001
6,736,550	USA	Housing for passively aligning an optical fiber with a lens	05-18-2004
6,608,855	USA	Single-mode DBR laser with improved phase-shift section	08-19-2003
6,669,367	USA	Optical fiber with mirror for semiconductor laser	12-30-2003
6,724,796 USA		Modified distributed bragg reflector (DBR) for vertical cavity	04-20-2004
		surface-emitting laser (VCSEL) resonant wavelength tuning	
		sensitivity control	
6,560,265	USA	Method and apparatus for polarizing light in a VCSEL	05-06-2003

6,636,544 USA		Overlapping wavelength-tunable vertical cavity surface-emitting	10-21-2003
		laser (VCSEL) arrays	
6,696,307 USA		Patterned phase shift layers for wavelength-selectable vertical	02-24-2004
		cavity surface-emitting laser (VCSEL) arrays	
6,782,019	USA	VCSEL with heat-spreading layer	08-24-2004
6,471,582	USA	Adapter for coupling air duct to fan-driven vent	10-29-2002
7,010,012	USA	Method and apparatus for reducing specular reflections in	03-07-2006
		semiconductor lasers	
6,765,948	USA	VCSEL assembly with edge-receiving optical devices	07-20-2004
6,560,259	USA	Spatially coherent surface-emitting, grating coupled quantum	05-06-2003
		cascade laser with unstable resonance cavity	
6,746,777	USA	Alternative substrates for epitaxial growth	06-08-2004
6,455,908 USA		Multispectral radiation detectors using strain-compensating	09-24-2002
		superlattices	
6,603,184	USA	Double heterostructure photodiode with graded minority-carrier	08-05-2003
		blocking structures	
6,406,795	USA	Compliant universal substrates for optoelectronic and electronic	06-18-2002
		devices	
6,638,773	USA	Method for fabricating single-mode DBR laser with improved	05-31-2002
		yield	

PATENT APPLICATIONS

Appl. No.	Country	Title	
13/975,867	USA	Heated laser package with increased efficiency for optical transmitter systems	
13/893,802	USA	Compact multi-channel optical transceiver module	
14/088,883	USA	Aligning and directly optically coupling photodetectors to optical	
		demultiplexer outputs in a multichannel receiver optical subassembly	
13/760,533	USA	Thermally shielded multi-channel transmitter optical subassembly and optical	
		transceiver module including same	
13/708,569	USA	Thermally isolated multi-channel transmitter optical subassembly and optical	
		transceiver module including same	
13/709,195	USA	Multi-channel optical transceiver module including dual fiber type direct link	
		adapter for optically coupling optical subassemblies in the transceiver module	
13/644,113	USA	Filtered laser array assembly with external optical modulation and WDM	
		optical system including same	
13/595,505	USA	External cavity laser array system and WDM optical system including same	
13/357,142	USA	Optically matched laser array coupling assembly for coupling laser array to	
		arrayed waveguide grating	
13/357,130	USA	Laser array mux assembly with external reflector for providing a selected	
		wavelength or multiplexed wavelengths	
13/543,259	USA	Extended cavity fabry-perot laser assembly capable of high speed optical	
		modulation with narrow mode spacing and WDM optical system including	
		same	
13/357,116	USA	Wavelength-selectable laser device providing spatially-selectable	
		wavelength(s)	
12/245,008	USA	Reducing cross-modulation in multichannel modulated optical systems	
13/916,652	USA	Tunable laser with multiple in-line sections including sampled gratings	
14/551,353	USA	Tunable laser with multiple in-line sections including sampled gratings	
13/235,976	USA	Receptacle diplexer	
14/188,768	USA	Improved optical networking unit (ONU) packaging	
14/190,765	USA	Scribe etch process for semiconductor laser chip manufacturing	

14/456,124	USA	Monitoring a multiplexed laser array in an optical communication system	
14/672,802	USA	Improved coupling of photodetector array to optical demultiplexer outputs	
		with index matched material	
14/248,553	USA	Switched radio frequency (RF) driver for tunable laser with multiple in-line	
		sections	
14/312,930	USA	Laser transceiver with improved bit error rate	
14/528,011	USA	Transceiver with a thermal AWG DEMUX and thermal AWG MUX	
14/665,639	USA	Multichannel receiver optical subassembly with improved sensitivity	
14/501,751	USA	Semiconductor laser diode with integrated heating region	
14/452,715	USA	Test Fixture with thermoelectric cooler and spring-operated holding	
14/465,240	USA	Method and system for alignment of photodector array to optical	
		demultiplexer outputs	
14/295,459	USA	Monitoring and controlling temperature across a laser array in a transmitter	
		optical subassembly (TOSA) package	
14/661,772	USA	Tunable laser including parallel lasing cavities with a common	
14/590,456	USA	Two-section semiconductor laser with modulation-independent grating section	
		to reduce chirp	
14/588,608	USA	System, method and fixture for performing both optical power and wavelength	
		measurements of light emitted from a laser diode	
14/263,142	USA	Pluggable optical transceiver module	
14/482,550	USA	Optical coupling element and optical module having the same	
14/504,500	USA	Pluggable optical transceiver module	

TRADEMARK SECURITY AGREEMENT

This **TRADEMARK SECURITY AGREEMENT** (this "Agreement"), dated as of June 30, 2015 and, by and between, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "Borrower") and each Additional Grantor that may become a party hereto after the date hereof (each of the Borrower and each Additional Grantor being a "Grantor" and collectively the "Grantors") and EAST WEST BANK, as Agent for and representative of the Lender Parties (in such capacity, the "Agent"). If there are no Guarantors that are party to this Agreement, the term "Grantors" shall refer only to the Borrower.

WHEREAS, the Grantor has granted a security interest in and continuing Lien on the Collateral (including the trademarks subject of this Agreement) to the Agent pursuant to the Security Agreement, and desires to enter into this Agreement for purposes of affirming such grant to the Agent, and providing a short-form medium to facilitate the recording of such security interest in the applicable governmental offices.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Grantor agrees as follows:

Section 1. <u>Affirmation and Grant of Security</u>. The Grantor hereby affirms its grant to the Agent of, and grants to the Agent, a security interest in and continuing lien on all of such Grantor's right, title and interest in, to and under the trademarks listed on Schedule I to secure the Secured Obligations.

Section 2. <u>Recordation</u>. The Grantor authorizes and requests that the Commissioner of Patents and Trademarks and any other applicable government officer record this Agreement.

Section 3. <u>Grants, Rights and Remedies</u>. This Agreement has been entered into in conjunction with the provisions of the Security Agreement. The Grantor does hereby acknowledge and confirm that the grant of the security interest hereunder to, and the rights and remedies of, the Agent with respect to the Collateral are more fully set forth in the Security Agreement, the terms and provisions of which are incorporated herein by reference as if fully set forth herein.

Section 4. <u>Miscellaneous</u>. The Grantor agrees that neither this Agreement nor any term hereof may be changed, waived, discharged or terminated, except by an instrument in writing signed by the party (including, if applicable, any party required to evidence its consent to or acceptance of this Agreement) against whom enforcement of such change, waiver, discharge or termination is sought. This Agreement may be executed in any number of counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.

Section 5. <u>Governing Law</u>. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT GIVING EFFECT TO ITS CHOICE OF LAW PRINCIPLES, EXCEPT TO THE EXTENT THAT THE UCC PROVIDES THAT THE PERFECTION OF THE SECURITY INTEREST HEREUNDER, OR REMEDIES HEREUNDER, IN RESPECT OF ANY PARTICULAR COLLATERAL ARE GOVERNED BY THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF NEW YORK.

IN WITNESS WHEREOF, the Grantor has caused this Trademark Security Agreement to be duly executed and delivered by its officer thereunto duly authorized as of the date first above written.

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

SCHEDULE I REGISTERED TRADEMARKS

Registration No.	Country	Mark	Registration Date
2,535,730	USA	AOI	02-05-2002
2,832,440	USA	AOI (Stylized)	04-13-2004
3,001,557	USA	APPLIED OPTOELECTRONICS, INC.	09-27-2005

TRADEMARK APPLICATIONS

Appl. No. Country		Mark	
86/059,040	USA	AAOI	

APPLIED OPTOELECTRONICS, INC.

REVOLVING CREDIT NOTE

\$12,000,000

Sugar Land, TX June 30, 2015

FOR VALUE RECEIVED, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "<u>Borrower</u>"), promises to pay to the order of EAST WEST BANK (the "<u>Lender</u>") the principal amount of TWELVE MILLION DOLLARS (\$12,000,000) or, if less, the aggregate amount of Facility A Revolving Loans (as defined in the Credit Agreement referred to below) made by the Lender to the Borrower pursuant to the Credit Agreement referred to below outstanding on the Expiration Date (as defined in such Credit Agreement) or in such amounts on such earlier dates as may be provided in such Credit Agreement.

The Borrower also promises to pay interest on the unpaid principal amount hereof from the date hereof until paid at the rates and at the times determined in accordance with the provisions of the Credit Agreement.

All payments of principal and interest in respect of this Note shall be made in lawful money of the United States of America in same day funds at the office of the Lender described in the Credit Agreement.

This Note is a Revolving Note referred to in, and is entitled to the benefits of, the Credit Agreement, dated as of June 30, 2015 (as amended, modified or supplemented, the "<u>Credit Agreement</u>") between the Borrower, the Agent and the Lenders, which among other things provides for the acceleration of the maturity hereof upon the occurrence of certain events and for repayments in certain circumstances and upon certain terms and conditions. Terms defined in the Credit Agreement have the same meanings herein.

This Note is secured by the Security Documents.

The Borrower hereby expressly waives presentment, demand, notice, protest and all other demands and notices in connection with the delivery, acceptance, performance, default or enforcement of this Note and the Agreement, and an action for amounts due hereunder or thereunder shall immediately accrue.

The Lender and the Borrower intend to conform to all applicable laws limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower. Accordingly, notwithstanding any other provision hereof, the Borrower shall not be required to make any payment to or for the account of the Lender, and the Lender shall refund any payment made by the Borrower, to the extent that such requirement or such failure to refund would violate or conflict with mandatory and nonwaivable provisions of applicable law limiting the maximum amount of interest which may be charged or collected by the Lender from the Borrower. To the fullest extent permitted by law, in any action, suit or proceeding pertaining to this Note, the burden of proof, by clear and convincing evidence, shall be on the Borrower to demonstrate that this Paragraph applies to limit any obligation of the Borrower under this Note or to require the Lender to make any refund, or claiming that this Note conflicts with any applicable law limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower, as to each element of such claim.

This Note shall be governed by, and construed in accordance with, the laws of the State of New York without giving effect to its choice of law principles.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer, as of the date and the place first above written.

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

APPLIED OPTOELECTRONICS, INC.

REVOLVING CREDIT NOTE

\$3,000,000

FOR VALUE RECEIVED, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "<u>Borrower</u>"), promises to pay to the order of EAST WEST BANK (the "<u>Lender</u>") the principal amount of THREE MILLION DOLLARS (\$3,000,000) or, if less, the aggregate amount of Facility B Revolving Loans (as defined in the Credit Agreement referred to below) made by the Lender to the Borrower pursuant to the Credit Agreement referred to below outstanding on the Expiration Date (as defined in such Credit Agreement) or in such amounts on such earlier dates as may be provided in such Credit Agreement.

The Borrower also promises to pay interest on the unpaid principal amount hereof from the date hereof until paid at the rates and at the times determined in accordance with the provisions of the Credit Agreement.

All payments of principal and interest in respect of this Note shall be made in lawful money of the United States of America in same day funds at the office of the Lender described in the Credit Agreement.

This Note is a Revolving Note referred to in, and is entitled to the benefits of, the Credit Agreement, dated as of June 30, 2015 (as amended, modified or supplemented, the "<u>Credit Agreement</u>") between the Borrower, the Agent and the Lenders, which among other things provides for the acceleration of the maturity hereof upon the occurrence of certain events and for repayments in certain circumstances and upon certain terms and conditions. Terms defined in the Credit Agreement have the same meanings herein.

This Note is secured by the Security Documents.

The Borrower hereby expressly waives presentment, demand, notice, protest and all other demands and notices in connection with the delivery, acceptance, performance, default or enforcement of this Note and the Agreement, and an action for amounts due hereunder or thereunder shall immediately accrue.

The Lender and the Borrower intend to conform to all applicable laws limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower. Accordingly, notwithstanding any other provision hereof, the Borrower shall not be required to make any payment to or for the account of the Lender, and the Lender shall refund any payment made by the Borrower, to the extent that such requirement or such failure to refund would violate or conflict with mandatory and nonwaivable provisions of applicable law limiting the maximum amount of interest which may be charged or collected by the Lender from the Borrower. To the fullest extent permitted by law, in any action, suit or proceeding pertaining to this Note, the burden of proof, by clear and convincing evidence, shall be on the Borrower to demonstrate that this Paragraph applies to limit any obligation of the Borrower under this Note or to require the Lender to make any refund, or claiming that this Note conflicts with any applicable law limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower, as to each element of such claim.

This Note shall be governed by, and construed in accordance with, the laws of the State of New York without giving effect to its choice of law principles.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer, as of the date and the place first above written.

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

APPLIED OPTOELECTRONICS, INC.

TERM NOTE

\$6,000,000

Sugar Land, TX June 30, 2015

FOR VALUE RECEIVED, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "<u>Borrower</u>"), promises to pay to the order of EAST WEST BANK (the "<u>Lender</u>") the principal amount of SIX MILLION DOLLARS (\$6,000,000), or, if less, the aggregate amount of Term Loans (as defined in the Credit Agreement referred to below) made by the Lender to the Borrower pursuant to the Credit Agreement referred to below outstanding on the Term Loan Maturity Date (as defined in the Credit Agreement) or in such amounts on such earlier dates as are provided for in the Credit Agreement.

The Borrower also promises to pay interest on the unpaid principal amount hereof from the date hereof until paid at the rates and at the times determined in accordance with the provisions of the Credit Agreement.

All payments of principal and interest in respect of this Note shall be made in lawful money of the United States of America in same day funds at the office of the Lender described in the Credit Agreement.

This Note is a Term Note referred to in, and is entitled to the benefits of, the Credit Agreement, dated as of June 30, 2015 (as amended, modified or supplemented, the "<u>Credit Agreement</u>") between the Borrower, the Agent and the Lenders, which among other things provides for the acceleration of the maturity hereof upon the occurrence of certain events and for repayments in certain circumstances and upon certain terms and conditions. Terms defined in the Credit Agreement have the same meanings herein.

This Note is secured by the Security Documents.

The Borrower hereby expressly waives presentment, demand, notice, protest and all other demands and notices in connection with the delivery, acceptance, performance, default or enforcement of this Note and the Agreement, and an action for amounts due hereunder or thereunder shall immediately accrue.

The Lender and the Borrower intend to conform to all applicable laws limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower. Accordingly, notwithstanding any other provision hereof, the Borrower shall not be required to make any payment to or for the account of the Lender, and the Lender shall refund any payment made by the Borrower, to the extent that such requirement or such failure to refund would violate or conflict with mandatory and nonwaivable provisions of applicable law limiting the maximum amount of interest which may be charged or collected by the Lender from the Borrower. To the fullest extent permitted by law, in any action, suit or proceeding pertaining to this Note, the burden of proof, by clear and convincing evidence, shall be on the Borrower to demonstrate that this Paragraph applies to limit any obligation of the Borrower under this Note or to require the Lender to make any refund, or claiming that this Note conflicts with any applicable law limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower, as to each element of such claim.

This Note shall be governed by, and construed in accordance with, the laws of the State of New York without giving effect to its choice of law principles.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer, as of the date and the place first above written.

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

APPLIED OPTOELECTRONICS, INC.

REVOLVING CREDIT NOTE

\$8,000,000

Sugar Land, TX June 30, 2015

FOR VALUE RECEIVED, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "Borrower"), promises to pay to the order of COMERICA BANK (the "Lender") the principal amount of EIGHT DOLLARS (\$8,000,000) or, if less, the aggregate amount of Facility A Revolving Loans (as defined in the Credit Agreement referred to below) made by the Lender to the Borrower pursuant to the Credit Agreement referred to below outstanding on the Expiration Date (as defined in such Credit Agreement) or in such amounts on such earlier dates as may be provided in such Credit Agreement.

The Borrower also promises to pay interest on the unpaid principal amount hereof from the date hereof until paid at the rates and at the times determined in accordance with the provisions of the Credit Agreement.

All payments of principal and interest in respect of this Note shall be made in lawful money of the United States of America in same day funds at the office of the Lender described in the Credit Agreement.

This Note is a Revolving Note referred to in, and is entitled to the benefits of, the Credit Agreement, dated as of June 30, 2015 (as amended, modified or supplemented, the "<u>Credit Agreement</u>") between the Borrower, the Agent and the Lenders, which among other things provides for the acceleration of the maturity hereof upon the occurrence of certain events and for repayments in certain circumstances and upon certain terms and conditions. Terms defined in the Credit Agreement have the same meanings herein.

This Note is secured by the Security Documents.

The Borrower hereby expressly waives presentment, demand, notice, protest and all other demands and notices in connection with the delivery, acceptance, performance, default or enforcement of this Note and the Agreement, and an action for amounts due hereunder or thereunder shall immediately accrue.

The Lender and the Borrower intend to conform to all applicable laws limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower. Accordingly, notwithstanding any other provision hereof, the Borrower shall not be required to make any payment to or for the account of the Lender, and the Lender shall refund any payment made by the Borrower, to the extent that such requirement or such failure to refund would violate or conflict with mandatory and nonwaivable provisions of applicable law limiting the maximum amount of interest which may be charged or collected by the Lender from the Borrower. To the fullest extent permitted by law, in any action, suit or proceeding pertaining to this Note, the burden of proof, by clear and convincing evidence, shall be on the Borrower to demonstrate that this Paragraph applies to limit any obligation of the Borrower under this Note or to require the Lender to make any refund, or claiming that this Note conflicts with any applicable law limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower, as to each element of such claim.

This Note shall be governed by, and construed in accordance with, the laws of the State of New York without giving effect to its choice of law principles.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer, as of the date and the place first above written.

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

APPLIED OPTOELECTRONICS, INC.

REVOLVING CREDIT NOTE

\$2,000,000

FOR VALUE RECEIVED, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "<u>Borrower</u>"), promises to pay to the order of COMERICA BANK (the "<u>Lender</u>") the principal amount of TWO MILLION DOLLARS (\$2,000,000) or, if less, the aggregate amount of Facility B Revolving Loans (as defined in the Credit Agreement referred to below) made by the Lender to the Borrower pursuant to the Credit Agreement referred to below outstanding on the Expiration Date (as defined in such Credit Agreement) or in such amounts on such earlier dates as may be provided in such Credit Agreement.

The Borrower also promises to pay interest on the unpaid principal amount hereof from the date hereof until paid at the rates and at the times determined in accordance with the provisions of the Credit Agreement.

All payments of principal and interest in respect of this Note shall be made in lawful money of the United States of America in same day funds at the office of the Lender described in the Credit Agreement.

This Note is a Revolving Note referred to in, and is entitled to the benefits of, the Credit Agreement, dated as of June 30, 2015 (as amended, modified or supplemented, the "<u>Credit Agreement</u>") between the Borrower, the Agent and the Lenders, which among other things provides for the acceleration of the maturity hereof upon the occurrence of certain events and for repayments in certain circumstances and upon certain terms and conditions. Terms defined in the Credit Agreement have the same meanings herein.

This Note is secured by the Security Documents.

The Borrower hereby expressly waives presentment, demand, notice, protest and all other demands and notices in connection with the delivery, acceptance, performance, default or enforcement of this Note and the Agreement, and an action for amounts due hereunder or thereunder shall immediately accrue.

The Lender and the Borrower intend to conform to all applicable laws limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower. Accordingly, notwithstanding any other provision hereof, the Borrower shall not be required to make any payment to or for the account of the Lender, and the Lender shall refund any payment made by the Borrower, to the extent that such requirement or such failure to refund would violate or conflict with mandatory and nonwaivable provisions of applicable law limiting the maximum amount of interest which may be charged or collected by the Lender from the Borrower. To the fullest extent permitted by law, in any action, suit or proceeding pertaining to this Note, the burden of proof, by clear and convincing evidence, shall be on the Borrower to demonstrate that this Paragraph applies to limit any obligation of the Borrower under this Note or to require the Lender to make any refund, or claiming that this Note conflicts with any applicable law limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower, as to each element of such claim.

This Note shall be governed by, and construed in accordance with, the laws of the State of New York without giving effect to its choice of law principles.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer, as of the date and the place first above written.

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

APPLIED OPTOELECTRONICS, INC.

TERM NOTE

\$4,000,000

FOR VALUE RECEIVED, APPLIED OPTOELECTRONICS, INC., a Delaware corporation (the "<u>Borrower</u>"), promises to pay to the order of COMERICA BANK (the "<u>Lender</u>") the principal amount of FOUR MILLION DOLLARS (\$4,000,000), or, if less, the aggregate amount of Term Loans (as defined in the Credit Agreement referred to below) made by the Lender to the Borrower pursuant to the Credit Agreement referred to below outstanding on the Term Loan Maturity Date (as defined in the Credit Agreement) or in such amounts on such earlier dates as are provided for in the Credit Agreement.

The Borrower also promises to pay interest on the unpaid principal amount hereof from the date hereof until paid at the rates and at the times determined in accordance with the provisions of the Credit Agreement.

All payments of principal and interest in respect of this Note shall be made in lawful money of the United States of America in same day funds at the office of the Lender described in the Credit Agreement.

This Note is a Term Note referred to in, and is entitled to the benefits of, the Credit Agreement, dated as of June 30, 2015 (as amended, modified or supplemented, the "Credit Agreement") between the Borrower, the Agent and the Lenders, which among other things provides for the acceleration of the maturity hereof upon the occurrence of certain events and for repayments in certain circumstances and upon certain terms and conditions. Terms defined in the Credit Agreement have the same meanings herein.

This Note is secured by the Security Documents.

The Borrower hereby expressly waives presentment, demand, notice, protest and all other demands and notices in connection with the delivery, acceptance, performance, default or enforcement of this Note and the Agreement, and an action for amounts due hereunder or thereunder shall immediately accrue.

The Lender and the Borrower intend to conform to all applicable laws limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower. Accordingly, notwithstanding any other provision hereof, the Borrower shall not be required to make any payment to or for the account of the Lender, and the Lender shall refund any payment made by the Borrower, to the extent that such requirement or such failure to refund would violate or conflict with mandatory and nonwaivable provisions of applicable law limiting the maximum amount of interest which may be charged or collected by the Lender from the Borrower. To the fullest extent permitted by law, in any action, suit or proceeding pertaining to this Note, the burden of proof, by clear and convincing evidence, shall be on the Borrower to demonstrate that this Paragraph applies to limit any obligation of the Borrower under this Note or to require the Lender to make any refund, or claiming that this Note conflicts with any applicable law limiting the maximum rate of interest that may be charged or collected by the Lender from the Borrower, as to each element of such claim.

This Note shall be governed by, and construed in accordance with, the laws of the State of New York without giving effect to its choice of law principles.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer, as of the date and the place first above written.

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

SECOND LIEN CONSTRUCTION DEED OF TRUST, ASSIGNMENT OF LEASES AND RENTS, SECURITY AGREEMENT AND FIXTURE FILING

RECORDATION REQUESTED BY:

East West Bank Loan Servicing Department 9300 Flair Drive, 6th Floor El Monte, CA 91731

WHEN RECORDED MAIL TO:

East West Bank Loan Servicing Department 9300 Flair Drive, 6th Floor El Monte, CA 91731

SEND TAX NOTICES TO:

Applied Optoelectronics, Inc. 13115 Jess Pirtle Blvd. Sugar Land, TX 77478

SPACE ABOVE THIS LINE IS FOR RECORDER'S USE ONLY

THIS SECOND LIEN CONSTRUCTION DEED OF TRUST, ASSIGNMENT OF LEASES AND RENTS, SECURITY AGREEMENT AND FIXTURE FILING is dated June 30, 2015, among Applied Optoelectronics, Inc., a Delaware corporation ("Grantor"); East West Bank, whose address is Loan Servicing Department, 9300 Flair Drive, 6th Floor, El Monte, CA 91731, as Administrative Agent under the Credit Agreement (defined below) for the benefit of itself and the other lenders named therein (together with its successors and assigns, referred to below as "Beneficiary" or "Lender"); and Doug Krause, whose address is 135 North Los Robles Avenue, 7th Floor, Pasadena, CA 91101 (referred to below as "Trustee").

DEFINED TERMS. Certain capitalized terms used herein are defined at the end of this Deed of Trust.

CONVEYANCE AND GRANT. For valuable consideration, Grantor conveys to Trustee in trust, with power of sale, for the benefit of Lender as Beneficiary, the following described real property, together with all existing or subsequently erected or affixed buildings, Improvements and Fixtures; and all easements, rights of way, and appurtenances; all water and water rights; and all other rights, royalties, and profits relating to the real property, including without limitation such rights as Grantor may have in all minerals, oil, gas, geothermal and similar matters, (the "**Real Property**") located in Fort Bend County, State of Texas and described as follows:

(Continued)

See Exhibit "A", which is attached to this Deed of Trust and made a part of this Deed of Trust as if fully set forth herein.

The Real Property or its address is commonly known as 13115 Jess Pirtle Boulevard and vacant land, Sugar Land, TX 77478.

Grantor hereby absolutely assigns to Lender (also known as Beneficiary in this Deed of Trust) all of Grantor's right, title, and interest in and to all present and future leases of the Property and all Rents from the Property. In addition, Grantor grants to Lender a Uniform Commercial Code security interest in the Personal Property and Rents.

THIS DEED OF TRUST, INCLUDING THE ASSIGNMENT OF RENTS AND THE SECURITY INTEREST IN THE RENTS AND PERSONAL PROPERTY, IS GIVEN TO SECURE (A) PAYMENT OF THE INDEBTEDNESS AND (B) PERFORMANCE OF ANY AND ALL OBLIGATIONS UNDER THE NOTE, THE RELATED DOCUMENTS, AND THIS DEED OF TRUST. THIS DEED OF TRUST, INCLUDING THE ASSIGNMENT OF RENTS AND THE SECURITY INTEREST IN THE RENTS AND PERSONAL PROPERTY, IS ALSO GIVEN TO SECURE ANY AND ALL OF GRANTOR'S OBLIGATIONS UNDER THAT CERTAIN CONSTRUCTION LOAN AGREEMENT BETWEEN GRANTOR AND LENDER OF EVEN DATE HEREWITH ("LOAN AGREEMENT"). ANY EVENT OF DEFAULT UNDER THE CONSTRUCTION LOAN AGREEMENT, OR ANY OF THE RELATED DOCUMENTS REFERRED TO THEREIN, SHALL ALSO BE AN EVENT OF DEFAULT UNDER THIS DEED OF TRUST. THIS DEED OF TRUST. THIS DEED OF TRUST IS GIVEN AND ACCEPTED ON THE FOLLOWING TERMS:

PAYMENT AND PERFORMANCE. Except as otherwise provided in this Deed of Trust, Grantor shall pay to Lender all amounts secured by this Deed of Trust as they become due, and shall strictly and in a timely manner perform all of Grantor's obligations under the Note, this Deed of Trust, and the Related Documents.

PURPOSE OF LOAN. The Note represents, in part or in whole, cash or other financial accommodations advanced or committed by Lender to Grantor on June _____, 2015 at Grantor's request and which Grantor will use under its charter powers to discharge corporate debts. Grantor represents to Lender that its board of directors has authorized its legally elected, qualified, and acting officers to execute the Note and this Deed of Trust.

CONSTRUCTION MORTGAGE. This Deed of Trust is a "construction mortgage" for the purposes of Sections 9-334 and 2A-309 of the Uniform Commercial Code, as those sections have been adopted by the State of Texas.

POSSESSION AND MAINTENANCE OF THE PROPERTY. Grantor agrees that Grantor's possession and use of the Property shall be governed by the following provisions:

(Continued)

Possession and Use. Until the occurrence of an Event of Default, Grantor may (1) remain in possession and control of the Property; (2) use, operate or manage the Property; and (3) collect the Rents from the Property.

Duty to Maintain. Grantor shall maintain the Property in tenantable condition and promptly perform all repairs, replacements, and maintenance necessary to preserve its value.

Compliance With Environmental Laws. Grantor represents and warrants to Lender that: (1) during the period of Grantor's ownership of the Property, there has been no use, generation, manufacture, storage, treatment, disposal, release or threatened release of any Hazardous Substance by any person on, under, about or from the Property; (2) Grantor has no knowledge of, or reason to believe that there has been, except as previously disclosed to and acknowledged by Lender in writing, (a) any breach or violation of any Environmental Laws, (b) any use, generation, manufacture, storage, treatment, disposal, release or threatened release of any Hazardous Substance on, under, about or from the Property by any prior owners or occupants of the Property, or (c) any actual or threatened litigation or claims of any kind by any person relating to such matters; and (3) except as previously disclosed to and acknowledged by Lender in writing, (a) neither Grantor nor any tenant, contractor, agent or other authorized user of the Property shall use, generate, manufacture, store, treat, dispose of or release any Hazardous Substance on, under, about or from the Property; and (b) any such activity shall be conducted in compliance with all applicable federal, state, and local laws, regulations and ordinances, including without limitation all Environmental Laws. Grantor authorizes Lender and its agents to enter upon the Property to make such inspections and tests, at Grantor's expense, as Lender may deem appropriate to determine compliance of the Property with this section of this Deed of Trust. Any inspections or tests made by Lender shall be for Lender's purposes only and shall not be construed to create any responsibility or liability on the part of Lender to Grantor or to any other person. The representations and warranties contained herein are based on Grantor's due diligence in investigating the Property for Hazardous Substances. Grantor hereby (1) releases and waives any future claims against Lender for indemnity or contribution in the event Grantor becomes liable for cleanup or other costs under any such laws; and (2) agrees to indemnify, defend, and hold harmless Lender against any and all claims, losses, liabilities, damages, penalties, and expenses which Lender may directly or indirectly sustain or suffer resulting from a breach of this section of this Deed of Trust or as a consequence of any use, generation, manufacture, storage, disposal, release or threatened release occurring prior to Grantor's ownership or interest in the Property, whether or not the same was or should have been known to Grantor. The provisions of this section of this Deed of Trust, including the obligation to indemnify and defend, shall survive the payment of the Indebtedness and the satisfaction and reconveyance of the lien of this Deed of Trust and shall not be affected by Lender's acquisition of any interest in the Property, whether by foreclosure or otherwise.

(Continued)

Nuisance, Waste. Grantor shall not cause, conduct or permit any nuisance nor commit, permit, or suffer any stripping of or waste on or to the Property or any portion of the Property. Without limiting the generality of the foregoing, Grantor will not remove, or grant to any other party the right to remove, any timber, minerals (including oil and gas), coal, clay, scoria, soil, gravel or rock products without Lender's prior written consent. This restriction will not apply to rights and easements (such as gas and oil) not owned by Grantor and of which Grantor has informed Lender in writing prior to Grantor's signing of this Deed of Trust or are listed in the lender's policy of title insurance provided by Borrower to Lender in connection with the Prior Deed of Trust (as hereinafter defined).

Removal of Improvements. Grantor shall not demolish or remove any Improvements from the Real Property without Lender's prior written consent. As a condition to the removal of any Improvements, Lender may require Grantor to make arrangements satisfactory to Lender to replace such Improvements with Improvements of at least equal value.

Lender's Right to Enter. Lender and Lender's agents and representatives may enter upon the Real Property at all reasonable times to attend to Lender's interests and to inspect the Real Property for purposes of Grantor's compliance with the terms and conditions of this Deed of Trust.

Compliance with Governmental Requirements. Grantor shall promptly comply with all laws, ordinances, and regulations, now or hereafter in effect, of all governmental authorities applicable to the use or occupancy of the Property, including without limitation, the Americans With Disabilities Act. Grantor may contest in good faith any such law, ordinance, or regulation and withhold compliance during any proceeding, including appropriate appeals, so long as Grantor has notified Lender in writing prior to doing so and so long as, in Lender's sole opinion, Lender's interests in the Property are not jeopardized. Lender may require Grantor to post adequate security or a surety bond, reasonably satisfactory to Lender, to protect Lender's interest.

Duty to Protect. Grantor agrees neither to abandon nor leave unattended the Property. Grantor shall do all other acts, in addition to those acts set forth above in this section, which from the character and use of the Property are reasonably necessary to protect and preserve the Property.

Construction Loan. If some or all of the proceeds of the loan creating the Indebtedness are to be used to construct or complete construction of any improvements on the Property, the Improvements shall be completed no later than the maturity date of the Note (or such earlier date as Lender may reasonably establish) and Grantor shall pay in full all costs and expenses in connection with the work. Lender will disburse loan proceeds under such terms and conditions as Lender may deem reasonably necessary to insure that the interest created by this Deed of Trust shall have priority over all possible liens, including those of material suppliers and workmen. Lender may require, among other things, that disbursement requests be supported by receipted bills, expense affidavits, waivers of liens, construction progress reports, and such other documentation as Lender may request.

(Continued)

DUE ON SALE – CONSENT BY LENDER. Lender may, at Lender's option, declare immediately due and payable all sums secured by this Deed of Trust upon the sale or transfer, without Lender's prior written consent, of all or any part of the Real Property, or any interest in the Real Property. A "sale or transfer" means the conveyance of Real Property or any right, title or interest in the Real Property, whether legal, beneficial or equitable; whether voluntary or involuntary; whether by outright sale, deed, installment sale contract, land contract, contract for deed, leasehold interest with a term greater than three (3) years, lease-option contract, or by sale, assignment, or transfer of any beneficial interest in or to any land trust holding title to the Real Property, or by any other method of conveyance of an interest in the Real Property. If any Grantor is a corporation, partnership or limited liability company, transfer also includes any change in ownership of more than twenty-five percent (25%) of the voting stock, partnership interests or limited liability company interest, as the case may be, of such Grantor. However, this option shall not be exercised by Lender if such exercise is prohibited by federal law or by Texas law.

TAXES AND LIENS. The following provisions relating to the taxes and liens on the Property are part of this Deed of Trust:

Payment. Grantor shall pay when due (and in all events prior to delinquency) all taxes, special taxes, assessments, charges (including water and sewer), fines and impositions levied against or on account of the Property, and shall pay when due all claims for work done on or for services rendered or material furnished to the Property. Grantor shall maintain the Property free of all liens having priority over or equal to the interest of Lender under this Deed of Trust, except for the lien of taxes and assessments not due and except as otherwise provided in this Deed of Trust.

Right to Consent. Grantor may withhold payment of any tax, assessment, or claim in connection with a good faith dispute over the obligation to pay, so long as Lender's interest in the Property is not jeopardized. If a lien arises or is filed as a result of nonpayment, Grantor shall within fifteen (15) days after the lien arises, or, if a lien is filed, within fifteen (15) days after Grantor has notice of the filing, secure the discharge of the lien, or if requested by Lender, deposit with Lender cash or a sufficient corporate surety bond or other security satisfactory to Lender in an amount sufficient to discharge the lien plus any costs and Lender's reasonable attorney's fees, or other charges that could accrue as a result of a foreclosure or sale under the lien. In any contest, Grantor shall defend itself and Lender and shall satisfy any adverse judgment before enforcement against the Property. Grantor shall name Lender as an additional obligee under any surety bond furnished in the contest proceedings.

(Continued)

Evidence of Payment. Grantor shall upon demand furnish to Lender satisfactory evidence of payment of the taxes or assessments and shall authorize the appropriate governmental official to deliver to Lender at any time a written statement of the taxes and assessments against the Property.

Notice of Construction. Grantor shall notify Lender at least fifteen (15) days before any work is commenced, any services are furnished, or any materials are supplied to the Property, if any mechanic's lien, materialmen's lien, or other lien could be asserted on account of the work, services, or materials and the cost exceeds \$10,000.00. Grantor will upon request of Lender furnish to Lender advance assurances satisfactory to Lender that Grantor can and will pay the cost of such Improvements.

PROPERTY DAMAGE INSURANCE. The following provisions relating to insuring the Property are a part of this Deed of Trust.

Maintenance of Insurance. Grantor shall procure and maintain policies of fire insurance with standard extended coverage endorsements on a replacement basis for the full insurable value covering all Improvements on the Real Property in an amount sufficient to avoid application of any coinsurance clause, and with a standard mortgagee clause in favor of Lender. Grantor shall also procure and maintain comprehensive general liability insurance in such coverage amounts as Lender may request with Trustee and Lender being named as additional insureds in such liability insurance policies. Additionally, Grantor shall maintain such other insurance, including but not limited to hazard, business interruption, and boiler insurance, as Lender may reasonably require. Policies shall be written in form, amounts, coverages and basis reasonably acceptable to Lender, with losses made payable to Lender. GRANTOR MAY FURNISH THE REQUIRED INSURANCE WHETHER THROUGH EXISTING POLICIES OWNED OR CONTROLLED BY GRANTOR OR THROUGH EQUIVALENT INSURANCE FROM ANY INSURANCE COMPANY AUTHORIZED TO TRANSACT BUSINESS IN THE STATE OF TEXAS. If Grantor fails to provide any required insurance or fails to continue such insurance in force, Lender may, but shall not be required to, do so at Grantor's expense, and the cost of the insurance will be added to the Indebtedness. If any such insurance is procured by Lender, Grantor will be so notified, and Grantor will have the option of furnishing equivalent insurance through any insurer authorized to transact business in Texas. Grantor, upon request of Lender, will deliver to Lender from time to time the policies or certificates of insurance in form satisfactory to Lender, including stipulations that coverages will not be cancelled or diminished without at least thirty (30) days prior written notice to Lender. Each insurance policy also shall include an endorsement providing that coverage in favor of Lender will not be impaired in any way by an act, omission or default of Grantor or any other person. Should the Real Property be located in an area designated by the Administrator of the Federal Emergency Management Agency as a special flood hazard area, Grantor agrees to obtain and maintain Federal Flood Insurance, if available, within 45 days after notice is given by Lender that the Property is located in a special flood hazard area, for the full unpaid principal balance of the loan and any prior liens on the property securing the loan, up to the maximum policy limits set under the National Flood Insurance Program, or as otherwise required by Lender, and to maintain such insurance for the term of the loan.

(Continued)

Application of Proceeds. Grantor shall promptly notify Lender of any loss or damage to the Property if the estimated cost of repair or replacement exceeds \$10,000.00, Lender may make proof of loss if Grantor fails to do so within fifteen (15) days of the casualty. Whether or not Lender's security is impaired, Lender may, at Lender's election, receive and retain the proceeds of any insurance and apply the proceeds to the reduction of the Indebtedness, payment of any lien affecting the Property, or the restoration and repair of the Property. If Lender elects to apply the proceeds to restoration and repair, Grantor shall repair or replace the damaged or destroyed improvements in a manner satisfactory to Lender. Lender shall, upon satisfactory proof of such expenditure, pay or reimburse Grantor from the proceeds for the reasonable cost of repair or restoration if Grantor is not in default under this Deed of Trust. Any proceeds which have not been disbursed within 180 days after their receipt and which Lender has not committed to the repair or restoration of the Property shall be used first to pay any amount owing to Lender under this Deed of Trust, then to pay accrued interest, and the remainder, if any, shall be applied to the principal balance of the Indebtedness. If Lender holds any proceeds after payment in full of the Indebtedness, such proceeds shall be paid to Grantor as Grantor's interests may appear.

Grantor's Report on Insurance. Upon request of Lender, however not more than once a year, Grantor shall furnish to Lender a report on each existing policy of insurance showing: (1) the name of the insurer; (2) the risks insured; (3) the amount of the policy; (4) the property insured, the then current replacement value of such property, and the manner of determining that value; and (5) the expiration date of the policy. Grantor shall, upon request of Lender, have an independent appraiser satisfactory to Lender determine the cash value replacement cost of the Property.

The Prior Deed of Trust. For so long as the Lender hereunder is also the lender named in the Prior Deed of Trust, Grantor's actions and deliveries under the Prior Deed of Trust shall be deemed to also satisfy the requirements of this section titled "Property Damage Insurance" so long as such actions and deliveries satisfy the requirements of the Prior Deed of Trust.

(Continued)

LENDER'S EXPENDITURES. If any action or proceeding is commenced that would materially affect Lender's interest in the Property or if Grantor fails to comply with any provision of this Deed of Trust or any Related Documents, including but not limited to Grantor's failure to discharge or pay when due any amounts Grantor is required to discharge or pay under this Deed of Trust or any Related Documents. Lender on Grantor's behalf may (but shall not be obligated to) take any action that Lender deems appropriate, including but not limited to discharging or paying all taxes, liens, security interests, encumbrances and other claims, at any time levied or placed on the Property and paying all costs for insuring, maintaining and preserving the Property. All such expenditures paid by Lender for such purposes will then bear interest at the Note rate from the date paid by Lender to the date of repayment by Grantor. To the extent permitted by applicable law, all such expenses will become a part of the Indebtedness and, at Lender's option, will (A) be payable on demand; (B) be added to the balance of the Note and be apportioned among and be payable with any installment payments to become due during either (1) the term of any applicable insurance policy; or (2) the remaining term of the Note; or (C) be treated as a balloon payment which will be due and payable at the Note's maturity. The Deed of Trust also will secure payment of these amounts. Such right shall be in addition to all other rights and remedies to which Lender may be entitled upon Default.

WARRANTY; DEFENSE OF TITLE. The following provisions relating to ownership of the Property are a part of this Deed of Trust:

Title. Grantor warrants that: (a) Grantor holds good and marketable title of record to the Property in fee simple, free and clear of all liens and encumbrances other than those set forth in the Real Property description or in any title insurance policy, title report, or final title opinion issued in favor of, and accepted by, Lender in connection with this Deed of Trust, including without limitation the Prior Deed of Trust, and (b) Grantor has the full right, power, and authority to execute and deliver this Deed of Trust to Lender.

Defense of Title. Subject to the exception in the paragraph above, Grantor warrants and will forever defend the title to the Property against the lawful claims of all persons. In the event any action or proceeding is commenced that questions Grantor's title or the interest of Trustee or Lender under this Deed of Trust, Grantor shall defend the action at Grantor's expense. Grantor may be the nominal party in such proceeding, but Lender shall be entitled to participate in the proceeding and to be represented in the proceeding by counsel of Lender's own choice, and Grantor will deliver, or cause to be delivered, to Lender such instruments as Lender may request from time to time to permit such participation.

Compliance With Laws. Grantor warrants that the Property and Grantor's use of the Property complies with all existing applicable laws, ordinances, and regulations of governmental authorities.

(Continued)

Survival of Representations and Warranties. All representations, warranties, and agreements made by Grantor in this Deed of Trust shall survive the execution and delivery of this Deed of Trust, shall be continuing in nature, and shall remain in full force and effect until such time as Grantor's Indebtedness shall be paid in full.

CONDEMNATION, JUDGMENTS AND AWARDS. The following provisions relating to condemnation proceedings, judgments, decrees and awards for injury to the Property are a part of this Deed of Trust:

Proceedings. If any proceeding in condemnation is filed, Grantor shall promptly notify Lender in writing, and Grantor shall promptly take such steps as may be necessary to defend the action and obtain the award. Grantor may be the nominal party in such proceeding, but Lender shall be entitled to participate in the proceeding and to be represented in the proceeding by counsel of its own choice, and Grantor will deliver or cause to be delivered to Lender such instruments and documentation as may be requested by Lender from time to time to permit such participation.

Application of Net Proceeds. To the extent permitted by applicable law, all judgments, decrees and awards for injury or damage to the Property, or any part of the Property, and awards pursuant to proceedings for condemnation of the Property, are hereby absolutely assigned to Lender, and if all or any part of the Property is condemned by eminent domain proceedings or by any proceeding or purchase in lieu of condemnation, Lender may at its election require that all or any portion of the net proceeds of the award be applied to the Indebtedness or the repair or restoration of the Property. The net proceeds of the award, judgment or decree shall mean the award after payment of all reasonable costs, expenses, and attorneys' fees incurred by Trustee or Lender in connection with the condemnation.

SECURITY AGREEMENT; FINANCING STATEMENTS. The following provisions relating to this Deed of Trust as a security agreement are a part of this Deed of Trust:

Security Agreement. This instrument shall constitute a Security Agreement to the extent any of the Property constitutes Fixtures, and Lender shall have all of the rights of a secured party under the Uniform Commercial Code as amended from time to time.

Security Interest. Upon request by Lender, Grantor shall take whatever action is requested by Lender to perfect and continue Lender's security interest in the Rents and Personal Property. In addition to recording this Deed of Trust in the real property records, Lender may, at any time and without further authorization from Grantor, file executed counterparts, copies or reproductions of this Deed of Trust as a financing statement. Grantor shall reimburse Lender for all expenses incurred in perfecting or continuing this security interest. Upon default, Grantor shall not remove, sever or detach the Personal Property from the Property. Upon default, Grantor shall assemble any Personal Property not affixed to the Property in a manner and at a place reasonably convenient to Grantor and Lender and make it available to Lender within three (3) days after receipt of written demand from Lender to the extent permitted by applicable law.

(Continued)

Address. The mailing addresses of Grantor (debtor) and Lender (secured party) from which information concerning the security interest granted by this Deed of Trust may be obtained (each as required by the Uniform Commercial Code) are as stated on the first page of this Deed of Trust.

FURTHER ASSURANCES; ATTORNEY-IN-FACT. The following provisions relating to further assurances and attorney-in-fact are a part of this Deed of Trust:

Further Assurance. At any time, and from time to time, upon request of Lender, Grantor will make, execute and deliver, or will cause to be made, executed or delivered, to Lender or to Lender's designee, and when requested by Lender, cause to be filed, recorded, refiled, or rerecorded, as the case may be, at such times and in such offices and places as Lender may deem appropriate, any and all such mortgages, deeds of trust, security deeds, security agreements, financing statements, continuation statements, instruments of further assurance, certificates, and other documents as may, in the sole opinion of Lender, be necessary or desirable in order to effectuate, complete, perfect, continue, or preserve (1) Grantor's obligations under the Note, this Deed of Trust, and the Related Documents, and (2) the liens and security interests created by this Deed of Trust as first and prior liens on the Property, whether now owned or hereafter acquired by Grantor. Unless prohibited by law or Lender agrees to the contrary in writing, Grantor shall reimburse Lender for all costs and expenses incurred in connection with the matters referred to in this paragraph.

Attorney-in-Fact. If Grantor fails to do any of the things referred to in the preceding paragraph, Lender may do so for and in the name of Grantor and at Grantor's expense. For such purposes, Grantor hereby irrevocably appoints Lender as Grantor's attorney-in-fact for the purpose of making, executing, delivering, filing, recording, and doing all other things as may be necessary or desirable, in Lender's sole opinion, to accomplish the matters referred to in the preceding paragraph.

FULL PERFORMANCE. If Grantor pays all the Indebtedness when due, and otherwise performs all the obligations imposed upon Grantor under this Deed of Trust, Lender shall execute and deliver to Grantor a release of this Deed of Trust lien and suitable statements of termination of any financing statement on file evidencing Lender's security interest in the Rents and the Personal Property, However, it is agreed that the payment of all the Indebtedness and performance of such obligations shall not terminate this Deed of Trust unless the liens and interests created hereby are released by Lender by a proper recordable instrument. Any filing fees required by law shall be paid by Grantor, if permitted by applicable law.

Loan No: _____

(Continued)

ASSIGNMENT OF RENTS AND OTHER CONTRACTS. The following provisions relating to this Deed of Trust as an assignment of rents and other contracts are a part of this Deed of Trust:

Assignment of Rents. As additional security for the payment of the Indebtedness, Grantor hereby collaterally and unconditionally assigns to Lender all Rents as defined in the Definitions section of this Deed of Trust and all proceeds and other amounts paid or owing to Grantor under or pursuant to any and all contracts and bonds relating to the construction, erection or renovation of the Property.

Assignment of Lease Guaranties and Other Property. In order to secure payment and performance of the Indebtedness, Grantor hereby assigns to Lender all existing and future lease guaranties, all claims and rights to the payment of damages and other claims arising from any rejection by a lessee of any Lease under the United States Bankruptcy Code (a "<u>Bankruptcy Claim</u>"), all proceeds from the sale or other disposition of the Leases, the Rents, the Lease Guaranties, the Bankruptcy Claims, and all other rights and benefits of Grantor as a lessor under the Leases and under the lease guaranties, plus all amendments, replacements, and renewals of the foregoing. Prior to an Event of Default, Grantor shall have the right, without joinder of Lender, to enforce the Leases, unless Lender directs otherwise.

Warranties Concerning Leases and Rents. Grantor represents and warrants that:

(a) Grantor has good title to the Leases and Rents and authority to assign and grant a security interest in the property described above, and no other person or entity has any right, title or interest therein;

(b) all existing Leases are valid, unmodified and in full force and effect, except as indicated herein or in a separate document provided to Lender by Grantor, and to Grantor's knowledge, no default exists thereunder;

(c) unless otherwise provided herein, no Rents have been or will be assigned, mortgaged or pledged; and

(d) no Rents have been or will be anticipated, waived, released, discounted, set off or compromised except for prudent business reasons.

Grantor's Covenants of Performance. Grantor covenants to:

(a) reserved;(b) reserved;

(Continued)

Loan No: _____

(c) defend, at Grantor's expense, any proceeding pertaining to the Leases, including, if Lender so requests, any such proceeding to which Lender is a party;

(d) neither create nor permit any encumbrance upon its interest as lessor of the Leases, except this Deed of Trust and any other encumbrances permitted by this Deed of Trust;

(e) if Grantor collects or receives Rents that Lender is entitled to collect and receive under this Section of this Deed of Trust or applicable law, Grantor shall deliver such Rents to Lender immediately upon receipt or upon demand by Lender; and

(f) assign and add Lender as an additional beneficiary to all letters of credit given by tenants of the Property to Grantor as security for a

Lease.

Prior Approval for Actions Affecting Leases. Grantor shall not, without the prior written consent of Lender:

- (a) receive or collect Rents more than one (1) month in advance;
- (b) encumber or assign future Rents;
- (c) reserved;
- (d) reserved; or

(e) permit any assignment of the Leases or of the other property pledged to Lender under this section of this Deed of Trust.

Lender in Possession. Lender's acceptance of this assignment shall not, prior to entry upon and taking possession of the Property by Lender, be deemed to constitute Lender a "mortgagee in possession," nor obligate Lender to appear in or defend any proceedings relating to any of the Leases or to the Property, take any action hereunder, expend any money, incur any expenses, or perform any obligation or liability under the Leases, or assume any obligation for any deposits delivered to Grantor by any tenant and not delivered to Lender. Lender shall not be liable for any injury or damage to any person or property in or about the Mortgaged Property.

Indemnification. GRANTOR HEREBY INDEMNIFIES AND HOLDS LENDER (WHICH SHALL INCLUDE THE DIRECTORS, OFFICERS, PARTNERS, EMPLOYEES, REPRESENTATIVES, ATTORNEYS AND AGENTS OF LENDER AND ANY PERSONS OR ENTITIES OWNED OR CONTROLLED BY, OWNING OR CONTROLLING, OR UNDER COMMON CONTROL OR AFFILIATED WITH LENDER) HARMLESS FROM ALL LIABILITY, DAMAGE OR EXPENSE IMPOSED ON OR INCURRED BY LENDER FROM ANY CLAIMS UNDER THE LEASES. IN ADDITION, GRANTOR SHALL HAVE NO RIGHT OR CLAIM AGAINST ANY LESSEE FOR THE PAYMENT OF ANY RENT TO LENDER PURSUANT TO PROVISIONS IN THIS SECTION OF THIS DEED OF TRUST. The foregoing indemnities shall not terminate upon the foreclosure, release or other termination of this Deed of Trust but will survive foreclosure of this Deed of Trust or conveyance in lieu of foreclosure and the repayment of the Indebtedness and the discharge and release of this Deed of Trust and the other Loan Documents.

(Continued)

Merger. There shall be no merger of the leasehold estate, created by the Leases, with the fee estate of the Land without prior written consent of Lender.

Right to Rely. Grantor hereby irrevocably authorizes and directs the tenants under the Leases to pay Rents to Lender upon written demand by Lender after the occurrence of any Event of Default, without further consent of Grantor, and the tenants may rely upon any written statement delivered by Lender to the tenants. Any such payment to Lender shall constitute payment to Grantor under the Leases. The assignment of Rents set forth in section of this Deed of Trust is not contingent upon any notice or demand by Lender to the tenants.

Termination of Assignment. Upon payment in full of the Indebtedness and the delivery and recording of a release of the Deed of Trust lien created by this Deed of Trust duly executed by Lender, the assignments and security interest described above in this Article Four shall become null and void and shall be of no further force and effect.

EVENTS OF DEFAULT. An Event of Default under the Credit Agreement shall constitute an Event of Default under this Deed of Trust.

RIGHTS AND REMEDIES ON EVENT OF DEFAULT. If an Event of Default occurs under this Deed of Trust, at any time thereafter, Trustee or Lender may exercise any one or more of the following rights and remedies:

Election of Remedies. Election by Lender to pursue any remedy shall not exclude pursuit of any other remedy, and an election to make expenditures or to take action to perform an obligation of Grantor under this Deed of Trust, after Grantor's failure to perform, shall not affect Lender's right to declare a default and exercise its remedies.

Accelerate Indebtedness. Lender may declare the unpaid principal balance of the Indebtedness due and payable. In no event will Grantor be required to pay any unearned interest.

Foreclosure. If Lender invokes the power of sale, Trustee, at the request of Lender, may sell all or any portion of the Property at public auction to the highest bidder for cash at the location within the courthouse designated by the County Commissioners Court, or if no such area has been designated, at the area designated in the notice of sale within the courthouse, between the hours of 10:00 A.M. and 4:00 P.M. on the first Tuesday of any month, after the Trustee or its agent has given notice of the time and place of sale and of the property to be sold as required by the Texas Property Code, as then amended.

Loan No: _____

Loan No: _____

DEED OF TRUST

(Continued)

UCC Remedies. With respect to all or any part of the Personal Properly, Lender shall have all the rights and remedies of a secured party under the Uniform Commercial Code.

Collect Rents. Lender shall have the right at any time, and even though no Event of Default shall have occurred under this Deed of Trust, to and may immediately collect and receive the Rents; provided, however, that Lender shall not be deemed to have taken possession of the Property except on the exercise of its option to do so, evidenced by its demand and overt act for such purpose. It shall not be necessary for Lender to institute any type of legal proceedings or take any other action whatsoever to enforce the assignment provisions in this section of this Deed of Trust. Rents so received by Lender for any period prior to foreclosure or acceptance of a deed in lieu of such foreclosure shall be applied by Lender to the Indebtedness as set forth in the Note. Neither this assignment nor the receipt of Rents by Lender shall effect a pro tanto payment of the debt evidenced by, or arising under the Indebtedness. Lender shall provide any notice required by applicable law with regard to such enforcement of its right to collect and receive the Rents. In addition, if the Property is vacant, Lender may rent or lease the Property. Lender shall not be liable for its failure to rent the Property, to collect any Rents, or to exercise diligence in any matter relating to the Rents; Lender shall be accountable only for Rents actually received. Lender neither has nor assumes any obligation as lessor or landlord with respect to any occupant of the Property. Rents so received shall be paid to the person or persons legally entitled to the residue.

Trustee's Powers. Grantor hereby jointly and severally authorizes and empowers Trustee to sell all or any portion of the Property together or in lots or parcels, as Trustee may deem expedient, and to execute and deliver to the purchaser or purchasers of such Property good and sufficient deeds of conveyance of fee simple title, or of lessor estates, and bills of sale and assignments, with covenants of general warranty made on Grantor's behalf. In no event shall Trustee be required to exhibit, present or display at any such sale any of the Property to be sold at such sale. The Trustee making such sale shall receive the proceeds of the sale and shall apply the same as provided below. Payment of the purchase price to Trustee shall satisfy the liability of the purchaser at any such sale of the Property, and such person shall not be bound to look after the application of the proceeds.

Appoint Receiver. Lender, in accordance with applicable laws, shall have the right to have a receiver appointed to take possession of all or any part of the Property, with the power to protect and preserve the Property, to operate the Property preceding foreclosure or sale, and to collect the Rents from the Property and apply the proceeds, over and above the cost of the receivership, against the Indebtedness. The receiver may serve without bond if permitted by law. Lender's right to the appointment of a receiver shall exist whether or not the apparent value of the Property exceeds the Indebtedness by a substantial amount. Employment by Lender shall not disqualify a person from serving as a receiver.

(Continued)

Tenancy at Sufferance. If Grantor remains in possession of the Properly after the Property is sold as provided above or Lender otherwise becomes entitled to possession of the Property upon default of Grantor, Grantor shall become a tenant at sufferance of Lender or the purchaser of the Property and shall, at Lender's option, either (1) pay a reasonable rental for the use of the Property, (2) vacate the Properly immediately upon the demand of Lender, or (3) if such tenants refuse to surrender possession of the Property upon demand, the purchaser shall be entitled to institute and maintain the statutory action of forcible entry and detainer and procure a writ of possession thereunder, and Grantor expressly waives all damages sustained by reason thereof.

Other Remedies. Trustee or Lender shall have any other right or remedy provided in this Deed of Trust or the Note or available at law or in equity.

Sale of the Property. To the extent permitted by applicable law, Grantor hereby waives any and all rights to have the Property marshalled. In exercising its rights and remedies, the Trustee or Lender shall be free to sell all or any part of the Property together or separately, in one sale or by separate sales. Lender shall be entitled to bid at any public sale on all or any portion of the Property. Trustee may convey all or any part of the Property to the highest bidder for cash with a general warranty binding Grantor, subject to prior liens and to other exceptions to conveyance and warranty. Grantor waives all requirements of appraisement, if any. The affidavit of any person having knowledge of the facts to the effect that proper notice as required by the Texas Property Code was given shall be prima facie evidence of the fact that such notice was in fact given. Recitals and statements of fact in any notice or in any conveyance to the purchaser or purchasers of the Property in any foreclosure sale under this Deed of Trust shall be prima facie evidence of the truth of such facts, and all prerequisites and requirements necessary to the validity of any such sale shall be presumed to have been performed. Any sale under the powers granted by this Deed of Trust shall be a perpetual bar against Grantor, Grantor's heirs, successors, assigns and legal representatives.

Proceeds. Trustee shall pay the proceeds of any sale of the Property (a) first, to the expenses of foreclosure, including reasonable fees or charges paid to the Trustee, including but not limited to fees for enforcing the lien, posting for sale, selling, or releasing the Property, (b) then to Lender the full amount of the Indebtedness, (c) then to any amount required by law to be paid before payment to Grantor, and (d) the balance, if any, to Grantor.

Loan No: _

DEED OF TRUST

(Continued)

Attorneys' Fees; Expenses. If Lender institutes any suit or action to enforce any of the terms of this Deed of Trust, Lender shall be entitled to recover such sum as the court may adjudge reasonable as Lender's attorneys' fees at trial and upon any appeal. Whether or not any court action is involved, and to the extent not prohibited by law, all reasonable expenses Lender incurs that in Lender's opinion are necessary at any time for the protection of its interest or the enforcement of its rights shall become a part of the Indebtedness payable on demand and shall bear interest at the Note rate from the date of the expenditure until repaid. Expenses covered by this paragraph include, without limitation, however subject to any limits under applicable law, Lender's reasonable attorneys' fees and Lender's legal expenses, whether or not there is a lawsuit, including Lender's reasonable attorneys' fees and expenses for bankruptcy proceedings (including efforts to modify or vacate any automatic stay or injunction), appeals, and any anticipated post-judgment collection services, the cost of searching records, obtaining title reports (including foreclosure reports), surveyors' reports, and appraisal fees, title insurance, and fees for the Trustee, to the extent permitted by applicable law. Grantor also will pay any court costs, in addition to all other sums provided by law. In the event of foreclosure of this Deed of Trust, Lender shall be entitled to recover from Grantor Lender's reasonable attorneys' fees and actual disbursements that Lender necessarily incurs in pursuing such foreclosure.

Waiver of Deficiency Statute. (a) In the event an interest in any of the Property is foreclosed upon pursuant to a judicial or nonjudicial foreclosure sale, Grantor agrees as follows: notwithstanding the provisions of Sections 51.003, 51.004, and 51.005 of the Texas Property Code (as the same may be amended from time to time), and to the extent permitted by law, Grantor agrees that Lender shall be entitled to seek a deficiency judgment from Grantor and/or any other party obligated on the Indebtedness equal to the difference between the amount owing on the Indebtedness and the amount for which the Property was sold pursuant to judicial or nonjudicial foreclosure sale. Grantor expressly recognizes that this Section 6.5(a) constitutes a waiver of the above-cited provisions of the Texas Property Code which would otherwise permit Grantor and other persons against whom recovery of deficiencies is sought or Guarantor independently (even absent the initiation of deficiency proceedings against them) to present competent evidence of the fair market value of the Property as of the date of the foreclosure sale and offset against any deficiency the amount by which the foreclosure sale price is determined to be less than such fair market value. Grantor further recognizes and agrees that this waiver creates an irrebuttable presumption that the foreclosure sale price is equal to the fair market value of the Property for purposes of calculating deficiencies owed by Grantor and/or others against whom recovery of a deficiency is sought. (b) Alternatively, in the event the waiver provided for in <u>Subsection (a)</u> above is determined by a court of competent jurisdiction to be unenforceable, the following shall be the basis for the finder of fact's determination of the fair market value of the Property as of the date of the foreclosure sale in proceedings governed by Sections 51.003, 51.004 and 51.005 of the Texas Property Code (as amended from time to time): (i) the Property shall be valued in an "as is" condition as of the date of the foreclosure sale, without any assumption or expectation that the Property will be repaired or improved in any manner before a resale of the Property after foreclosure; (ii) the valuation shall be based upon an assumption that the foreclosure purchaser desires a resale of the Property for cash promptly (but not later than twelve (12) months) following the foreclosure sale; (iii) all reasonable closing costs customarily borne by the seller in commercial real estate transactions should be deducted from the gross fair market value of the Property, including, without limitation, brokerage commissions, title insurance, a survey of the Property, tax prorations, attorneys' fees, and marketing costs; (iv) the gross fair market value of the Property shall be further discounted to account for any estimated holding costs associated with maintaining the Property pending sale, including, without limitation, utilities expenses, property management fees, taxes and assessments (to the extent not accounted for in (iii) above), and other maintenance, operational and ownership expenses; and (v) any expert opinion testimony given or considered in connection with a determination of the fair market value of the Property must be given by persons having at least five (5) years' experience in appraising property similar to the Property and who have conducted and prepared a complete written appraisal of the Property taking into consideration the factors set forth above.

Loan No: _____

(Continued)

POWERS AND OBLIGATIONS OF TRUSTEE. The following provisions relating to the powers and obligations of Trustee are part of this Deed of Trust:

Powers of Trustee. In addition to all powers of Trustee arising as a matter of law, Trustee shall have the power to take the following actions with respect to the Property upon the written request of Lender and Grantor: (a) join in preparing and filing a map or plat of the Real Property, including the dedication of streets or other rights to the public; (b) join in granting any easement or creating any restriction on the Real Property; and (c) join in any subordination or other agreement affecting this Deed of Trust or the interest of Lender under this Deed of Trust.

Obligations to Notify. Trustee shall not be obligated to notify any other lienholder of the Property of the commencement of a foreclosure proceeding or of the commencement of any other action to which Lender may avail itself as a remedy, except to the extent required by applicable law or by written agreement.

Trustee. In addition to the rights and remedies set forth above, with respect to all or any part of the Property, the Trustee shall have the right to foreclose by notice and sale, and Lender shall have the right to foreclose by judicial foreclosure, in either case in accordance with and to the full extent provided by applicable law.

(Continued)

Substitute Trustee. Lender, at Lender's option, from time to time, and more than once, may appoint in writing a successor or substitute trustee, with or without cause, including the resignation, absence, death, inability, refusal or failure to act of the Trustee. The successor or substitute trustee may be appointed without ever requiring the resignation of the former trustee and without any formality except for the execution and acknowledgment of the appointment by the beneficiary of this Deed of Trust. The successor or substitute trustee shall then succeed to all rights, obligations, and duties of the Trustee. This appointment may be made on Lender's behalf by the President, any Vice President, Secretary, or Cashier of Lender.

NOTICES. Any notice required to be given under this Deed of Trust, including without limitation any notice of default and any notice of sale shall be given in writing, and shall be effective when actually delivered, when actually received by telefacsimile (unless otherwise required by law), when deposited with a nationally recognized overnight courier, or, if mailed, when deposited in the United States mail, as first class, certified or registered mail postage prepaid, directed to the addresses shown near the beginning of this Deed of Trust. Any party may change its address for notices under this Deed of Trust by giving formal written notice to the other parties, specifying that the purpose of the notice is to change the party's address. For notice purposes, Grantor agrees to keep Lender informed at all times of Grantor's current address. Unless otherwise provided or required by law, if there is more than one Grantor, any notice given by Lender to any Grantor is deemed to be notice given to all Grantors.

CHOICE OF VENUE. If there is a lawsuit, Grantor agrees upon Lender's request to submit to the jurisdiction of the courts of Harris County, State of Texas,

SECURITY AGREEMENT INTEREST. This instrument shall also constitute a security agreement with respect to the Personal Property and Trustor grants Beneficiary a security interest in the Personal Property. Without limiting the generality of the definition of the term "Personal Property" contained herein, the term Personal Property shall also include the following, which may or may not be affixed to the Property: all inventory, chattel paper, accounts, equipment, deposit accounts, and general intangibles, without limitation all building materials, located at, arising out of the operations of, or otherwise related to the Property of the business conducted at the Property, including plans and permits, insurance proceeds, trade names, and all records relating to the same; all deposit accounts of Trustor.

TEXAS ASSIGNMENT OF RENTS ACT. Notwithstanding anything to the contrary contained herein, Lender is entitled to all the rights and remedies of an assignee set forth in Chapter 64 of the Texas Property Code, the Texas Assignment of Rents Act ("**TARA**"). This Deed of Trust shall constitute and serve as a security instrument under TARA. Lender shall have the ability to exercise its rights related to the Leases and Rents, in Lender's sole discretion and without prejudice to any other remedy available, as provided in this Deed of Trust or as otherwise allowed by applicable law, including, without limitation, TARA. Notwithstanding anything to the contrary contained in this Deed of Trust or the Credit Agreement, to the extent this Deed of Trust or the Credit Agreement contain any notice or cure period, the date enforcement of Lender's right under TARA begins shall not be affected, extended or otherwise modified by reason of such periods.

Loan No: _

(Continued)

FIXTURE FILING. This Deed of Trust shall be effective as a financing statement filed as a Fixture filing with respect to all Fixtures included within the Real Property and is to be filed for record in the real property records in the Office of the County Clerk for the county or counties where the Real Property (including said Fixtures) is situated.

PRIOR DEEDS OF TRUST. This Deed of Trust is subject and subordinate only to a superior in priority Construction Deed of Trust dated January 26, 2015 from Grantor to Doug Krause, as trustee for the benefit of Beneficiary, to secure an indebtedness described therein (as amended from time to time, the "**Prior Deed of Trust**"). Grantor and Beneficiary acknowledge that all right, title and interest of Beneficiary hereunder are and shall be subject and subordinate to the right, title and interest of the holder or holders of the Prior Deed of Trust. In the event the Prior Deed of Trust is hereafter foreclosed, Grantor hereby assigns to Beneficiary any proceeds from the foreclosure sale (or the transfer in lieu thereof) in excess of the amount necessary to pay off the indebtedness then validly secured by the Prior Deed of Trust thus foreclosed, which excess proceeds shall be applied by Beneficiary to reduce the Indebtedness pursuant to the terms of the Credit Agreement.

MISCELLANEOUS PROVISIONS. The following miscellaneous provisions are a part of this Deed of Trust:

Amendments. This Deed of Trust, together with any Related Documents, constitutes the entire understanding and agreement of the parties as to the matters set forth in this Deed of Trust. No alteration of or amendment to this Deed of Trust shall be effective unless given in writing and signed by the party or parties sought to be charged or bound by the alteration or amendment.

Annual Reports. Grantor shall furnish to Lender, upon request, a certified statement of net operating income received from the Property during Grantor's previous fiscal year in such form and detail as Lender shall require. "Net operating income" shall mean all cash receipts from the Property less all cash expenditures made in connection with the operation of the Property.

Caption Headings. Caption headings in this Deed of Trust are for convenience purposes only and are not to be used to interpret or define the provisions of this Deed of Trust.

Merger. There shall be no merger of the interest or estate created by this Deed of Trust with any other interest or estate in the Property at any time held by or for the benefit of Lender in any capacity, without the written consent of Lender.

Loan No: _

DEED OF TRUST

(Continued)

Governing Law. This Deed of Trust will be governed by federal law applicable to Lender and, to the extent not preempted by federal law, the laws of the State of Texas without regard to its conflicts of law provisions. This Deed of Trust has been accepted by Lender in the State of Texas.

No Waiver by Lender. Lender shall not be deemed to have waived any rights under this Deed of Trust unless such waiver is given in writing and signed by Lender. No delay or omission on the part of Lender in exercising any right shall operate as a waiver of such right or any other right. A waiver by Lender of a provision of this Deed of Trust shall not prejudice or constitute a waiver of Lender's right otherwise to demand strict compliance with that provision or any other provision of this Deed of Trust. No prior waiver by Lender, nor any course of dealing between Lender and Grantor, shall constitute a waiver of any of Lender's rights or of any of Grantor's obligations as to any future transactions. Whenever the consent of Lender is required under this Deed of Trust, the granting of such consent by Lender in any instance shall not constitute continuing consent to subsequent instances where such consent is required and in all cases such consent may be granted or withheld in the sole discretion of Lender.

Severability. If a court of competent jurisdiction finds any provision of this Deed of Trust to be illegal, invalid, or unenforceable as to other circumstance, that finding shall not make the offending provision illegal, invalid, or unenforceable as to any other circumstance. If feasible, the offending provision shall be considered modified so that it becomes legal, valid and enforceable. If the offending provision cannot be so modified, it shall be considered deleted from this Deed of Trust. Unless otherwise required by law, the illegality, invalidity, or unenforceability of any provision of this Deed of Trust shall not affect the legality, validity or enforceability of any other provision of this Deed of Trust.

Successors and Assigns. Subject to any limitations stated in this Deed of Trust on transfer of Grantor's interest, this Deed of Trust shall be binding upon and inure to the benefit of the parties, their successors and assigns. If ownership of the Property becomes vested in a person other than Grantor, Lender, without notice to Grantor, may deal with Grantor's successors with reference to this Deed of Trust and the Indebtedness by way of forbearance or extension without releasing Grantor from the obligations of this Deed of Trust or liability under the Indebtedness.

Time is of the Essence. Time is of the essence in the performance of this Dead of Trust.

Waive Jury. All parties to this Deed of Trust hereby waive the right to any jury trial in any action, proceeding, or counterclaim brought by any party against any other party.

(Continued)

DEFINITIONS. The following capitalized words and terms shall have the following meanings when used in this Deed of Trust. Unless specifically stated to the contrary, all references to dollar amounts shall mean amounts in lawful money of the United States of America. Words and terms used in the singular shall include the plural, and the plural shall include the singular, as the context may require. Words and terms not otherwise defined in this Deed of Trust shall have the meanings attributed to such terms in the Uniform Commercial Code:

Borrower. The word "Borrower" means Applied Optoelectronics, Inc. and includes all co-signers and co-makers signing the Note and all their successors and assigns.

Credit Agreement.. The term "Credit Agreement" means that certain Credit Agreement dated June ___, 2015 by and between Grantor and Lender.

Environmental Laws. The words "Environmental Laws" mean any and all state, federal and local statutes, regulations and ordinances relating to the protection of human health or the environment, including without limitation the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 as amended, 42 U.S.C. Section 9601, et seq. ("CERCLA"), the Superfund Amendments and Reauthorization Act of 1986, Pub. L. No. 99-499 ("SARA"), the Hazardous Materials Transportation Act, 49 U.S.C. Section 1801, et seq., the Resource Conservation and Recovery Act, 42 U.S.C. Section 6901, et seq., or other applicable state or federal laws, rules, or regulations adopted pursuant thereto.

Event of Default. The words "Event of Default" mean any of the events of default set forth in this Deed of Trust in the Events of Default section of this Deed of Trust.

Fixture. The word "Fixture": means any and all materials, supplies, equipment, systems, apparatus, and other items now owned or hereafter acquired by Grantor and now or hereafter attached to, installed in, or used in connection with (temporarily or permanently) any of the Improvements located on the Real Property, which are now owned or hereafter acquired by Grantor and are now or hereafter attached to the Real Property or the Improvements, together with all accessions, appurtenances, replacements, betterments, and substitutions for any of the foregoing and the proceeds thereof.

Guaranty. The word "Guaranty" means the guaranty from guarantor, endorser, surety, or accommodation party to Lender, including without limitation a guaranty of all or part of the Note.

Hazardous Substances. The words "Hazardous Substances" mean materials that, because of their quantity, concentration or physical, chemical or infectious characteristics, may cause or pose a present or potential hazard to human health or the environment when improperly used, treated, stored, disposed of, generated, manufactured, transported or otherwise handled. The words "Hazardous Substances" are used in their very broadest sense and include without limitation any and all hazardous or toxic substances, materials or waste as defined by or listed under the Environmental Laws. The term "Hazardous Substances" also includes, without limitation, petroleum and petroleum by-products or any fraction thereof and asbestos.

(Continued)

Improvements. The word "Improvements" means all existing and future improvements, buildings, structures, mobile homes affixed on the Real Property, facilities, additions, replacements and other construction on the Real Property.

Indebtedness. The word "Indebtedness" means all principal, interest, and other amounts, costs and expenses payable under the Note or Related Documents, together with all renewals of, extensions of, modifications of, consolidations of and substitutions for the Note or Related Documents and any amounts expended or advanced by Lender to discharge Grantor's obligations or expenses incurred by Trustee or Lender to enforce Grantor's obligations under this Deed of Trust, together with interest on such amounts as provided in this Deed of Trust.

Lease. The word "Lease" means any and all leases, licenses, or other agreements (whether written or oral, or now or hereafter in effect) which grant to third parties a possessory interest in and to, or the right to use or occupy, all or any part of the Property, together with all security, including letters of credit, and other deposits or payments made in connection therewith.

Note. The word "Note" means, collectively, that certain (i) promissory note dated as of even date herewith, in the original principal amount of \$10,000,000.00, (ii) promissory note dated as of even date herewith in the original principal amount of \$20,000,000.00 and (iii) promissory note dated as of even date herewith in the original principal amount of \$5,000,000.00, each from Grantor to Lender, together with all renewals of, extensions of, modifications of, refinancings of, consolidations of, and substitutions for the promissory note or agreement. NOTICE TO GRANTOR: THE NOTE CONTAINS A VARIABLE INTEREST RATE, AND THE INDIVIDUAL PROMISSORY NOTES IN ITEMS (ii) AND (iii) OF THIS DEFINITION CONTAIN A REVOLVING CREDIT FEATURE.

Personal Property. The words "Personal Property" mean all equipment, Fixtures, and other articles of personal property now of hereafter owned by Grantor, and now or hereafter attached or affixed to the Real Property; together with all accessions, parts, and additions to, all replacements of, and all substitutions for, any of such property; and together with all proceeds (including without limitation all insurance proceeds and refunds of premiums) from any sale or other disposition of the Property.

Property. The word "Property" means collectively the Real Property and the Personal Property.

(Continued)

Real Property. The words "Real Property" mean the real property, interests and rights, as further described in this Deed of Trust.

Related Documents. The words "Related Documents" mean all promissory notes, credit agreements, loan agreements, environmental agreements, guaranties, security agreements. mortgages, deeds of trust, security deeds, collateral mortgages, and all other instruments, agreements and documents, whether now or hereafter existing, executed in connection with the Indebtedness.

Rents. The word "Rents" means all present and future rents, revenues, incomes, issues, royalties, profits, and other benefits derived from the Property, including without limitation such rents, revenues, incomes, issues, royalties, profits, and other benefits under all existing or future Leases. The word "Rents" shall also mean all "Rents" as defined In Chapter 64 of the Texas Property Code.

Uniform Commercial Code. The term "Uniform Commercial Code" or "UCC" means the Uniform Commercial Code in effect in the State of Texas as of the effective date of this Deed of Trust, and all amendments thereto.

[Remainder of this page intentionally left blank. Signature page follows.]

Loan No: _____

(Continued)

GRANTOR ACKNOWLEDGES HAVING READ ALL THE PROVISIONS OF THIS DEED OF TRUST, AND GRANTOR AGREES TO ITS TERMS.

GRANTOR:

APPLIED OPTOELECTRONICS, INC.

By: <u>/s/ Stefan J. Murry</u> Name: Stefan J. Murry Title: CFO and Chief Strategy Officer

CORPORATE ACKNOWLEDGMENT

STATE OF TEXAS)		
COUNTY OF) SS)		
This instrument was acknowledged before me on June	2015 by		the	of Applied Optoplectropics Inc. 2

This instrument was acknowledged before me on June ____, 2015 by ______, the ______ of Applied Optoelectronics, Inc. a Delaware Corporation, on behalf of said corporation.

Notary Public, State of Texas

(Continued)

"EXHIBIT A"

For Tax Map ID(s):

0015-00-000-1735-907, 0015-00-000-1742-907, 6442-00-001-0000-907 and 6442-00-002-0030-907

TRACT I:

6.312 acre of land, being all of Commercial Reserve "A-1", of **AOI 13139 JESS PIRTLE**, a subdivision in Fort Bend County, Texas, according to the map or plat thereof recorded at Plat No. 20140271 of the Plat Records of Fort Bend County, Texas.

TRACT II:

Commercial Reserve "B-3" of **PARK WEST SECOND REPLAT OF RESERVE "B"**, a subdivision in Fort Bend County, Texas, according to the map or plat thereof recorded in Slide No. 2339/A of the Plat Records of Fort Bend County, Texas.